Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 00.51.76 Decc Main Docket #0768 Date Filed: 05/21/2025 Document جمyو یا ماریک

HUNTON ANDREWS KURTH LLP

Joseph P. Rovira (admitted *pro hac vice*) Catherine A. Rankin (admitted *pro hac vice*) 600 Travis Street, Suite 4200 Houston, Texas 77002 Telephone: (713) 220-4200 HUNTON ANDREWS KURTH LLP

Tyler P. Brown (VSB No. 28072) Henry P. (Toby) Long, III (VSB No. 75134) Riverfront Plaza, East Tower 951 East Byrd Street Richmond, Virginia 23219 Telephone: (804) 788-8200

Counsel for Debtor and Debtor-in-Possession

UNITED STATES BANKRUPTCY COURT EASTERN DISTRICT OF VIRGINIA RICHMOND DIVISION

In re:	: : Chapter 11
HOPEMAN BROTHERS, INC.,	: : Case No. 24-32428 (KLP)
Debtor.	:

NOTICE OF FILING OF REDLINE VERSIONS OF (I) AMENDED CHAPTER 11 PLAN OF REORGANIZATION AND (II) RELATED AMENDED DISCLOSURE STATEMENT

PLEASE TAKE NOTICE that on April 29, 2025, the Debtor and Committee filed the *Plan of Reorganization of Hopeman Brothers, Inc. Under Chapter 11 of the Bankruptcy Code* [Docket No. 689] (the "<u>Plan</u>") and the *Disclosure Statement with Respect to the Plan of Reorganization of Hopeman Brothers, Inc. Under Chapter 11 of the Bankruptcy Code* [Docket No. 690] (the "<u>Disclosure Statement</u>").

PLEASE TAKE FURTHER NOTICE that on May 20, 2025, the Debtor and Committee filed the *Amended Plan of Reorganization of Hopeman Brothers, Inc. Under Chapter* 11 of the Bankruptcy Code [Docket No. 766] (the "<u>Amended Proposed Plan</u>") and the Disclosure Statement for Amended Plan of Reorganization of Hopeman Brothers, Inc. Under Chapter 11 of the Bankruptcy Code [Docket No. 767] (the "<u>Amended Disclosure Statement</u>").

PLEASE TAKE FURTHER NOTICE THAT attached hereto as (a) <u>Exhibit A</u> is a redline of the Amended Proposed Plan to the Plan and (b) <u>Exhibit B</u> is a redline of the Amended Disclosure Statement to the Disclosure Statement (without exhibits).



Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 2 of 129

Dated: May 21, 2025 Richmond, Virginia

/s/ Henry P. (Toby) Long, III

Tyler P. Brown (VSB No. 28072) Henry P. (Toby) Long, III (VSB No. 75134) **HUNTON ANDREWS KURTH LLP** Riverfront Plaza, East Tower 951 East Byrd Street Richmond, Virginia 23219 Telephone: (804) 788-8200 Facsimile: (804) 788-8218 Email: tpbrown@HuntonAK.com hlong@HuntonAK.com

- and –

Joseph P. Rovira (admitted *pro hac vice*) Catherine A. Rankin (admitted *pro hac vice*) **HUNTON ANDREWS KURTH LLP** 600 Travis Street, Suite 4200 Houston, TX 77002 Telephone: (713) 220-4200 Facsimile: (713) 220-4285 Email: josephrovira@HuntonAK.com crankin@HuntonAK.com

Counsel for the Debtor and Debtor in Possession

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 3 of 129

EXHIBIT A

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 4 of 129

HUNTON ANDREWS KURTH LLP

Joseph P. Rovira (admitted *pro hac vice*) Catherine A. Rankin (admitted *pro hac vice*) 600 Travis Street, Suite 4200 Houston, Texas 77002 Telephone: (713) 220-4200

Attorneys for Debtor and Debtor in Possession

CAPLIN & DRYSDALE, CHARTERED

Kevin C. Maclay (admitted *pro hac vice*) Todd E. Phillips (admitted *pro hac vice*) Jeffrey A. Liesemer (VSB No. 35918) Nathaniel R. Miller (admitted *pro hac vice*) 1200 New Hampshire Avenue NW, 8th Floor Washington, DC 20036 Telephone: (202) 862-5000

Counsel for the Official Committee of Unsecured Creditors

HUNTON ANDREWS KURTH LLP

Tyler P. Brown (VSB No. 28072) Henry P. (Toby) Long, III (VSB No. 75134) Riverfront Plaza, East Tower 951 East Byrd Street Richmond, Virginia 23219 Telephone: (804) 788-8200

MORGAN, LEWIS & BOCKIUS LLP

Brady Edwards (admitted *pro hac vice*) 1000 Louisiana Street, Suite 4000 Houston, TX 77002-5006 Telephone: (713) 890-5000

W. Brad Nes (admitted *pro hac vice*) 1717 Main Street, Suite 3200 Dallas, TX 75201-7347 Telephone: (214) 466-4000

Jeffrey S. Raskin (admitted *pro hac vice*) One Market, Spear Street Tower, 28th Floor San Francisco, CA 94105-1596 Telephone: (415) 442-1000

David Cox (admitted *pro hac vice*) 300 South Grand Avenue, 22nd Floor Los Angeles, CA 90071-3132 Telephone: (213) 612-7315

Special Insurance Counsel for the Official Committee of Unsecured Creditors

UNITED STATES BANKRUPTCY COURT EASTERN DISTRICT OF VIRGINIA RICHMOND DIVISION

:

:

:

:

In re:

HOPEMAN BROTHERS, INC.,

Debtor.

Chapter 11

Case No. 24-32428 (KLP)

<u>AMENDED</u> PLAN OF REORGANIZATION OF HOPEMAN BROTHERS, INC. UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

Dated: April 29 May 21, 2025

PLAN OF REORGANIZATION OF HOPEMAN BROTHERS, INC. UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

INTRODUCTION

Hopeman and the Committee, as the Plan Proponents, now jointly propose the following chapter 11 plan of reorganization for Hopeman Brothers, Inc., in accordance with section 1121(a) of title 11 of the United States Code. Capitalized terms used in this plan of reorganization and not otherwise defined shall have the meanings ascribed to them in <u>Article 1.A</u> hereof.

This Plan provides, among other things, for the issuance of the Asbestos Permanent Channeling Injunction under sections 105(a) and 524(g) of the Bankruptcy Code that will, among other things, result in the channeling of certain asbestos-related liabilities of Hopeman (excluding liabilities for worker compensation claims) to a newly created Asbestos Trust, as more fully described herein. The terms, conditions, and Entities subject to the Asbestos Permanent Channeling Injunction are set forth in <u>Section 10.3</u> of this Plan. Except as otherwise permitted under the terms of this Plan, the Asbestos Permanent Channeling Injunction will prohibit any acts to collect, recover, or offset any asbestos-related liabilities of Hopeman against Hopeman, Reorganized Hopeman, certain third parties, and various persons or entities related to those parties. Under section 524(g), all distributions paid under this Plan to the holders of Asbestos Claims and Demands will be the responsibility of the Asbestos Trust, and Hopeman and Reorganized Hopeman shall have no further personal liability for such claims.

Additionally, under the terms of <u>Section 8.12</u> of this Plan, holders of Asbestos Claims and Demands may bring actions against Reorganized Hopeman and, to the extent they have obtained a judgment against Reorganized Hopeman or have the right to pursue direct actions under applicable nonbankruptcy law, such holders may bring judgment-enforcement or direct actions against Non-Settling Asbestos Insurers, in accordance with <u>Section 8.13</u> of this Plan, to obtain the benefits of Asbestos Insurance Coverage. All other claims against Hopeman will receive payments and distributions in accordance with this Plan and the Asbestos Trust Documents.

The foregoing description of this Plan is qualified in its entirety by the terms, conditions, and definitions set forth below.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 6 of 129

TABLE OF CONTENTS

ARTICLE I	LE I DEFINITIONS AND INTERPRETATIONS		
	A	A. Definitions	1
	E	B. Interpretation; Application of Definitions and Rules of Construction	13
	C	2. Reference to Hopeman or Reorganized Hopeman	14
	Γ	D. Reference to Monetary Figures	14
ARTICLE II	ADM	INISTRATIVE EXPENSE AND PRIORITY TAX CLAIMS	14
	2.1.	Administrative Expense Claims	14
	2.2.	Professional Fee Claims	15
	2.3.	Priority Tax Claims	15
ARTICLE III	CLA	SSIFICATION OF CLAIMS AND INTERESTS	16
	3.1.	Classification	16
ARTICLE IV	TRE	ATMENT OF CLASSIFIED CLAIMS AND INTERESTS	16
	4.1.	Class 1 – Priority Non-Tax Claims	16
	4.2.	Class 2 - Secured Claims	16
	4.3.	Class 3 - General Unsecured Claims	17
	4.4.	Class 4 – Channeled Asbestos Claims	17
	4.5.	Class 5 – Equity Interests	17
ARTICLE V		RIBUTIONS UNDER THE PLAN ON ACCOUNT OF IMS OTHER THAN CHANNELED ASBESTOS CLAIMS	18
	5.1.	Distributions	
	5.2.	Record Date for Holders of Claims	
	5.3.	Postpetition Interest on Claims	
	5.4.	Means of Cash Payment	
	5.5.	Delivery of Distributions	
	5.6.	Time Bar to Cash Payments	
	5.7.	Prepayment of Claims	
	5.8.	Minimum Distributions; Fractional Cents	
	5.9.	Setoff and Recoupment	
ARTICLE VI		ATMENT OF EXECUTORY CONTRACTS AND UNEXPIRE	
MATCHE VI		SES	
	6.1.	General Treatment	19

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 7 of 129

	6.2.	Asbestos Insurance Agreements	20
	6.3.	Bar to Rejection Damages	20
ARTICLE VII	PROC	CEDURES FOR RESOLVING AND TREATING DISPUTED	
		MS OTHER THAN CHANNELED ASBESTOS CLAIMS AND	•
		ESSIONAL FEE CLAIMS	
	7.1.	Disputed Claims	
	7.2.	Objection to Claims	
	7.3.	Payments and Distributions with Respect to Disputed Claims	
ARTICLE VIII		NS FOR IMPLEMENTATION OF THE PLAN	
	8.1.	Generally	21
	8.2.	Transactions on the Effective Date	22
	8.3.	The Asbestos Trust	22
	8.4.	Amended Certificate of Incorporation and By-Laws	26
	8.5.	Net Reserve Funds	26
	8.6.	Reorganized Hopeman Common Stock	26
	8.7.	Corporate Governance of Reorganized Hopeman	26
	8.8.	Operations of Hopeman Between Confirmation and the Effective Date	27
	8.9.	Corporate Action	27
	8.10.	Restructuring Transactions	27
	8.11.	Continued Corporate Existence	27
	8.12.	Actions Against Reorganized Hopeman or Wayne to Obtain Benefits of Asbestos Insurance Coverage	27
	8.13.	Actions Against Non-Settling Asbestos Insurers to Obtain Benefits of Asbestos Insurance Coverage	28
	8.14.	No Actions Against Released Parties	30
	8.15.	Special Provision Pertaining to HII	30
	8.16.	Payments or Distributions from the Asbestos Trust	31
	8.17.	Settlements with Non-Settling Asbestos Insurers After the Effective Date	32
	8.18.	Insurance Neutrality	32
	8.19.	Effectuating Documents; Further Transactions	32
ARTICLE IX	EFFE	CT OF CONFIRMATION	
	9.1.	Dissolution of Committee; Discharge of the Future Claimants' Representative	
	9.2.	Vesting of Assets	

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 8 of 129

	9.3.	Preservation of Certain Causes of Action, Rights to Settle Claims and Compromise Controversies, and Defenses	33
	9.4.	Terms of Injunction and Automatic Stay	34
	9.5.	No Liability for Certain Tax Claims	34
	9.6.	No Successor Liability	34
ARTICLE X		HARGE, INJUNCTIONS, EXCULPATION, RELEASES, AND LEMENT	34
	10.1.	Discharge of Hopeman and Reorganized Hopeman	34
	10.2.	Hopeman Discharge Injunction	35
	10.3.	Asbestos Permanent Channeling Injunction	35
	(a) Scope of Injunction	35
	(b) Reservations	36
	10.4.	Exculpation	37
	10.5.	Releases by Hopeman and Estate	37
	10.6.	Release of Hopeman's Directors and Officers by Releasing Parties.	37
	10.7.	Release of Avoidance Actions and Estate Causes of Action Against Vendor Released Parties	38
	10.8.	No Actions on Account of Released Claims	38
	10.9.	Compromise and Settlement of Claims and Interests and Approval of Certain Settling Insurers Agreement	38
	10.10.	Disallowed Claims	39
ARTICLE XI		DITIONS PRECEDENT TO CONFIRMATION AND SUMMATION OF THE PLAN	39
	11.1.	Conditions Precedent to Confirmation of the Plan	39
	11.2.	Conditions Precedent to the Effective Date	44
	11.3.	Waiver of Conditions Precedent	45
	11.4.	Failure to Achieve the Effective Date	45
ARTICLE XII	RETA	INED JURISDICTION OF BANKRUPTCY COURT	46
	12.1.	Retention of Jurisdiction	46
	12.2.	Other Proceedings	48
	12.3.	Modification of Plan	48
	12.4.	Consent to Jurisdiction	49
ARTICLE XIII	MISC	ELLANEOUS PROVISIONS	49
	13.1.	Governing Law	49
	13.2.	Notices	49
	13.3.	Plan Supplement	50

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 9 of 129

13.4.	Inconsistencies	50
13.5.	Tax Reporting and Compliance	50
13.6.	Payment of Statutory Fees	51
13.7.	Exemption from Transfer Taxes	51
13.8.	Recordable Order	51
13.9.	Binding Effect	51
13.10.	Severability	51
13.11.	Further Authorizations	51
13.12.	General Statements	52
13.13.	Entire Agreement	52

EXHIBITS TO PLAN

Exhibit A	Asbestos Trust Agreement
Exhibit B	Asbestos Trust Distribution Procedures
Exhibit C	Amended By-Laws of Reorganized Hopeman
Exhibit D	Amended Certificate of Incorporation
Exhibit E	Asbestos Personal Injury Claimant Release
Exhibit F	Restructuring Transaction
Exhibit G	List of Vendor Released Parties
Exhibit H	Asbestos Insurance Policies

ARTICLE I

DEFINITIONS AND INTERPRETATIONS

A. Definitions. The following terms provided in this <u>Article I</u> apply to the Plan and shall have the respective meanings specified below:

1.1. Administrative Expense Claim means a Claim for any cost or expense of administration of the Estate under sections 503(b), 507(a)(1), 507(b), or 1114(e)(2) of the Bankruptcy Code, including: (a) any actual and necessary cost and expense of preserving the Estate or operating Hopeman's business incurred after the Petition Date and through the Effective Date; (b) any indebtedness or obligations incurred or assumed by Hopeman after the Petition Date and through the Effective Date; (c) any Allowed compensation for Professional services rendered, and Allowed reimbursement of expenses incurred, by a Professional retained by order of the Bankruptcy Court or otherwise Allowed pursuant to section 503(b) of the Bankruptcy Code; and (d) all fees due and payable pursuant to section 1930 of title 28 of the United States Code.

1.2. Administrative Expense Claim Bar Date means the first Business Day that is thirty (30) calendar days after the Effective Date, which will be the deadline by which parties seeking payment of Administrative Expense Claims (other than Professional Fee Claims) must file a motion seeking Allowance of such claim.

1.3. Allowed means, with respect to any Claim (other than a Channeled Asbestos Claim) or Interest, such Claim or Interest or portion thereof against or in Hopeman: (a) that has been listed by Hopeman in the Schedules as liquidated in amount and not disputed or contingent and for which no contrary Proof of Claim has been filed; (b) as to which the deadline for objecting or seeking estimation has passed, and no objection or request for estimation has been filed, or Hopeman otherwise has assented to the validity thereof; (c) as to which any objection or request for estimation that has been filed has been settled, waived, withdrawn, overruled, or denied by a Final Order; or (d) that is allowed pursuant to the terms of (i) a Final Order, (ii) an agreement by and between the holder of such Claim or Interest and Hopeman, or (iii) the Plan.

1.4. Allowed Amount means, with respect to any Claim (other than Channeled Asbestos Claims), the dollar amount of such Claim as Allowed. Unless otherwise provided in the Plan or a Final Order of the Bankruptcy Court or the District Court, the Allowed Amount of an Allowed Claim shall not include interest or penalties accruing on such Allowed Claim from and after the Petition Date.

1.5. Amended By-Laws means the amended and restated by-laws of Reorganized Hopeman, which shall be substantially in the form attached as <u>Exhibit C</u> as may be modified by, *inter alia*, a revised version filed in the Plan Supplement.

1.6. Amended Certificate of Incorporation means the amended and restated certificate of incorporation of Reorganized Hopeman, which shall be substantially in the form attached as Exhibit D hereto as may be modified by, *inter alia*, a revised version filed in the Plan Supplement.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 12 of 129

1.7. Asbestos CIP Agreement means an agreement between Hopeman and an Asbestos Insurer that (a) is based on, arises from, or is attributable to an Asbestos Insurance Policy and (b), among other things, establishes a framework or formula for the Asbestos Insurer's payment of indemnity, liability, or defense costs to Hopeman with respect to Asbestos Personal Injury Claims. The term "Asbestos CIP Agreement" includes that certain Settlement Agreement, dated December 18, 2009, between Hopeman and Century Indemnity Company, as successor to CCI Insurance Company, as successor to Insurance Company of North America. Notwithstanding any of the foregoing, "Asbestos CIP Agreement" does not include any prepetition settlement agreement (or any related indemnity obligations thereunder) that does not currently provide rights in favor of Hopeman to continuing coverage or to payment of insurance proceeds.

1.8. Asbestos Claim means an Asbestos Personal Injury Claim or an Asbestos Indirect Claim.

1.9. Asbestos Indirect Claim means any Claim, now existing or hereafter arising, that is held or asserted by an Entity that is not an Asbestos Insurer against Hopeman, or for which Hopeman is liable, for contribution, reimbursement, subrogation, or indemnification, or any other indirect or derivative recovery, on account of or with respect to any Asbestos Personal Injury Claim. Notwithstanding the foregoing, any indemnity claims of Released Parties asserted under <u>Section 8.14</u> hereof shall not be deemed to be or treated as an Asbestos Indirect Claim or an Asbestos Claim.

1.10. Asbestos Insurance Cooperation Obligations means, collectively, the assistance and cooperation, inspection and audit, and notice of occurrence provisions set forth in the Asbestos Insurance Policies and any other provisions purporting to require the cooperation of the insured party.

1.11. Asbestos Insurance Coverage means all rights, title, privileges, interests, claims, demands, benefits, or entitlements to proceeds, payments, indemnity, or reimbursement under any Asbestos Insurance Policy relating to any Channeled Asbestos Claim.

1.12. Asbestos Insurance Policies means the insurance policies identified on <u>Exhibit H</u> of the Plan and any other insurance policy of Hopeman, whether known or unknown, that provides or potentially provides coverage for any Channeled Asbestos Claim.

1.13. Asbestos Insurance Rights means any and all of Hopeman's rights, title, privileges, interests, claims, demands, or entitlements in or to any insurance coverage, defense, indemnity, proceeds, payments, escrowed funds, initial or supplemental dividends, scheme payments, supplemental scheme payments, state guaranty fund payments, causes of action, and choses in action under, for, or related to (i) the Asbestos Insurance Settlements, (ii) the Asbestos Insurance Policies, or (iii) the Asbestos CIP Agreements, whether now existing or hereafter arising, accrued or unaccrued, liquidated or unliquidated, matured or unmatured, disputed or undisputed, fixed or contingent, including:

(a) any and all rights of Hopeman to pursue or receive payment, reimbursement, or proceeds under any Asbestos Insurance Policy or any Asbestos CIP Agreement, whether for indemnity, liability, defense costs, or otherwise;

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 13 of 129

(b) the remaining balance of any proceeds of any and all payments received by Hopeman on account of an Asbestos Insurance Policy or Asbestos CIP Agreement at any time on, before, or after the Effective Date, together with all interest earned on such proceeds;

(c) any and all rights of Hopeman to pursue or receive payments from any insolvent Asbestos Insurer, whether in receivership, liquidation, rehabilitation, runoff, or scheme of arrangement, or any other form of proceeding, or from any insolvent Asbestos Insurer's estate, and the remaining balance of any proceeds of all payments received by Hopeman from any such Asbestos Insurer or such insolvent insurer's estate, at any time on, before, or after the Effective Date, together with all interest earned on such proceeds;

(d) any and all rights of Hopeman to pursue or receive payments with respect to Asbestos Personal Injury Claims from any insurance guaranty association;

(e) any and all rights of Hopeman to pursue or receive payment to any exception to a workers' compensation exclusion in any Asbestos Insurance Policy; *provided*, *however*, that Asbestos Insurance Rights shall not include any rights or obligations under any insurance policy, settlement agreement, or coverage in-place agreement to which any Asbestos Insurer is a party to the extent, but only to the extent, that such rights or obligations pertain solely to coverage for Worker Compensation Claims; and

(f) any and all Extracontractual Claims, and any and all rights of Hopeman to pursue or receive payments or recoveries on account thereof.

1.14. Asbestos Insurance Settlement means (a) the Certain Settling Insurers Agreement; (b), with the exception of the Certain Settling Insurers Agreement, each agreement that, prior to the Effective Date, has been entered into by an Asbestos Insurer and Hopeman with consent of the Committee and approved by Final Order of the Bankruptcy Court; or (c) any agreement that satisfies the requirements of clauses (a) through (c) of Section 8.17.

1.15. Asbestos Insurer means any Entity, including any insurance company, broker, or guaranty association, that has issued, or that has any actual or potential liabilities, duties or obligations under or with respect to any Asbestos Insurance Policy.

1.16. Asbestos Permanent Channeling Injunction means the injunction provided for in <u>Section 10.3</u> hereof.

1.17. Asbestos Personal Injury Claim means any Claim or allegation or portion thereof against, or any debt, liability, or obligation of, Hopeman, whether now existing or hereafter arising, whether in the nature of or sounding in tort, or under contract, warranty, or any other theory of law, equity, or admiralty for, arising out of, resulting from, or attributable to, directly or indirectly, death, bodily injury, sickness, disease, or any other actual or alleged personal injury, physical, emotional, or otherwise, to persons, caused or allegedly caused, directly or indirectly, by the presence of, or exposure to, asbestos, including asbestos-containing products or materials engineered, designed, marketed, manufactured, fabricated, constructed, sold, supplied, produced, installed, maintained, serviced, specified, selected, repaired, removed, replaced, released, distributed, or in any other way used by Hopeman or any other Entity for whose products or operations Hopeman has liability or is alleged to have liability, but only to the extent arising,

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 14 of 129

directly or indirectly, from acts, omissions, business, or operations of Hopeman (including the acts, omissions, business, or operations of any other Entity for whose products or operations Hopeman has liability, but only to the extent of Hopeman's liability for such acts, omissions, business, or operations), including all related claims, debts, obligations, or liabilities (such as any claim or demand for compensatory damages; loss of consortium; medical monitoring; wrongful death; survivorship; proximate, consequential, general, special, or punitive damages). As used herein, the term "Asbestos Personal Injury Claim" shall not include any Worker Compensation Claim.

1.18. Asbestos Personal Injury Claimant Release means a general release by a holder of an Asbestos Personal Injury Claim in substantially the form attached as <u>Exhibit E</u> hereto.

1.19. Asbestos Trust means the Asbestos Trust established pursuant to section 524(g) of the Bankruptcy Code and in accordance with the Plan, the Confirmation Order and the Asbestos Trust Agreement, which trust shall constitute a "qualified settlement fund" under section 468B of the Internal Revenue Code and shall be a Delaware statutory trust.

1.20. Asbestos Trust Advisory Committee means the Asbestos Trust advisory committee established pursuant to the terms of the Plan and the Asbestos Trust Agreement and identified in the Asbestos Trust Agreement.

1.21. Asbestos Trust Agreement means the agreement, to be dated as of the Effective Date, by and among Reorganized Hopeman, the Future Claimants' Representative, the Asbestos Trustee, the Asbestos Trust Advisory Committee, and the Delaware Trustee, governing the creation and the terms of the Asbestos Trust, which shall be substantially in the form attached as <u>Exhibit A</u> hereto.

1.22. Asbestos Trust Assets means, collectively: (a) the Asbestos Trust Contribution; (b) the Asbestos Insurance Rights; (c) all other assets, rights (including Causes of Action), and benefits assigned, transferred or conveyed to the Asbestos Trust in connection with the Plan or any Plan Documents; and (d) all proceeds of the foregoing.

1.23. Asbestos Trust Contribution means a contribution or contributions by or on behalf of Hopeman or Reorganized Hopeman to the Asbestos Trust of (a) all Cash held by Hopeman (provided, however, that any Cash, up to an amount to be disclosed in the Plan Supplement held by Hopeman as of the Effective Date shall be set aside by Hopeman or Reorganized Hopeman, as applicable, in a segregated account as the Net Reserve Funds in accordance with Section 8.5 of the Plan), (b) the Excess Net Reserve Funds, (c) all of Hopeman's or Reorganized Hopeman's, as applicable, rights or Proceeds payable under any and all agreements with Settled Asbestos Insurers (including any Proceeds held in or deposited into any qualified settlement fund pursuant to, or in connection with, an agreement with Settled Asbestos Insurers); and (d) all of Hopeman's or Reorganized Hopeman's, as applicable Hopeman's, as applicable, Asbestos Insurers Rights.

1.24. Asbestos Trust Distribution Procedures means the trust distribution procedures for the Asbestos Trust, which shall be substantially in the form attached as <u>Exhibit B</u> hereto, and such additional procedures as may subsequently be adopted by the Asbestos Trust, which provide for the resolution, liquidation, and satisfaction of Channeled Asbestos Claims.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 15 of 129

1.25. Asbestos Trust Documents means, collectively: (a) the Asbestos Trust Agreement; (b) the Asbestos Trust Distribution Procedures; and (c) any other agreements, instruments, and documents governing the establishment and administration of the Asbestos Trust, which shall be materially consistent with the terms of the Plan, the Asbestos Trust Agreement, and the Asbestos Trust Distribution Procedures, as the same may be amended or modified from time to time, in accordance with the terms thereof.

1.26. Asbestos Trust Expenses means any of the liabilities, costs, or expenses incurred by or on behalf of the Asbestos Trust (other than liabilities to holders of Asbestos Personal Injury Claims in respect of such Claims), in carrying out the terms of the Asbestos Trust Agreement.

1.27. Asbestos Trustee means the individual set forth in the Asbestos Trust Agreement and appointed pursuant to the Confirmation Order to serve as the trustee for the Asbestos Trust in accordance with the terms of the Plan and the Asbestos Trust Agreement, and any successor trustee thereto appointed in accordance with the Asbestos Trust Agreement.

1.28. Avoidance Action(s) means any avoidance or recovery action under any of sections 502(d), 542, 544, 545, 547, 548, 549, 550, 551 and 553 of the Bankruptcy Code, or under similar state or federal statutes and common law, whether or not litigation has been commenced with respect to such Cause of Action as of the Effective Date.

1.29. Bankruptcy Code means title 11 of the United States Code, 11 U.S.C. §§ 101 et seq., as in effect on the Petition Date, together with all amendments, modifications and replacements of the foregoing, as the same may exist on any relevant date to the extent applicable to the Chapter 11 Case.

1.30. Bankruptcy Court means the United States Bankruptcy Court for the Eastern District of Virginia, Richmond Division, or such other court as may have jurisdiction over the Chapter 11 Case or any proceeding within, or appeal of an order entered in, the Chapter 11 Case including, to the extent of a withdrawal of reference under 28 U.S.C. § 157 or the requirement for final approval, the District Court.

1.31. Bankruptcy Rules means, collectively: (a) the Federal Rules of Bankruptcy Procedure as promulgated by the United States Supreme Court under section 2075 of title 28 of the United States Code; (b) the Federal Rules of Civil Procedure, as applicable to the Chapter 11 Case or any proceedings therein; and (c) the local rules of the Bankruptcy Court, all as amended from time to time and applicable to the Chapter 11 Case.

1.32. Business Day means any day other than a Saturday, Sunday or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).

1.33. Cash means legal tender of the United States of America and equivalents thereof.

1.34. Causes of Action is intended to be read as broadly as possible and includes, but is not limited to, any Claims, causes of action (including Avoidance Actions), demands, actions, suits, obligations, liabilities, crossclaims, counterclaims, defenses, offsets, or set offs of any kind or character whatsoever, in each case whether known or unknown, contingent or noncontingent, matured or unmatured, suspected or unsuspected, foreseen or unforeseen, direct or indirect, choate

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 16 of 129

or inchoate, existing or hereafter arising, under statute, in contract, in tort, in law, or in equity, or pursuant to any other theory of law, federal or state, whether asserted or assertable directly or derivatively in law or equity or otherwise by way of claim, counterclaim, crossclaim, third party action, action for indemnity, contribution, or otherwise.

1.35. Certain Settling Insurers Agreement means the Settlement Agreement and Release, dated as of July 10, 2024, by and among Hopeman, Continental Casualty Company, Fidelity & Casualty Company, Lexington Insurance Company, Granite State Insurance Company, the Insurance Company of the State of Pennsylvania, National Union fire Insurance Company of Pittsburgh, PA, and General Reinsurance Corporation.

1.36. Channeled Asbestos Claimant means the holder of a Channeled Asbestos Claim.

1.37. Channeled Asbestos Claims means, collectively, the Asbestos Claims and Demands.

1.38. Chapter 11 Case means Hopeman's case under chapter 11 of the Bankruptcy Code, captioned *In re Hopeman Brothers, Inc.*, Case No. 24-32428 (KLP) pending in the Bankruptcy Court.

1.39. Claim means a claim, as defined in section 101(5) of the Bankruptcy Code.

1.40. Claims and Balloting Agent means the claims, noticing and balloting agent in the Chapter 11 Case, Kurtzman Carson Consultants, LLC dba Verita Consulting.

1.41. Class means a category of Claims or Interests established under section 3.1 of the Plan pursuant to sections 1122 and 1123(a)(1) of the Bankruptcy Code.

1.42. Committee means the Official Committee of Unsecured Creditors appointed by the Office of the United States Trustee for the Eastern District of Virginia in this Chapter 11 Case on July 22, 2024.

1.43. Confirmation Date means the date on which the Confirmation Order is entered on the docket of the Chapter 11 Case within the meaning of Bankruptcy Rules 5003 and 9021 or on the docket of the District Court.

1.44. Confirmation Hearing means the hearing(s) to be held by the Bankruptcy Court or the District Court pursuant to section 1128 of the Bankruptcy Code to consider confirmation of the Plan, as such hearing(s) may be adjourned or continued from time to time.

1.45. Confirmation Order means (a) the order of the District Court confirming the Plan under section 1129 of the Bankruptcy Code or (b) collectively, the order of the Bankruptcy Court confirming the Plan under section 1129 of the Bankruptcy Code and the order of the District Court affirming such order, which in either case shall contain the Asbestos Permanent Channeling Injunction.

1.46. Contribution Claim has the meaning ascribed to that term in <u>Section 8.13</u> of this Plan.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 17 of 129

1.47. Debtor in Possession means Hopeman as debtor in possession in the Chapter 11 Case pursuant to section 1101(1) of the Bankruptcy Code.

1.48. Delaware Trustee means the Entity appointed in accordance with section 8.3(c), herein, to serve as the "Delaware Trustee" in accordance with the terms of the Plan and the Asbestos Trust Agreement.

1.49. Demand means a "demand," as defined in section 524(g)(5) of the Bankruptcy Code, against Hopeman.

1.50. Designated Person means each of the following former directors and officers of Hopeman or Wayne: Albert Arendt Hopeman, Jr., Bertram C. Hopeman, Charles Johnson, and Kenneth Wood.

1.51. Disallowed means, when used with respect to a Claim against Hopeman, other than an Asbestos Personal Injury Claim, a Claim that is not Allowed and (a) has been disallowed by a Final Order, (b) is listed in the Schedules as contingent, disputed, or unliquidated and as to which no Proof of Claim or request for payment of an Administrative Expense Claim has been timely filed or deemed timely filed with the Bankruptcy Court, (c) is not listed in the Schedules and as to which no Proof of Claim or request for payment of an Administrative Expense Claim has been timely filed or deemed timely filed with the Bankruptcy Court, or (d) has been withdrawn, in whole or in part (but solely to the extent of such withdrawal), by the holder thereof.

1.52. Disclosure Statement means the written disclosure statement that relates to the Plan, including the exhibits and schedules thereto, as approved by the Bankruptcy Court after the Petition Date as containing adequate information pursuant to section 1125 of the Bankruptcy Code and Rule 3017 of the Bankruptcy Rules, as such disclosure statement may be amended, modified, or supplemented from time to time.

1.53. Disputed Claim means a Claim against Hopeman or any portion thereof, other than a Channeled Asbestos Claim, that is neither Allowed nor Disallowed or is contingent, disputed, or unliquidated.

1.54. Distribution means any: (a) Cash; (b) property; or (c) interest in property to be paid or distributed hereunder to the holders of Allowed Claims, not including Channeled Asbestos Claims.

1.55. Distribution Record Date means the record date for determining an entitlement to receive Distributions under the Plan on account of Allowed Claims, other than Channeled Asbestos Claims, which date shall be the Confirmation Date.

1.56. District Court means the United States District Court for the Eastern District of Virginia.

1.57. Effective Date means the first Business Day upon which each condition set forth in section 11.2 has been satisfied or waived.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 18 of 129

1.58. Encumbrance means, with respect to any property (whether real or personal, or tangible or intangible), any mortgage, Lien, pledge, charge, security interest, assignment, or encumbrance of any kind or nature in respect of such property (including any conditional sale or other title retention agreement, any security agreement, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code or comparable law of any jurisdiction) to secure payment of a debt or performance of an obligation.

1.59. Entity means any Person or organization created by law, including, without limitation, any individual, company, corporation, limited liability company, partnership, association, joint stock company, joint venture, estate, trust, unincorporated organization, or government or any political subdivision thereof.

1.60. Equity Interest means any right, title and ownership interest in Hopeman.

1.61. Estate means the estate created under section 541 of the Bankruptcy Code in the Chapter 11 Case.

1.62. Excess Net Reserve Funds means any Net Reserve Funds remaining after the satisfaction in full of (a) all Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Priority Claims, Allowed Secured Claims and Allowed General Unsecured Claims (including the payment of any interest on any such Claims that may be allowed under the Plan or required to be paid by the Bankruptcy Code), (b) any fees and expenses of Hopeman, Reorganized Hopeman, the Committee, the Future Claimants' Representative that are payable by Hopeman or Reorganized Hopeman, as applicable, pursuant to Section 9.1 of the Plan, (c) any fees payable pursuant to section 1930 of title 28 of the United States Code (whether those fees pursuant to section 1930 of title 28 are payable before or after the Effective Date), (d) any other amounts that the Plan provides are to be paid from the Net Reserve Funds, and (e) such amounts as Reorganized Hopeman determines, in the reasonable exercise of its discretion, are or will be sufficient to fully satisfy (as and when due) all franchise taxes and other expenditures that are necessary to maintain its corporate existence in good standing under the laws of the state of its formation or that otherwise are necessary for Reorganized Hopeman to conduct the business for which Section 8.11 of the Plan provides.

1.63. Exculpated Parties means, collectively, (a) Hopeman, (b) the Committee, solely in its capacity as such, (c) the Future Claimants' Representative, solely in her capacity as such, and (d) Professionals of any of the foregoing Entities, solely in their capacity as such. <u>An Entity</u> shall be an "Exculpated Party" only to the extent that it is a fiduciary that has performed duties in connection with the Chapter 11 Case.

1.64. Executory Contract means any unexpired lease or executory contract of Hopeman that is subject to treatment under section 365 of the Bankruptcy Code.

1.65. Extracontractual Claim means any claim against an Asbestos Insurer for "bad faith," extracontractual, or tort liability that is based on, arises from, or is attributable to an Asbestos Insurance Policy or Asbestos CIP Agreement.

1.66. Final Order means a judgment or an order, as the case may be, as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 19 of 129

which no appeal, petition for certiorari or other proceedings for reargument or rehearing shall then be pending; provided, however, that if an appeal, writ of certiorari, reargument or rehearing thereof has been filed or sought: (a)(i) such judgment or order shall have been affirmed by the highest court to which such judgment or order was appealed; or (ii) certiorari shall have been denied or reargument or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for certiorari or move for reargument or rehearing shall have expired; or (b) such appeal, writ of certiorari, or request for reargument or rehearing shall have been dismissed with prejudice by the filing or seeking party.

1.67. Future Claimants' Representative means (a) after the Petition Date but prior to the Effective Date, Marla Rosoff Eskin (or any court-appointed alternative or successor) in her capacity as the court-appointed legal representative for all holders of Demands pursuant to section 524(g) of the Bankruptcy Code; or (b) the individual identified in the Asbestos Trust Agreement and appointed by the Bankruptcy Court to serve as the legal representative for all holders of Demands on and after the Effective Date, pursuant to section 524(g) of the Bankruptcy Code, and any successor to such individual who is appointed to serve in such capacity in accordance with the Asbestos Trust Agreement.

1.68. General Unsecured Claim means a Claim against Hopeman that is not secured by a valid and enforceable Lien against property of Hopeman and that is not a Secured Claim, an Administrative Expense Claim, a Priority Claim, a Priority Tax Claim, or Channeled Asbestos Claim.

1.69. General Unsecured Recovery Pool means Cash in the amount of [\$200,000] to be paid from the Net Reserve Funds.

1.70. HII means Huntington Ingalls Industries, Inc.

1.71. Hopeman means (a) Hopeman Brothers, Inc., a Virginia corporation, and its predecessors; and (b) the debtor and Debtor in Possession in the Chapter 11 Case.

1.72. Impaired means, when used with respect to a Claim or an Equity Interest, a Claim or Equity Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.

1.73. Insurance Policy Action has the meaning ascribed to that term in <u>Section 8.13</u> of this Plan.

1.74. Insured Asbestos Claim means a Channeled Asbestos Claim that is not an Uninsured Asbestos Claim.

1.75. Lien means any charge against or interest in property to secure payment of a debt or performance of an obligation.

1.76. Net Reserve Funds means the lesser of (i) Cash in the amount to be disclosed in the Plan Supplement or (ii) the amount of Cash held by Hopeman as of the Effective Date, to be set aside by Hopeman on the Effective Date for the purpose of paying or making the Distributions contemplated by the Plan with respect to (a) Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Priority Claims, Allowed Secured Claims, and Allowed General

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 20 of 129

Unsecured Claims (including the payment of any interest on any such Claims that may be allowed under the Plan or required to be paid by the Bankruptcy Code), (b) any fees and expenses of Hopeman, Reorganized Hopeman, the Committee and the Future Claimants' Representative that are payable by Hopeman or Reorganized Hopeman, as applicable, pursuant to the Plan, (c) any fees payable pursuant to section 1930 of title 28 of the United States Code (whether those fees pursuant to section 1930 of title 28 are payable before or after the Effective Date, (d) any other amounts that the Plan provides are to be paid from the Net Reserve Funds, and (e) such amounts as Reorganized Hopeman determines, in the reasonable exercise of its discretion, are or will be sufficient to fully satisfy (as and when due) all franchise taxes and other expenditures that are necessary to maintain its corporate existence in good standing under the laws of the state of its formation or that otherwise are necessary for Reorganized Hopeman to conduct its business after the Effective Date.

1.77. Non-Asbestos Claim means any Claim against Hopeman that is not an Asbestos Personal Injury Claim.

1.78. Non-Asbestos Insurer means any Entity, including any insurance company, broker, or guaranty association, that has issued, or that has any actual or potential liabilities, duties or obligations under or with respect to any Non-Asbestos Insurance Policy.

1.79. Non-Asbestos Insurance Policy means any insurance policy that provides coverage to Hopeman that is not an Asbestos Insurance Policy.

1.80. Non-Settling Asbestos Insurer means an Asbestos Insurer that is not a Settled Asbestos Insurer. For the avoidance of doubt, notwithstanding any provision herein to the contrary, the term "Non-Settling Asbestos Insurer" shall include Liberty Mutual Insurance Company.

1.81. Person means person as defined in section 101(41) of the Bankruptcy Code.

1.82. Petition Date means June 30, 2024, the date on which the petition was filed by Hopeman pursuant to section 301 of the Bankruptcy Code to commence this Chapter 11 Case.

1.83. Plan means this plan of reorganization of Hopeman under chapter 11 of the Bankruptcy Code, including any supplements, schedules and exhibits hereto, either in its present form or as the same may be amended, modified or supplemented from time to time in accordance with the terms hereof.

1.84. Plan Documents means, collectively, (a) the Plan and all exhibits thereto, (b) the Disclosure Statement and all exhibits thereto, (c) the Plan Supplement, and (d) any other document necessary to implement the Plan.

1.85. Plan Proponents means Hopeman and the Committee.

1.86. Plan Supplement means the supplement to the Plan to be filed with the Bankruptcy Court no later than five (5) Business Days prior to the deadline for the filing and service of objections to the Plan or such later date as the Bankruptcy Court may approve, all of which are incorporated herein by reference.

1.87. Prepetition Non-Asbestos Claims Bar Dates means (a) November 4, 2024, which the Bankruptcy Court fixed as the deadline to file a Proof of Claim for any Claim arising before the Petition Date other than an Asbestos Personal Injury Claim and a Claim held by a Governmental Unit, and (b) December 27, 2024, which the Bankruptcy Court fixed as the deadline for Governmental Units to file a Proof of Claim for any Claim arising before the Petition Date other than an Asbestos Personal Injury Claim arising before the Petition Date other than an Asbestos Personal Injury Claim arising before the Petition Date other than an Asbestos Personal Injury Claim arising before the Petition Date other than an Asbestos Personal Injury Claim.

1.88. Priority Non-Tax Claim means any Claim entitled to priority in right of payment pursuant to section 507(a) of the Bankruptcy Code that is not an Administrative Expense Claim or a Priority Tax Claim.

1.89. Priority Tax Claim means any Claim entitled to priority pursuant to section 507(a)(8) of the Bankruptcy Code.

1.90. Proceeds means all cash, interest, profits, dividends, proceeds, products, and rents earned, accrued, collected, derived, received, or recovered on account of the liquidation, sale, transfer, enforcement, or other disposition of property, including all "proceeds" as defined under section 9-102(64) of the Uniform Commercial Code, and all proceeds on proceeds.

1.91. Pro Rata means the proportion that the face amount of a Claim in a particular Class or Classes bears to the aggregate face amount of all Claims (including Disputed Claims but excluding Disallowed Claims) in such Class or Classes, unless the Plan provides otherwise.

1.92. Professional means any person retained or to be compensated pursuant to section 327, 328, 330, 363, 503(b), or 1103 of the Bankruptcy Code.

1.93. Professional Fee Claim means any Claim of a (a) Professional for allowance of compensation and/or reimbursement of costs and expenses, and (b) member of the Committee for reimbursement of costs and expenses, in each case incurred in the Chapter 11 Case on or before the Effective Date.

1.94. Proof of Claim means any proof of claim or interest filed with the Bankruptcy Court or the Claims and Balloting Agent pursuant to section 501 of the Bankruptcy Code and Rule 3001 or 3002 of the Bankruptcy Rules that asserts a Claim against or Interest in Hopeman.

1.95. Protected Party shall mean each of the following persons or entities (collectively referred to herein as "**Protected Parties**"):

- (a) Hopeman or Reorganized Hopeman;
- (b) current and former directors, officers, or employees of Hopeman, or any past or present Affiliate of Hopeman, solely in their respective capacities as such; or
- (c) any Settled Asbestos Insurer, solely in its capacity as such.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 22 of 129

1.96. Released Parties means current and former directors, officers, or employees of Hopeman, or any past or present Affiliate of Hopeman, except Wayne, solely in their respective capacities as such.

1.97. Releasing Party means collectively: (a) all holders of <u>Asbestos</u> Claims that vote to accept or are presumed to accept the Plan;<u>and</u> (b) all holders of <u>Claims that abstain from</u> voting on the Plan and who do not<u>Non-Asbestos Claims or Equity Interests who</u> affirmatively opt out of<u>in to</u> the releases provided by the Plan by checking the box on the applicable form indicating that they opt <u>notin</u> to grant the releases provided in the Plan in accordance with the procedures set forth in the Solicitation Procedures Order; (c) all holders of Claims and Interests that vote to reject the Plan or are deemed to reject the Plan and who do not affirmatively opt out of the releases provided by the Plan by checking the box on the applicable form indicating that they opt not to grant the releases provided in the Plan in accordance with the procedures set forth in the Solicitation Procedures Order; and (d) with respect to Hopeman and each of the foregoing Entities in clauses (a) through (c), such Entity and its current and former affiliates, and such Entities' and their current and former affiliates' current and former directors, managers, officers, equity holders (regardless of whether such interests are held directly or indirectly), interest holders, predecessors, successors, and assigns.

1.98. Reorganized Hopeman means Hopeman on and after the Effective Date.

1.99. Reorganized Hopeman Common Stock means all of the issued and outstanding shares of common stock of Reorganized Hopeman to be deemed authorized and issued under Section 8.6.

1.100. Representative means, with respect to any specified Entity, any current or former officer, director, employee, agent, attorney, accountant, financial advisor, expert, consultant, or other representative of any specified Entity.

1.101. Restructuring Transactions means the acquisition by Hopeman or Reorganized Hopeman of the low-cost, income-generating business or interest in such business, which acquisition cost will be \$500,000 or less, described in <u>Exhibit F</u>.

1.102. Schedules means the schedules of assets and liabilities and the statements of financial affairs of Hopeman as filed with the Bankruptcy Court by Hopeman after the Petition Date in accordance with section 521 of the Bankruptcy Code and Bankruptcy Rule 1007, as such schedules and statements may be amended or supplemented from time to time.

1.103. Secured Claim means a Claim that is: (a) secured by a valid, duly perfected, nonavoidable security interest in the interest of Hopeman in property, to the extent of the value, as of the Effective Date or such other date as is established by the Bankruptcy Court, of such claimholder's interest in Hopeman's interest in such property, as determined by a Final Order of the Bankruptcy Court pursuant to section 506(a) of the Bankruptcy Code or as otherwise agreed in writing by Hopeman and the claimholder; or (b) secured by the amount of any valid, nonavoidable rights of setoff of the holder thereof under section 553 of the Bankruptcy Code.

1.104. Settled Asbestos Insurer means any Asbestos Insurer that is a party to an Asbestos Insurance Settlement.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 23 of 129

1.105. Solicitation Procedures Order means the order of the Bankruptcy Court approving the Disclosure Statement as containing adequate information within the meaning of section 1125 of the Bankruptcy Code, approving the combined hearing of the Plan and Disclosure Statement, and approving the method of solicitation of votes on the Plan. Hopeman anticipates filing a motion, contemporaneously with the filing of this Plan, seeking approval of the Solicitation Procedures Order.

1.106. Trust Transfer has the meaning ascribed to that term in <u>Section 8.13</u>.

1.107. Unimpaired means, when used with respect to a Claim or an Equity Interest, any Claim or Equity Interest that is not Impaired.

1.108. Uninsured Asbestos Claim means a Channeled Asbestos Claim (a) with a date of first exposure to asbestos or asbestos-containing products or things falling after January 1, 1985, or (b) for which no coverage under any Asbestos Insurance Policy is available due to settlement (including an Asbestos Insurance Settlement), exhaustion, or a final and non-appealable ruling on a coverage issue or defense.

1.109. United States Trustee means the United States Trustee appointed under section 581 of title 28 of the United States Code to serve in the Eastern District of Virginia.

1.110. United States Trustee Fees means the fees payable to the United States Trustee in accordance with 28 U.S.C. § 1930.

1.111. Vendor Released Parties means those Entities listed on <u>Exhibit G</u> hereto, each of which is a vendor, including professionals, that provided services to Hopeman prior to the Petition Date.

1.112. Wayne means Wayne Manufacturing Corporation, a dissolved company, and its predecessors.

1.113. Worker Compensation Claim means any Claim (a) for the benefits under a government-mandated workers' compensation system, which a past, present, or future employee of Hopeman or any predecessor of Hopeman is receiving, or may in the future have a right to receive, or (b) for reimbursement, contribution, subrogation, or indemnity brought by any insurer or Governmental Unit as a result of payments made to or for the benefit of such employees under such a system and the fees and expenses incurred under any insurance policies or laws or regulations covering such employee claims.

B. Interpretation; Application of Definitions and Rules of Construction.

Unless otherwise specified, all Article, schedule or exhibit references in the Plan are to the respective Article of or schedule or exhibit to the Plan or the Plan Supplement, as the same may be amended or modified from time to time. The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Plan as a whole and not to any particular Article, Section, subsection, paragraph, or clause. The word "will" is to be construed as having the same meaning and effect as the word "shall," and vice-versa. With respect to a distribution under the Plan, "on" a date means on or as soon as reasonably practical thereafter. A capitalized term used

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 24 of 129

but not defined herein shall have the meaning ascribed to that term in the Bankruptcy Code. Pronouns stated in the masculine, feminine, or neuter gender shall include the masculine, feminine, and neuter genders. The rules of construction contained in section 102 of the Bankruptcy Code shall apply to the construction of the Plan, except that, in addition to section 102(7), the plural includes the singular. The headings in the Plan are for convenience of reference only and shall not limit or otherwise affect the provisions hereof. Any immaterial effectuating provisions may be interpreted by the Plan Proponents in a manner that is consistent with the overall purpose and intent of the Plan, all without further notice or order of the Bankruptcy Court.

C. Reference to Hopeman or Reorganized Hopeman.

Except as otherwise specifically provided in the Plan to the contrary, references in the Plan to Hopeman or to Reorganized Hopeman shall mean Hopeman and Reorganized Hopeman, as applicable, to the extent the context requires.

D. Reference to Monetary Figures.

All references in the Plan to dollar amounts or other monetary figures refer to the currency of the United States of America, unless otherwise expressly provided.

ARTICLE II

ADMINISTRATIVE EXPENSE AND PRIORITY TAX CLAIMS

2.1. Administrative Expense Claims.

Filing Administrative Expense Claims. The holder of an Administrative (a) Expense Claim, other than (a) a Professional Fee Claim covered by Section 2.2, (b) a liability incurred and payable in the ordinary course of business by Hopeman after the Petition Date that has been paid by Hopeman, or (c) an Administrative Expense Claim that has been Allowed and/or paid in full on or before the Effective Date, must file and serve on Reorganized Hopeman a request for payment of such Administrative Expense Claim pursuant to section 503(a) of the Bankruptcy Code so that it is received no later than the Administrative Expense Claim Bar Date. Holders required to file and serve but who fail to file and serve a request for payment of Administrative Expense Claims by the Administrative Expense Claim Bar Date shall be forever barred from asserting such Administrative Expense Claims against Hopeman, Reorganized Hopeman, and their property, and such Administrative Expense Claims shall be deemed waived and released as of the Effective Date. Notwithstanding the foregoing, pursuant to section 503(b)(1)(D) of the Bankruptcy Code, no Governmental Unit shall be required to file a request for payment of any Administrative Expense Claim of a type described in sections 503(b)(1)(B) or 503(b)(1)(C) of the Bankruptcy Code as a condition to such Claim being Allowed.

(b) <u>Allowance of Administrative Expense Claims</u>. An Administrative Expense Claim, with respect to which a request for payment has been properly and timely filed pursuant to <u>Section 2.1(a)</u>, shall become an Allowed Administrative Expense Claim if no objection to such request is filed with the Bankruptcy Court and served on Reorganized Hopeman and the requesting party on or before the thirtieth (30th) calendar day after the Administrative Expense Claim Bar Date, as the same may be modified or extended from time to time by order of the Bankruptcy

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 25 of 129

Court. If an objection is timely filed, the Administrative Expense Claim shall become an Allowed Administrative Expense Claim only to the extent Allowed by a Final Order or as such Claim is settled, compromised, or otherwise resolved pursuant to <u>Article VII</u>.

Payment of Allowed Administrative Expense Claims. Except to the extent (c) that an Administrative Expense Claim already has been paid during the Chapter 11 Case or the holder of an Allowed Administrative Expense Claim agrees to less favorable treatment, and except as provided for Professional Fee Claims in Section 2.2, each holder of an Allowed Administrative Expense Claim against Hopeman shall receive, in full and complete settlement, release, and discharge of such Claim, Cash, to be paid from the Net Reserve Funds, equal to the unpaid amount of such Allowed Administrative Expense Claim on the latest of (i) the Effective Date or as soon thereafter as reasonably practicable; (ii) the first Business Day that is at least thirty (30) calendar days after the date on which such Administrative Expense Claim becomes Allowed; and (iii) such other date as may be agreed to by such holder and Reorganized Hopeman or as otherwise ordered by the Bankruptcy Court; provided, however, that Allowed Administrative Expense Claims (other than a Professional Fee Claim covered by Section 2.2) representing liabilities incurred in the ordinary course of business by Hopeman, as Debtor in Possession, may be paid by Hopeman, as applicable, in the ordinary course of business, consistent with past practice and in accordance with the terms and subject to the conditions of any agreements governing, instruments evidencing, or other documents relating to such transactions.

2.2. Professional Fee Claims. Each Professional requesting compensation pursuant to section(s) 327, 328, 330, 331, 363, 503(b), or 1103 of the Bankruptcy Code for services rendered in connection with the Chapter 11 Case before the Effective Date shall (a) file with the Bankruptcy Court, and serve on counsel to Reorganized Hopeman, an application for allowance of final compensation and reimbursement of expenses in the Chapter 11 Case on or before the date that is forty-five (45) calendar days after the Effective Date, and (b) after notice and a hearing in accordance with the procedures established by the Bankruptcy Code and the Bankruptcy Rules and any prior orders of the Bankruptcy Court in the Chapter 11 Case, be paid by Reorganized Hopeman, in Cash from the Net Reserve Funds, in such amounts as are Allowed by the Bankruptcy Court (i) no later than thirty (30) calendar days after the date upon which the order relating to any such Allowed Claim is entered or (ii) upon such other terms as may be mutually agreed upon between the holder of such an Allowed Claim and Reorganized Hopeman or as otherwise ordered by the Bankruptcy Court.

2.3. Priority Tax Claims. Except to the extent that the holder of an Allowed Priority Tax Claim has been paid by Hopeman prior to the Effective Date or agrees to less favorable treatment, each holder of an Allowed Priority Tax Claim shall receive, in full and complete settlement, release, and discharge of such Claim, Cash from the Net Reserve Funds in an amount equal to the unpaid portion of such Allowed Priority Tax Claim, on the latest of (i) the Effective Date, (ii) thirty (30) days after the date such Priority Tax Claim becomes an Allowed Claim, or as soon thereafter as is practicable, and (iii) the date such Allowed Priority Tax Claim becomes due and payable under applicable non-bankruptcy law.

ARTICLE III

CLASSIFICATION OF CLAIMS AND INTERESTS

Pursuant to section 1122 of the Bankruptcy Code, set forth below is a designation of classes of Claims against and Equity Interests in Hopeman.

3.1. Classification. The categories of Claims and Equity Interests listed below, other than Administrative Expense Claims (including Professional Fee Claims) and Priority Tax Claims, are classified for all purposes, including voting, confirmation, and distribution pursuant to the Plan, as follows:

Class	Designation	Impairment	Entitled to Vote
Class 1	Priority Non-Tax Claims	Unimpaired	No (presumed to accept)
Class 2	Secured Claims	Unimpaired	No (presumed to accept)
Class 3	General Unsecured Claims	Impaired	Yes
Class 4	Channeled Asbestos Claims	Impaired	Yes
Class 5	Equity Interests	Impaired	No (deemed to reject)

ARTICLE IV

TREATMENT OF CLASSIFIED CLAIMS AND INTERESTS

4.1. Class 1 – Priority Non-Tax Claims.

(a) <u>Classification</u>. Class 1 consists of all Priority Non-Tax Claims.

(b) <u>Treatment</u>. Except to the extent that the holder of an Allowed Priority Non-Tax Claim agrees to less favorable treatment, each holder of an Allowed Priority Non-Tax Claim shall receive, in full and complete settlement, release, and discharge of, and in exchange for, such Allowed Priority Non-Tax Claim, Cash to be paid from the Net Reserve Funds in an amount equal to the Allowed Amount of such Claim on the later of (i) the Effective Date, and (ii) the date on which such Claim becomes Allowed, or, in each case, as soon as reasonably practicable thereafter.

(c) <u>Impairment and Voting</u>. Class 1 is Unimpaired under the Plan. Each holder of a Priority Non-Tax Claim is conclusively presumed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code and, accordingly, is not entitled to vote to accept or reject the Plan.

4.2. Class 2 - Secured Claims.

(a) <u>Classification</u>. Class 2 consists of all Secured Claims.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 27 of 129

(b) <u>Treatment</u>. Except to the extent that the holder of an Allowed Secured Claim agrees to less favorable treatment, on the Effective Date or as soon as reasonably practicable thereafter, each holder of an Allowed Secured Claim shall receive, at the option of Reorganized Hopeman and in full and complete settlement, release, and discharge of, and in exchange for, such Claim (i) payment in full in Cash to be paid from the Net Reserve Funds; (ii) the collateral securing such Allowed Secured Claim; or (iii) other treatment rendering such Claim Unimpaired.

(c) <u>Impairment and Voting</u>. Class 2 is Unimpaired under the Plan. Each holder of a Secured Claim is conclusively presumed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code and, accordingly, is not entitled to vote to accept or reject the Plan.

4.3. Class 3 - General Unsecured Claims.

(a) <u>Classification</u>. Class 3 consists of all General Unsecured Claims.

(b) <u>Treatment</u>. Except to the extent that the holder of an Allowed General Unsecured Claim agrees to less favorable treatment, each holder of an Allowed General Unsecured Claim shall receive, in full and complete settlement, release, and discharge of, and in exchange for, such Allowed General Unsecured Claim, Cash in an amount equal to its Pro Rata share of the General Unsecured Recovery Pool on the later of (i) the Effective Date, and (ii) the date on which such Claim becomes Allowed, or, in each case, as soon as reasonably practicable thereafter. Solely for purposes of calculating Distributions to holders of Allowed General Unsecured Claims, on the Effective Date, all Disputed General Unsecured Claims will be treated as though they are Allowed in the amounts asserted or as estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code, and a reserve will be set aside for such Disputed General Unsecured Claims.

(c) <u>Impairment and Voting</u>. Class 3 is Impaired under the Plan. Each holder of a General Unsecured Claim is entitled to vote to accept or reject the Plan.

4.4. Class 4 – Channeled Asbestos Claims.

(a) <u>Classification</u>. Class 4 consists of all Channeled Asbestos Claims.

(b) <u>Treatment</u>. As of the Effective Date, liability for all Channeled Asbestos Claims shall automatically, and without further act, deed, or court order, be channeled exclusively to and assumed by the Asbestos Trust in accordance with, and to the extent set forth in, <u>Articles VIII</u> and <u>X</u> hereof, the applicable Plan Documents, and the Confirmation Order. All Channeled Asbestos Claims shall be treated pursuant to the terms of <u>Article VIII</u>, the Asbestos Trust Agreement, and the Asbestos Trust Distribution Procedures. Except as provided in <u>Article VIII</u>, pursuant to section 524(g) of the Bankruptcy Code, the Plan, and the Confirmation Order, Channeled Asbestos Claims shall be subject to the Asbestos Permanent Channeling Injunction.

(c) <u>Impairment and Voting</u>. Class 4 is Impaired under the Plan. Each holder of an Asbestos Claim is entitled to vote to accept or reject the Plan.

4.5. Class 5 – Equity Interests.

(a) <u>Classification</u>. Class 5 consists of all Equity Interests.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 28 of 129

(b) <u>Treatment</u>. On the Effective Date, the Equity Interests in Hopeman shall be cancelled, annulled, and extinguished.

(c) <u>Impairment and Voting</u>. Class 5 is Impaired under the Plan. Holders of Interests are deemed to reject the Plan pursuant to section 1126(g) of the Bankruptcy Code and, accordingly, are not entitled to vote to accept or reject the Plan.

ARTICLE V

DISTRIBUTIONS UNDER THE PLAN ON ACCOUNT OF CLAIMS OTHER THAN CHANNELED ASBESTOS CLAIMS

5.1. Distributions. Other than with respect to payments to be made on account of Channeled Asbestos Claims and Asbestos Trust Expenses from the Asbestos Trust, Reorganized Hopeman shall make all Distributions required to be made under the Plan as provided under this <u>Article V</u>. All distributions to be made by the Asbestos Trust shall be made in accordance with the terms of Asbestos Trust Agreement and the Asbestos Trust Distribution Procedures.

5.2. Record Date for Holders of Claims. Except as otherwise provided in a Final Order, the transferees of Claims that are transferred pursuant to Rule 3001 of the Bankruptcy Rules on or prior to the Distribution Record Date shall be treated as the holders of such Claims for all purposes, notwithstanding that any period provided by Rule 3001 for objecting to such transfer has not expired by the Distribution Record Date.

5.3. Postpetition Interest on Claims. Except as otherwise provided for in this Plan (including Section 4.2), the Plan Documents, or the Confirmation Order, or any contract, instrument, release, settlement, or other agreement entered into in connection with the Plan, or unless required by applicable bankruptcy law, interest accruing on or after the Petition Date on account of any Claim shall not be paid.

5.4. Means of Cash Payment. At the option of Hopeman or Reorganized Hopeman, as applicable, any Cash payment to be made hereunder may be made by a check or wire transfer or as otherwise required or provided in any applicable agreement.

5.5. Delivery of Distributions. All Distributions to any holder of an Allowed Claim shall be made at the address of such holder as set forth on (i) the Schedules filed with the Bankruptcy Court, (ii) a proof of claim filed by or on behalf of such holder in the Chapter 11 Case, if the address is different from the Schedules, or (iii) if neither (i) or (ii) applies, the books and records of Hopeman, unless Hopeman or Reorganized Hopeman has been notified in writing of a change of address.

If any holder's Distribution is returned as undeliverable, then no further Distributions to such holder shall be made unless and until Reorganized Hopeman is notified of such holder's then-current address, at which time all missed Distributions shall be made to such holder without interest. A Cash Distribution that is not claimed by the expiration of six (6) months from the date that such Distribution was made – along with any further Distributions withheld under this Section 5.5 – shall be deemed unclaimed property under section 347(b) of the Bankruptcy Code and shall be returned to the Net Reserve Funds, and the Claim of any holder to

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 29 of 129

such Distributions or any further Distributions shall be discharged and forever barred. Nothing contained in the Plan shall require Hopeman or Reorganized Hopeman to attempt to locate any holder of an Allowed Claim.

5.6. Time Bar to Cash Payments. Checks issued by Reorganized Hopeman in respect of Distributions on Allowed Claims shall be null and void if not presented for payment within ninety (90) days after the date of issuance thereof. Requests for reissuance of any check shall be made in writing to Reorganized Hopeman by the holder of the Allowed Claim to whom such check originally was issued on or before thirty (30) days after the expiration of the ninety (90) day period following the date of issuance of such check. All funds held on account of a check voided in accordance with this Section 5.6 shall be returned to the Net Reserve Funds, and the Claim of any holder to such Distributions shall be discharged and forever barred.

5.7. Prepayment of Claims. Except as otherwise provided in the Plan, the other Plan Documents, or the Confirmation Order, Reorganized Hopeman shall have the right to prepay, without penalty, all or any portion of an Allowed Claim at any time; provided, that any such prepayment shall not violate or otherwise prejudice the relative priorities among the Classes of Claims.

5.8. Minimum Distributions; Fractional Cents. Notwithstanding any other provision in the Plan to the contrary, no Distribution will be made to any holder of a General Unsecured Claim for which the Allowed Amount is less than \$50.00, and no payment of fractional cents will be made pursuant to the Plan. Whenever any payment of a fraction of a cent under the Plan would otherwise be required, the actual Distribution made will reflect a rounding of such fraction to the nearest whole penny (up or down), with fractions of more than half a penny being rounded up and fractions of half a penny or less being rounded down.

5.9. Setoff and Recoupment. Hopeman or Reorganized Hopeman (or the Asbestos Trust to the extent it pertains to a Channeled Asbestos Claim) may, but shall not be required to, set off or recoup against any Claim (for purposes of determining the Allowed Amount of such Claim on which Distribution shall be made), any claims of any nature whatsoever that Hopeman or Reorganized Hopeman may have against the holder of such Claim, and the failure to do so shall not constitute a waiver or release by Hopeman or Reorganized Hopeman of any such Claims that Hopeman may have against the holder of such Claim.

ARTICLE VI

TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

6.1. General Treatment. As of the Effective Date and except as otherwise provided in the Plan (including with respect to <u>Sections 6.2</u> and <u>Section 8.3</u>) or in any contract, instrument, release or other agreement or document entered into in connection with the Plan, all Executory Contracts to which Hopeman is a party are hereby rejected, except for any Executory Contract that (a) has previously been rejected pursuant to a Final Order of the Bankruptcy Court, (b) is the subject of a separate rejection motion filed by Hopeman under section 365 of the Bankruptcy Code before the Confirmation Date, (c) is an asbestosicy or Asbestos CIP Agreement, or (d) is a Non-

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 30 of 129

Asbestos Insurance Policy. The Confirmation Order shall constitute the Bankruptcy Court's approval of the rejection of the contracts and leases rejected hereby.

6.2. Asbestos Insurance Agreements. For the avoidance of doubt, none of the Asbestos Insurance Policies or Asbestos CIP Agreements are being rejected, altered, or otherwise modified pursuant to this Plan, and all parties' respective rights, duties, defenses, obligations, and liabilities thereunder are hereby preserved, except to the extent of an Asbestos Insurance Policy or Asbestos CIP Agreement that is the subject of and only to the extent contemplated by and provided for in an Asbestos Insurance Settlement and only to the extent approved pursuant to the entry of an order by the Bankruptcy Court or the District Court.

Furthermore, Hopeman does not believe that any of the Asbestos Insurance Policies or Asbestos CIP Agreements constitute Executory Contracts. To the extent any of the Asbestos Insurance Policies or Asbestos CIP Agreements are Executory Contracts, then, notwithstanding anything contained in the Plan to the contrary, the Plan will constitute a motion to assume any such Asbestos Insurance Policy or Asbestos CIP Agreement. Subject to the occurrence of the Effective Date, the entry of the Confirmation Order will constitute approval of such assumption pursuant to section 365(a) of the Bankruptcy Code and a finding by the Bankruptcy Court that each such assumption is in the best interest of Hopeman and its Estate. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments are required to cure any defaults of Hopeman existing as of the Confirmation Date with respect to each such Asbestos Insurance Policy or Asbestos CIP Agreement.

For purposes of clarity, while Hopeman also does not believe that any such agreements constitute Executory Contracts, Hopeman is not assuming any prepetition settlement agreement (or any related indemnity obligations thereunder) that do not currently provide rights in favor of Hopeman to continuing coverage or payment of insurance proceeds.

In addition, nothing in this Plan, any Plan Supplement, or any other document related to or made as an exhibit to the Plan is intended to or shall limit the right of any Asbestos Insurer or Non-Asbestos Insurer to assert any insurance coverage defense available under the applicable Asbestos Insurance Policy to any Channeled Asbestos Claim, or the applicable Non-Asbestos Insurance Policy to any non-Channeled Asbestos Claim, as appropriate, asserted against Hopeman or administered by the Asbestos Trust.

6.3. Bar to Rejection Damages. In the event that the rejection of an Executory Contract by Hopeman or Reorganized Hopeman pursuant to the Plan results in damages to the nondebtor party or parties to such Executory Contract, a claim for such damages shall be forever barred and shall not be enforceable against Hopeman, Reorganized Hopeman or any of their respective properties or interests in property, and the nondebtor party or parties to such Executory Contract shall be barred from receiving any Distribution under the Plan on account of such Claim, unless a Proof of Claim with respect to such damages is filed with the Bankruptcy Court and served upon counsel for Hopeman or Reorganized Hopeman, as applicable, on or before (a) if such Executory Contract is rejected pursuant to <u>Section 6.1</u> above, thirty (30) days after entry of the Confirmation Order; or (b) if such Executory Contract is rejected pursuant to reject such Executory Contract, thirty (30) days after entry of such order. The notice of the Effective Date to be delivered pursuant to

Bankruptcy Rules 2002 and 3020(c) shall set forth the bar date for filing rejection damages claims under Section 6.2 of the Plan and constitute notice thereof.

ARTICLE VII

PROCEDURES FOR RESOLVING AND TREATING DISPUTED CLAIMS OTHER THAN CHANNELED ASBESTOS CLAIMS AND PROFESSIONAL FEE CLAIMS

7.1. Disputed Claims. All Disputed Claims against Hopeman, other than Channeled Asbestos Claims and Professional Fee Claims, shall be subject to the provisions of this <u>Article VII</u>.

7.2. **Objection to Claims**. Except as otherwise provided in this Plan (including Section 4.2 and Section 4.3), Hopeman or Reorganized Hopeman, as the case may be, shall be entitled to file objections to Claims that have been brought in the Bankruptcy Court or should properly have been brought in the Bankruptcy Court but were brought in other forums (other than Channeled Asbestos Claims), on or before the date that is one hundred and twenty (120) days after the Effective Date (unless such day is not a Business Day, in which case such deadline shall be the next Business Day thereafter), as the same may be extended from time to time by the Bankruptcy Court. In addition, Hopeman or Reorganized Hopeman, as the case may be, may before the expiration of such period request that the Bankruptcy Court estimate any contingent or unliquidated Claim (not including any Channeled Asbestos Claim) for any reason pursuant to section 502(c) of the Bankruptcy Code, regardless of whether Hopeman previously objected to such Claim or whether the Bankruptcy Court has ruled on any such objection, and the Bankruptcy Court will retain jurisdiction to estimate such Claim at any time, including during the pendency of litigation concerning any objection to any Claim or of any appeal relating thereto. Hopeman and Reorganized Hopeman shall be authorized to settle, compromise, withdraw or litigate to judgment such objections without further approval of the Bankruptcy Court.

7.3. Payments and Distributions with Respect to Disputed Claims. Notwithstanding any other provision hereof, no payments or Distributions shall be made with respect to all or any portion of a Disputed Claim unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by Final Order and the Disputed Claim, or some portion thereof, has become an Allowed Claim.

ARTICLE VIII

MEANS FOR IMPLEMENTATION OF THE PLAN

8.1. Generally. On and after the Confirmation Date, the Plan Proponents shall be empowered and authorized to take or cause to be taken, prior to the Effective Date, all actions necessary to enable them to implement the provisions of the Plan, including the creation of the Asbestos Trust and the preparations for the transfer of the Asbestos Trust Assets to the Asbestos Trust. On and after the Effective Date, Reorganized Hopeman shall be governed pursuant to the Amended Certificate of Incorporation and Amended By-Laws.

8.2. Transactions on the Effective Date.

(a) The following shall be deemed for all purposes to have occurred simultaneously on the Effective Date:

(i) the establishment of the Asbestos Trust;

(ii) the making of the Asbestos Trust Contribution, notwithstanding that contribution of the Excess Net Reserve Funds may occur after the Effective Date; and

(iii) the vesting in the Asbestos Trust of the Asbestos Trust Assets, as more fully described in, and subject to the conditions set forth in, <u>Section 8.3</u> below.

(b) Also on the Effective Date, but after the occurrence of the events in each of <u>Section 8.2(a)(i)</u> through (a)(ii) hereof, the following events shall be deemed for all purposes to have occurred simultaneously:

(i) the effectiveness of the Amended Certificate of Incorporation and Amended By-Laws of Reorganized Hopeman;

(ii) the appointment of the individual(s) who will act as the officers and as the director of Reorganized Hopeman, as identified in a notice to be filed jointly by the Committee and the Future Claimants' Representative no later than two (2) days prior to the deadline established to accept or reject the Plan; and

(iii) any Distributions required to be made on the Effective Date (or as soon thereafter as is reasonably practicable).

(c) Unless the Plan or the Confirmation Order provide otherwise, actions required to be taken on the Effective Date or as soon thereafter as is reasonably practicable shall be deemed to have been made on the Effective Date.

8.3. The Asbestos Trust

(a) <u>Creation of the Asbestos Trust</u>. On the Effective Date, the Asbestos Trust shall be created in accordance with the Plan Documents, the Asbestos Trust Documents, and section 524(g) of the Bankruptcy Code. Subject to the provisions of the Plan, and in consideration of the transfer of the Asbestos Trust Assets to the Asbestos Trust, the Asbestos Trust shall assume all liabilities and responsibility for all Channeled Asbestos Claims, and, among other things, to: (1) direct the processing, liquidation, and payment of all compensable Channeled Asbestos Claims in accordance with this Plan, the Asbestos Trust Documents, and the Confirmation Order; (2) preserve, hold, manage, and maximize the assets of the Asbestos Trust for use in paying and satisfying Channeled Asbestos Claims; and (3) qualify at all times as a qualified settlement fund. The Asbestos Trust shall use the Asbestos Trust's assets and income to resolve Channeled Asbestos Claims in accordance with the Asbestos Trust Agreement and the Asbestos Trust Distribution Procedures in such a way that holders of Channeled Asbestos Claims are treated fairly, equitably, and reasonably in light of the finite assets available to satisfy such claims, and shall

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 33 of 129

otherwise comply in all respects with the requirements of a trust set forth in section 524(g)(2)(B) of the Bankruptcy Code. On the Effective Date, all right, title, and interest in and to the Asbestos Trust Assets, and any proceeds thereof, will be transferred to, and indefeasibly vested in, the Asbestos Trust, free and clear of all Claims, Demands, Equity Interests, Encumbrances, and other interests of any Entity, without any further action of the Bankruptcy Court or any Entity, but subject to Section 8.5 hereof and the remaining provisions of this Section 8.3.

(b) <u>Transfer of the Asbestos Insurance Rights</u>. On the Effective Date, by virtue of Confirmation, without further notice, action, or deed, the Asbestos Insurance Rights shall be automatically transferred to, and indefeasibly vested in, the Asbestos Trust, and the Asbestos Trust shall thereby become the estate representative pursuant to sections 1123(a)(5) and 1123(b)(3)(B) of the Bankruptcy Code, with the exclusive right to enforce any and all of the Asbestos Insurance Rights against any Entity, subject to the provisions of <u>Section 8.13</u> and <u>Section 8.15</u> hereof, and the Proceeds of the recoveries of any such Asbestos Insurance Rights shall be the property of, and shall be deposited in, the Asbestos Trust. The Asbestos Insurance Rights shall be indefeasibly vested in the Asbestos Trust free and clear of all Claims, Demands, Equity Interests, Encumbrances, and other interests of any Entity.

(c) <u>Authority of the Asbestos Trust</u>. As of the Effective Date, without any further action of the Bankruptcy Court or any Entity, except as otherwise expressly set forth in the Plan including, without limitation, the rights reserved to HII under <u>Section 8.15</u>, the Asbestos Trust shall be empowered to initiate, prosecute, enforce, sue on, defend, settle, compromise, and resolve (or decline to do any of the foregoing) all claims, rights, Causes of Action, suits and proceedings, whether in law or in equity, whether known or unknown, related to or arising from any asset, liability, or responsibility of the Asbestos Trust, including any actions arising from or related to the Asbestos Insurance Rights, in any court of competent jurisdiction consistent with applicable law.

(d) <u>Appointment of Asbestos Trustee</u>. On the Confirmation Date, the Bankruptcy Court shall appoint an individual designated by the <u>Asbestos Claimants</u> Committee and the Future Claimants' Representative to serve as the initial Asbestos Trustee. The designated individual will be identified in the Asbestos Trust Agreement and appointed pursuant to the Confirmation Order. Such appointment shall be effective as of the Effective Date. The individual to serve as the initial Asbestos Trustee will be identified in the Plan Supplement. All subsequent Asbestos Trustees shall be appointed in accordance with the terms of the Asbestos Trust Agreement. For purposes of performing the duties and fulfilling the obligations under the Asbestos Trust Agreement and the Plan, the Asbestos Trustee shall be deemed to be a party in interest within the meaning of section 1109(b) of the Bankruptcy Code.

(e) <u>Appointment of Delaware Trustee</u>. [Wilmington Trust, N.A.] has been selected by agreement of the <u>Asbestos Claimants</u> Committee and the Future Claimants' Representative to serve as the initial Delaware Trustee and will be identified in the Asbestos Trust Agreement and appointed pursuant to the Confirmation Order. All subsequent Delaware Trustees shall be appointed in accordance with the terms of the Asbestos Trust Agreement.

(f) <u>Appointment of Post-Effective Date Future Claimants' Representative</u>. On the Effective Date, the individual to be identified in the Plan Supplement shall be appointed,

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 34 of 129

pursuant to this Plan, the Confirmation Order, and the Asbestos Trust Agreement, to serve as the Future Claimants' Representative on and after the Effective Date. Such Future Claimants' Representative shall have the functions, duties, and rights provided in, and shall serve in accordance with, the Asbestos Trust Agreement. In addition to the foregoing, such Future Claimants' Representative also may, at her option, participate in any: (1) appeal of the Confirmation Order; (2) hearing on a Professional Fee Claim; and (3) adversary proceeding pending on the Effective Date to which the Future Claimants' Representative was a party. Successor Future Claimants' Representatives will be appointed as provided in the Asbestos Trust Agreement.

(g) <u>Appointment of Asbestos Trust Advisory Committee</u>. Not later than ten (10) calendar days prior to the Confirmation Hearing, the <u>Asbestos Claimants</u>-Committee shall nominate six (6) individuals to serve as the initial members of the Asbestos Trust Advisory Committee. The Confirmation Order shall constitute an order of the Bankruptcy Court appointing the initial members of the Asbestos Trust Advisory Committee. The Asbestos Trust Advisory Committee shall have the functions, duties, and rights provided in, and shall serve in accordance with, the Asbestos Trust Agreement. Successor members of the Asbestos Trust Advisory Committee will be appointed as provided in the Asbestos Trust Agreement.

Transfer of Claims and Demands to the Asbestos Trust. On the Effective (h) Date, in consideration for the property transferred to the Asbestos Trust, and except as provided in Section 8.12, Section 8.13, and Section 8.15 hereof, all Channeled Asbestos Claims shall be transferred and channeled to, and assumed by, the Asbestos Trust pursuant to the Asbestos Permanent Channeling Injunction, and shall be resolved, liquidated, and (if eligible for payment) paid in accordance with the Asbestos Trust Agreement, the Asbestos Trust Distribution Procedures, and any other Asbestos Trust Document. The Asbestos Trust shall have no liability for any Claims other than Channeled Asbestos Claims and Asbestos Trust Expenses, and no Claims other than Channeled Asbestos Claims and Asbestos Trust Expenses shall be transferred and channeled to, or assumed by, the Asbestos Trust. Notwithstanding the Asbestos Trust's assumption of liability and responsibility for Channeled Asbestos Claims, such assumption shall not itself operate or be construed as a release, accord and satisfaction, mutual rescission, or novation of Hopeman's obligations on account of such Claims for purposes of any Asbestos Insurance Rights solely to the extent of actions or suits against Reorganized Hopeman directly in accordance with Section 8.12 hereof (subject, however, to the discharge of any "personal liability" of Hopeman as that term is used in section 524(a) of the Bankruptcy Code and as provided in Article X hereof).

(i) <u>Transfer of Rights and Defenses Related to Channeled Asbestos Claims</u>. On the Effective Date, all claims, defenses, rights and Causes of Action of Hopeman arising from or related to Channeled Asbestos Claims shall be transferred and assigned to the Asbestos Trust. In accordance with section 1123(b) of the Bankruptcy Code, the Asbestos Trust shall retain and may enforce such claims, defenses, rights, and Causes of Action relating to Channeled Asbestos Claims in any court of competent jurisdiction against any Entity other than a Protected Party, and shall retain and may enforce all defenses and counterclaims to all Asbestos Claims or Demands asserted against the Asbestos Trust, including setoff, recoupment, and any rights under section 502(d) of the Bankruptcy Code. The Asbestos Trust shall be deemed to be the appointed

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 35 of 129

representative of Hopeman and Reorganized Hopeman, and may, pursue, litigate, compromise, and settle any rights, claims, or Causes of Action transferred to it, as appropriate.

(j) <u>Release</u>. As a condition to making any payment to a Channeled Asbestos Claimant, the Asbestos Trust shall obtain from that holder a release in the form of the Asbestos Personal Injury Claimant Release.

(k) <u>Consideration for Asbestos Permanent Channeling Injunction</u>. The assignment, transfer, and conveyance of the Asbestos Trust Assets to the Asbestos Trust on the Effective Date supports the imposition of the Asbestos Permanent Channeling Injunction in favor of all Protected Parties as of the Effective Date.

(1) <u>Books and Records</u>. On the Effective Date, Hopeman shall transfer to Reorganized Hopeman all of Hopeman's books and records (including electronic records) necessary for the Asbestos Trust to investigate and resolve Channeled Asbestos Claims in accordance with <u>Section 8.3</u> and <u>Section 8.16</u> of this Plan, the Asbestos Trust Agreement and Asbestos Trust Distribution Procedures, including the books and records presently stored in Hopeman's warehouse in Waynesboro, Virginia, and in or in storage near the offices of Hopeman's prepetition claims administrator Special Claim Services, Inc. Notwithstanding anything to the contrary herein, holders of Asbestos Personal Injury Claims may pursue and obtain information stored in Hopeman's books and records (including electronic records) through discovery to the full extent permitted by applicable law. For the avoidance of doubt, privileges belonging to Hopeman as of the Effective Date, and the Asbestos Trust's access to such books and records shall not result in the destruction or waiver of any applicable privileges pertaining to such books and records.

(m) <u>Institution and Maintenance of Legal and Other Proceedings</u>. From and after the Effective Date, the Asbestos Trust shall be empowered and entitled, in its sole and absolute discretion and at its own expense, to pursue, compromise, or settle all legal actions and other proceedings related to any asset, liability, or responsibility of the Asbestos Trust that is not released pursuant to the Plan.

(n) <u>Asbestos Trust Expenses</u>. The Asbestos Trust shall pay all Asbestos Trust Expenses from the Asbestos Trust Assets. The Plan Proponents, Hopeman's Estate, Reorganized Hopeman, the Protected Parties, and any of them, shall not have any obligation to pay any Asbestos Trust Expenses or any other liabilities of the Asbestos Trust. The Asbestos Trust shall promptly pay all Asbestos Trust Expenses incurred by Reorganized Hopeman for any liabilities, costs, or expenses as a result of taking any action on behalf of, and at the direction of, the Asbestos Trust.

(o) <u>Indemnification</u>. The Asbestos Trust shall, pursuant to the terms of the Asbestos Trust Agreement, indemnify and hold harmless the Protected Parties for any liability or alleged liability arising out of, or resulting from, or attributable to, a Channeled Asbestos Claim, including fines and penalties resulting from the Asbestos Trust's failure to comply with <u>Section 8.14</u> of the Plan or the Asbestos Trust Agreement. Indemnification claims arising under this <u>Section 8.3(o)</u> will not be subject to the Asbestos Trust Distribution Procedures.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 36 of 129

(p) <u>Investment Policy</u>. Pursuant to the Asbestos Trust Agreement, all monies held in the Asbestos Trust shall be invested, subject to the investment limitations and provisions enumerated in the Asbestos Trust Agreement and shall not be limited to the types of investments described in section 345 of the Bankruptcy Code.

(q) <u>Excess Asbestos Trust Assets</u>. To the extent there are any Asbestos Trust Assets remaining at such time as the Asbestos Trust is dissolved, such excess Asbestos Trust Assets shall be transferred to a charity or charities in which the Asbestos Trustee has no financial interest or other connection for such charitable purposes as the Asbestos Trustee, in his or her reasonable discretion, shall determine, provided that, if practicable, the charity or charities to which such excess Asbestos Trust Assets are transferred shall be related to the treatment of, research on, or the relief of suffering of individuals suffering from asbestos-related disorders.

(r) <u>Dissolution of Asbestos Trust</u>. Upon dissolution of the Asbestos Trust: (1) the Asbestos Trustee, members of the Asbestos Trust Advisory Committee and the Future Claimants' Representative shall be released and discharged from all further authority, duties, responsibilities, and obligations relating to and arising from and in connection with the Chapter 11 Case; and (2) the Asbestos Trust Advisory Committee shall be dissolved.

8.4. Amended Certificate of Incorporation and By-Laws. The Amended Certificate of Incorporation and Amended By-Laws shall contain such provisions as are necessary to satisfy the provisions of the Plan and, to the extent necessary, to prohibit the issuance of nonvoting equity securities as required by section 1123(a)(6) of the Bankruptcy Code. Notwithstanding the foregoing, the Amended Certificate of Incorporation and Amended By-Laws may be amended after the Effective Date as permitted by applicable law.

8.5. Net Reserve Funds. No later than the Effective Date, Hopeman or Reorganized Hopeman, as applicable, shall either establish a bank account or utilize an existing bank account of Hopeman for the purpose of holding only the Net Reserve Funds. Hopeman, Reorganized Hopeman, or their respective designees, as applicable, shall make any and all disbursements from the Net Reserve Funds that are provided in the Plan. The Net Reserve Funds shall be used for the sole purpose of paying or making Distributions in accordance with the Plan. Any Excess Net Reserve Funds shall be contributed by Reorganized Hopeman to the Asbestos Trust, and that contribution shall be part of the Asbestos Trust Contribution.

8.6. Reorganized Hopeman Common Stock. On the Effective Date, all the existing Equity Interests in Hopeman shall be cancelled, annulled, and extinguished, and 100% of the Reorganized Hopeman Common Stock shall be authorized and issued to the Asbestos Trust.

8.7. Corporate Governance of Reorganized Hopeman. On the Effective Date, (a) the current officers and directors of Hopeman shall be deemed to resign from their respective positions by operation of the Plan, and (b) the individual(s) identified in a notice to be filed jointly by the Committee and the Future Claimants' Representative no later than two (2) days prior to the deadline established to accept or reject the Plan shall be appointed to serve as the officers and as the director of Reorganized Hopeman.

8.8. Operations of Hopeman Between Confirmation and the Effective Date. Hopeman shall continue to operate as a debtor and Debtor-in-Possession during the period from the Confirmation Date through and until the Effective Date. After the Effective Date, Reorganized Hopeman shall conduct the business for which <u>Section 8.11</u> of the Plan provides, including fully satisfying (as and when due) all franchise taxes and other expenditures that are necessary to maintain its corporate existence in good standing under the laws of the state of its formation or that otherwise are necessary for Reorganized Hopeman to conduct the business for which <u>Section</u> <u>8.11</u> provides.

8.9. Corporate Action. All matters provided for under the Plan involving the corporate structure of Hopeman or Reorganized Hopeman, or any corporate action to be taken by, or required of Hopeman or Reorganized Hopeman, shall be deemed to have occurred and be effective as provided herein, and shall be authorized and approved in all respects (and, if taken prior to the Effective Date, shall be deemed ratified in all respects on the Effective Date), without any requirement for further action by the stockholders or directors of any such entities.

8.10. Restructuring Transactions. On or after the Confirmation Date, Hopeman or Reorganized Hopeman may take such actions as it determines to be necessary or appropriate to effectuate, implement, and consummate the Restructuring Transactions set forth on <u>Exhibit F</u>.

8.11. Continued Corporate Existence. Except as otherwise provided herein (and subject to the Restructuring Transactions), Hopeman will, as a reorganized debtor, continue to exist after the Effective Date as a separate corporate or other legal Entity, with all powers of a corporation or other legal Entity under applicable law and without prejudice to any right to alter or terminate such existence (whether by merger, dissolution or otherwise) under applicable state law. On and after the Effective Date, Reorganized Hopeman may operate its business and may use, acquire, and dispose of property and compromise or settle any Claims without supervision or approval of the Bankruptcy Court and free of any restrictions of the Bankruptcy Code or Bankruptcy Rules, other than those restrictions expressly imposed by the Plan or Confirmation Order.

8.12. Actions Against Reorganized Hopeman or Wayne to Obtain Benefits of Asbestos Insurance Coverage.

(a) Except as otherwise permitted under <u>Section 8.16</u>, on and after the Effective Date, a Channeled Asbestos Claimant shall have the right to initiate, commence, continue, or prosecute an action against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust), and, where permitted by applicable nonbankruptcy law, any Non-Settling Asbestos Insurer for Wayne, in a court of competent jurisdiction to obtain the benefit of Asbestos Insurance Coverage.

(b) If a Channeled Asbestos Claimant commences such an action on account of its Channeled Asbestos Claim, the complaint shall name Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or any Non-Settling Asbestos Insurer for Wayne as a defendant and shall be deemed by operation of law to be an action against Reorganized Hopeman or any Non-Settling Asbestos Insurer for Wayne, as applicable. Such an action may be filed in any court where Hopeman was subject to *in personam* jurisdiction as of the Petition Date or any

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 38 of 129

other court of competent jurisdiction. Any such action shall be served on the Asbestos Trust, which shall provide notice of such action, as appropriate, to all Non-Settling Asbestos Insurers. Notwithstanding the foregoing, Reorganized Hopeman, the Asbestos Trust, and Wayne shall have no obligation to answer, appear, or otherwise participate in the action in any respect other than as set forth in this Plan and as may be necessary to comply with applicable Asbestos Insurance Cooperation Obligations.

(c) Any liability of Reorganized Hopeman or Wayne to any Entity, including any Channeled Asbestos Claimant or Asbestos Insurer, that is based on, arises from, or is attributable to any action commenced under this <u>Section 8.12</u> shall be enforceable only against the Asbestos Insurance Coverage provided by the Non-Settling Asbestos Insurers and not against any other asset, including any other Asbestos Insurance Right, of the Asbestos Trust or Reorganized Hopeman.

8.13. Actions Against Non-Settling Asbestos Insurers to Obtain Benefits of Asbestos Insurance Coverage.

(a) Except as otherwise permitted under this <u>Section 8.13</u> or in <u>Section 8.15</u>, the Asbestos Trust shall have the exclusive right to pursue, monetize, settle, or otherwise obtain the benefit of the Asbestos Insurance Rights, including with respect to any unpaid insurance Proceeds applicable to a judgment or settlement obtained or entered into by a Channeled Asbestos Claimant in accordance with <u>Section 8.12</u> hereof.

(b) If a Channeled Asbestos Claimant has entered into an enforceable settlement agreement with a Non-Settling Asbestos Insurer pertaining to his Channeled Asbestos Claim and such Non-Settling Asbestos Insurer has not timely paid or has refused to pay the amount provided in such settlement, such Channeled Asbestos Claimant may commence a breach-of-contract action or other form of collection action against such Non-Settling Asbestos Insurer to recover the settlement payment owed.

(c) Any Channeled Asbestos Claimant who (1) has obtained a judgment against Reorganized Hopeman or Wayne in accordance with <u>Section 8.12</u> hereof, or (2) has the right under applicable nonbankruptcy law to name, join, or substitute as a defendant an Asbestos Insurer, may, to obtain the benefits of Asbestos Insurance Coverage, commence a judgment-enforcement action or a direct action against the relevant Non-Settling Asbestos Insurer ("**Insurance Policy Action**") in accordance with the terms of this <u>Section 8.13</u>, subject to the following conditions:

(i) If any Non-Settling Asbestos Insurer against whom an Insurance Policy Action is brought asserts as a defense that it would have a claim as a result of contribution rights against one or more Settled Asbestos Insurers with respect to the Channeled Asbestos Claimant's claim that it could have asserted but for the Asbestos Permanent Channeling Injunction ("Contribution Claim"), the liability, if any, of the Non-Settling Asbestos Insurer to the Channeled Asbestos Claimant shall be reduced dollar-for-dollar by the amount, if any, of any judgment establishing the Contribution Claim in accordance with this <u>Section 8.13</u>.

(ii) In determining the amount of any Contribution Claim that operates to reduce the liability of a Non-Settling Asbestos Insurer in any Insurance Policy Action, the

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 39 of 129

Channeled Asbestos Claimant may assert the legal or equitable rights or defenses, if any, of the Settled Asbestos Insurers with respect to such Contribution Claims, and for purposes of this <u>Section 8.13(c)(ii)</u>, all Settled Asbestos Insurers, in exchange for their status as a Protected Party and for receiving the benefits of the Asbestos Permanent Channeling Injunction, shall be deemed to have transferred or assigned such legal or equitable rights or defenses to Channeled Asbestos Claimants; *provided* that the Channeled Asbestos Claimant shall not be permitted to argue that any Contribution Claims are not properly asserted against the Channeled Asbestos Claimant or that the Asbestos Permanent Channeling Injunction Claims in connection with the Channeled Asbestos Claimant's claim against the Non-Settling Asbestos Insurer.

(iii) If a court reduces the amount of liability of a Non-Settling Asbestos Insurer in an Insurance Policy Action based on the share attributable to a Settled Asbestos Insurer consistent with this <u>Section 8.13</u>, the Channeled Asbestos Claimant whose judgment has been reduced may seek payment from the Asbestos Trust for all or a portion of the amount of the judgment reduction attributable to the Settled Asbestos Insurer's share, but only as permitted by and in accordance with the Asbestos Trust Distribution Procedures.

(iv) If the Asbestos Trust enters into an Asbestos Insurance Settlement with an Asbestos Insurer that is a party to a pending Insurance Policy Action, and if such Asbestos Insurance Settlement is approved by the Bankruptcy Court, the Channeled Asbestos Claimant pursuing such Insurance Policy Action shall (I) be deemed to release any rights transferred to it via the Trust Transfer, without further notice or action by any Entity, and (II) terminate its Insurance Policy Action with respect to that Asbestos Insurer. The Channeled Asbestos Claimant shall have the right to submit an Uninsured Asbestos Claim to the Asbestos Trust for payment in accordance with, and as permitted by, the Asbestos Trust Distribution Procedures, as set forth in <u>Section 8.16</u> hereof.

(v) The Asbestos Trust may seek to intervene in any Insurance Policy Action at any time. A Channeled Asbestos Claimant pursuing an Insurance Policy Action shall not object to or oppose any request or motion of the Asbestos Trust to intervene in such Insurance Policy Action. For purposes of establishing the grounds in favor of the Asbestos Trust's request or motion to intervene, the Confirmation Order shall include the following finding of the Bankruptcy Court: The Asbestos Trust shall have, and is deemed to have, an interest relating to the Asbestos Insurance Coverage that is the subject of any Insurance Policy Action, and shall be, and is deemed to be, so situated that disposing of the Insurance Policy Action may, as a practical matter, impair or impede the Asbestos Trust's ability to protect its interest, and no party to the Insurance Policy Action can adequately represent that interest.

(d) In addition to the rights and remedies set forth in this <u>Section 8.13</u>, on and after the Effective Date, Channeled Asbestos Claimants may, only to the extent permitted or provided under applicable nonbankruptcy law, bring such Insurance Policy Actions against a Non-Settling Asbestos Insurer of Hopeman or Wayne with respect to potential liability of any Designated Person, subject to the terms and conditions set forth in <u>Section 8.13(c)</u>. For the avoidance of doubt, no Designated Person shall be named as a defendant in any such Insurance Policy Actions.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 40 of 129

(e) If a Channeled Asbestos Claimant intends to pursue an Extracontractual Claim against a Non-Settling Asbestos Insurer, the Channeled Asbestos Claimant shall send written notice to the Asbestos Trust, requesting the Asbestos Trust's leave to pursue such Extracontractual Claim. Within fourteen (14) calendar days after receiving the Channeled Asbestos Claimant's request, the Asbestos Trust shall respond in writing that it is either granting or denying the Channeled Asbestos Claimant's request.

(i) The Asbestos Trust's notice to the Channeled Asbestos Claimant of its decision to grant the Channeled Asbestos Claimant's request to pursue an Extracontractual Claim shall legally operate as, or shall effect, to the extent permitted or authorized by applicable nonbankruptcy law and without further notice or action by any Entity, a transfer to the Channeled Asbestos Claimant of the Asbestos Trust's rights to pursue such Extracontractual Claim ("**Trust Transfer**"). The Channeled Asbestos Claimant shall thereupon have standing and authority to pursue, settle, or resolve such Extracontractual Claim within his or her absolute discretion, subject to the terms set forth in this <u>Section 8.13</u>.

(ii) The Asbestos Trust may deny a Channeled Asbestos Claimant's request for leave to pursue an Extracontractual Claim if (I) the Asbestos Trust has determined that it is not able or permitted under applicable nonbankruptcy law to transfer or assign the Extracontractual Claim to the requesting Channeled Asbestos Claimant; (II) the Asbestos Trust intends, within ninety (90) calendar days after receiving the Channeled Asbestos Claimant's request, to commence litigation (or arbitration, to the extent required by the applicable Asbestos Insurance Policy or Asbestos CIP Agreement) against the relevant Non-Settling Asbestos Insurer with respect to the Channeled Asbestos Claimant's claim; or (III) the Asbestos Trust has already commenced such litigation or arbitration, which is pending or has been resolved. Such litigation commenced by the Asbestos Trust may include, or pertain to, multiple other Channeled Asbestos Claims or seek a declaration of rights generally with respect to the Non-Settling Asbestos Insurers' obligations in connection with Channeled Asbestos Claims.

8.14. No Actions Against Released Parties. In any action pursued in accordance with <u>Section 8.12</u> or <u>Section 8.13</u> hereof, Channeled Asbestos Claimants shall not name any Released Party as a defendant, or promptly shall dismiss from such action any Released Party previously named as a defendant. The Asbestos Trust shall indemnify any Released Party to the extent such Released Party is named as a defendant in such an action, and the Trust shall take all steps reasonably necessary to have any Released Party named in such an action dismissed from such action, including seeking relief from the Bankruptcy Court to enforce the terms of the Plan.

8.15. Special Provision Pertaining to HII. Notwithstanding any provision of this Plan or any other Plan Document to the contrary, on and after the Effective Date, HII may, (a) to the extent it is the holder of one or more Asbestos Indirect Claims and (b) only to the extent permitted or provided under applicable nonbankruptcy law, file claims, crossclaims, or third-party demands in a court of competent jurisdiction (including in any state court lawsuits) against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or Wayne, or pursue direct actions against any Non-Settling Asbestos Insurer of Hopeman or Wayne, including to bring an

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 41 of 129

action against a Non-Settling Asbestos Insurer of Hopeman or Wayne with respect to the potential liability of any Designated Person. For the avoidance of doubt, such Designated Persons will not be named as defendants in any such actions. Reorganized Hopeman, the Asbestos Trust, and Wayne shall have no obligation to answer, reply, appear, or otherwise participate in any action in which HII has filed a claim, crossclaim, third-party demand, or in any such direct action, other than as necessary to maintain coverage under the Asbestos Insurance Policies. Any judgment that may be obtained in connection with such a claim, crossclaim, third-party demand, or direct action cannot be enforced against the assets of Reorganized Hopeman or the Asbestos Trust, other than from the Asbestos Insurance Coverage. To the extent Hopeman's Asbestos Insurance Rights become subject to Asbestos Insurance Settlements and such Asbestos Insurance Rights are liquidated by the Trust, HII may submit Asbestos Indirect Claims to the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures. For the avoidance of doubt, the foregoing in no way limits HII's ability to seek payment or any other form of relief that may be made available according to the provisions contained within the Asbestos Trust Distribution Procedures, this Plan, or any other Plan Document. For the further avoidance of doubt, HII need not execute an Asbestos Personal Injury Claimant Release to obtain the benefits of this Section 8.15. Notwithstanding any provision of the Asbestos Trust Distribution Procedures, this Plan or any other Plan Document to the contrary, the rights contained within this Section 8.15 may not be impaired, impeded, abridged, or otherwise modified at any time (whether before or after entry of the Confirmation Order) without the prior written consent of HII; provided, however, that the foregoing shall not be construed to vest HII with consent or veto rights as to a proposed Asbestos Insurance Settlement described in Section 8.17, or to override, expand, or otherwise modify HII's consultation rights granted under Section 8.17.

8.16. Payments or Distributions from the Asbestos Trust. The sole and exclusive source of payment or recovery of a Channeled Asbestos Claimant on account of his Channeled Asbestos Claim shall be the Asbestos Insurance Coverage applicable to such Channeled Asbestos Claim, as provided in Section 8.12, Section 8.13, and Section 8.15 above, unless the Channeled Asbestos Claim (a) is an Uninsured Asbestos Claim, or (b) becomes an Uninsured Asbestos Claim before the Channeled Asbestos Claimant receives payment in full of any judgment obtained against Reorganized Hopeman or the Non-Settling Asbestos Insurer, or settlement reached with a Non-Settling Asbestos Insurer, in accordance with Section 8.12, Section 8.13, or Section 8.15 above. A Channeled Asbestos Claim shall become an Uninsured Asbestos Claim when (i) the Asbestos Trust has settled, in accordance with an Asbestos Insurance Settlement, all rights to the Asbestos Insurance Coverage applicable to the Channeled Asbestos Claim, or (ii) any Asbestos Insurance Coverage that otherwise may be applicable to such Channeled Asbestos Claim becomes unavailable due to exhaustion of the relevant Asbestos Insurance Coverage or due to a Final Order ruling on a coverage issue or defense, in which event such Channeled Asbestos Claimant may seek payment or distribution on account of his Channeled Asbestos Claim from the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures. Any treatment of a Channeled Asbestos Claim that is partially an Insured Asbestos Claim and partially an Uninsured Asbestos Claim shall be set forth in the Asbestos Trust Distribution Procedures. Notwithstanding any provision in this Plan or any other Plan Document to the contrary, a Channeled Asbestos Claimant shall not be entitled to receive a payment or distribution on account of his Channeled Asbestos Claim from the Asbestos Trust unless and until

his Channeled Asbestos Claim is eligible for payment or distribution from the Asbestos Trust under the Asbestos Trust Distribution Procedures.

Settlements with Non-Settling Asbestos Insurers After the Effective Date. The 8.17. Asbestos Trust shall have authority to enter into an Asbestos Insurance Settlement with any Asbestos Insurer on or after the Effective Date, but such Asbestos Insurance Settlement shall not be valid or enforceable unless and until (a) the Asbestos Trust Advisory Committee and the Future Claimants' Representative have consented to such Asbestos Insurance Settlement and determined in writing that such Asbestos Insurance Settlement is fair, reasonable, and sufficiently comprehensive to warrant that such Asbestos Insurer receive the protections of a Settled Asbestos Insurer under the Asbestos Permanent Channeling Injunction, any other applicable Plan Injunction, and section 524(g) of the Bankruptcy Code; (b) the Asbestos Trust has consulted with HII about the terms of such Asbestos Insurance Settlement prior to its submission to the Bankruptcy Court for approval, as provided herein; and (c) the Bankruptcy Court approves such Asbestos Insurance Settlement after notice to all parties directly affected by such Asbestos Insurance Settlement and a hearing thereon. To the extent necessary, the Asbestos Trust shall be allowed to seek to reopen the Chapter 11 Case to obtain the Bankruptcy Court's approval of any such settlement for good cause shown in accordance with section 350(b) of the Bankruptcy Code.

8.18. Insurance Neutrality. Nothing in the Plan, the Plan Documents, the Confirmation Order, any finding of fact and/or conclusion of law with respect to the confirmation of the Plan, or any order or opinion entered on appeal from the Confirmation Order shall limit the right of any insurer to assert any coverage defense; provided, however, that (a) the transfer of rights in and under the Asbestos Insurance Rights to the Asbestos Trust is valid and enforceable and transfers such rights under the Asbestos Insurance Rights as Hopeman or Reorganized Hopeman may have, and that such transfer shall not affect the liability of any insurer, and (b) the discharge and release of Hopeman and Reorganized Hopeman from all Claims and the injunctive protection provided to Hopeman, Reorganized Hopeman, and the Protected Parties with respect to Claims as provided herein shall not affect the liability of any insurer, except to the extent that any such insurer is a Settled Asbestos Insurer. Notwithstanding anything in this Section 8.18 to the contrary, nothing in this Section 8.18 shall affect or limit, or be construed as affecting or limiting, (1) the binding effect of the Plan and the Confirmation Order on Hopeman, Reorganized Hopeman, the Asbestos Trust, or the beneficiaries of the Asbestos Trust or (2) the protection afforded to any Settled Asbestos Insurer by the Asbestos Permanent Channeling Injunction. Further, nothing in this Section 8.18 is intended or shall be construed to preclude otherwise applicable principles of res judicata or collateral estoppel from being applied against any insurer with respect to any issue that is actually litigated by such insurer as part of its objections to confirmation of the Plan.

8.19. Effectuating Documents; Further Transactions. Any officer, member or manager of or director of Hopeman or Reorganized Hopeman, as applicable, shall be, and hereby is, authorized to execute, deliver, file, and record such contracts, instruments, releases, indentures, certificates, and other agreements or documents, and take such other actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan. The Secretary or other appropriate officer of Hopeman is hereby authorized to certify or attest to any of the foregoing, if necessary.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 43 of 129

Hopeman and Reorganized Hopeman, and all other parties, including all holders of Claims entitled to receive Distributions under the Plan, shall execute any and all documents and instruments that must be executed under or in connection with the Plan in order to implement the terms of the Plan or to effectuate the Distributions under the Plan, *provided* that such documents and instruments are reasonably acceptable to such party or parties.

ARTICLE IX

EFFECT OF CONFIRMATION

9.1. Dissolution of Committee; Discharge of the Future Claimants' Representative. Effective on the Effective Date, the Committee shall be dissolved automatically, whereupon its members, Professionals, and agents shall be released and discharged from any further authority, duties, obligations and responsibilities in the Chapter 11 Case and under the Bankruptcy Code.

Effective as of the dissolution of the Committee, the Asbestos Trust Advisory Committee shall succeed to, and exclusively hold, the attorney-client privilege and any other privilege held by the Committee and shall enjoy the work product protections that were applicable or available to the Committee before its dissolution.

Effective on the Effective Date, the Future Claimants' Representative shall be discharged from her duties in such capacity, whereupon the Future Claimants' Representative and her Professionals and agents shall be released and discharged from any further authority, duties, obligations and responsibilities in the Chapter 11 Case and under the Bankruptcy Code.

9.2. Vesting of Assets. Pursuant to section 1141(b) of the Bankruptcy Code, except as otherwise provided in the Plan, the Plan Documents or the Confirmation Order, the property of the Estate of Hopeman (except for the Asbestos Trust Contribution and any other property of Hopeman distributed pursuant to the Plan) shall vest in Reorganized Hopeman on the Effective Date free and clear of any and all Liens, Claims, Encumbrances and other interests of any Entity. From and after the Effective Date, Reorganized Hopeman may operate its business and may use, acquire, and dispose of property free of any restrictions imposed under or by the Bankruptcy Code, the Bankruptcy Rules, or the Bankruptcy Court.

9.3. Preservation of Certain Causes of Action, Rights to Settle Claims and Compromise Controversies, and Defenses. With the exception of those claims and rights transferred to the Asbestos Trust, including Causes of Action, pursuant to Section 8.3 of this Plan or released pursuant to <u>Article X</u> of this Plan, in accordance with section 1123(b) of the Bankruptcy Code, Reorganized Hopeman, as successor in interest to Hopeman and its Estate, shall retain and may initiate, prosecute, enforce, sue on, settle, compromise, or resolve (or decline to do any of the foregoing), all claims, rights, causes of action, suits and proceedings, whether in law or in equity, whether known or unknown, accruing to, or that are property of, Hopeman and its Estate, pursuant to the Bankruptcy Code or any statute or legal theory, including any rights to, claims, or causes of action for, recovery under any policies of insurance issued to or on behalf of Hopeman other than the Asbestos Insurance Rights, in any court of competent jurisdiction consistent with applicable law.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 44 of 129

9.4. Terms of Injunction and Automatic Stay. With the exception of the stay provided in the Bankruptcy Court's *Third Interim Order Extending the Automatic Stay to Asbestos-Related Actions Against Non-Debtor Defendants* [Docket No. 622] that will (unless extended) expire on the earlier of June 30, 2025, or the Effective Date, all of the injunctions and/or stays in existence immediately prior to the Confirmation Date provided for in or in connection with the Chapter 11 Case, whether pursuant to section 105, 362, or any other provision of the Bankruptcy Code, the Bankruptcy Rules or other applicable law, shall remain in full force and effect until the injunctions set forth in the Plan become effective pursuant to a Final Order, and shall continue to remain in full force and effect thereafter as and to the extent provided by the Plan, the Confirmation Order, or by their own terms. For the avoidance of doubt, upon effectiveness of the injunctions set forth in the Plan, the automatic stay imposed by section 362 of the Bankruptcy Code shall be terminated. In addition, on and after the Confirmation Date, Reorganized Hopeman may seek such further orders as it may deem necessary or appropriate to preserve the status quo during the time between the Confirmation Date and the Effective Date.

Each of the injunctions contained in the Plan or the Confirmation Order shall become effective on the Effective Date and shall continue in effect thereafter unless otherwise provided in this Plan or the Confirmation Order.

9.5. No Liability for Certain Tax Claims. Unless a taxing authority has properly asserted a Claim against Hopeman on or before the Prepetition Non-Asbestos Claims Bar Dates, no Claim of that taxing authority shall be Allowed against Hopeman or Reorganized Hopeman for taxes, penalties, interest, additions to tax, or other charges arising out of the failure, if any, of Hopeman or any other Entity to have paid tax or to have filed any tax return (including, but not limited to, any income tax return or franchise tax return) in or for any prior year or arising out of an audit of any return for a period before the Petition Date.

9.6. No Successor Liability. Except as otherwise expressly provided in the Plan, Hopeman, Reorganized Hopeman, the other Protected Parties, and the Asbestos Trust do not, nor shall they be deemed to, assume, agree to perform, pay, or indemnify creditors for any liabilities or obligations of Hopeman relating to or arising out of the operations of, or assets of, Hopeman whether arising prior to or resulting from actions, events, or circumstances occurring or existing at any time prior to the Confirmation Date. None of Reorganized Hopeman, the other Protected Parties, or the Asbestos Trust is, or shall be, a successor to Hopeman by reason of any theory of law or equity, and none shall have any successor or transferee liability of any kind or character, except that Reorganized Hopeman and the Asbestos Trust shall assume the obligations specified expressly in the Plan and the Confirmation Order.

ARTICLE X

DISCHARGE, INJUNCTIONS, EXCULPATION, RELEASES, AND SETTLEMENT

10.1. <u>Discharge of Hopeman and Reorganized Hopeman</u>. Except as specifically provided in the Plan, any of the other Plan Documents, or the Confirmation Order, pursuant to sections 524 and 1141(d)(1)(A) of the Bankruptcy Code, confirmation of the Plan shall discharge Hopeman and Reorganized Hopeman on the Effective Date from any and all Claims and Demands of any nature whatsoever, including, without limitation, all Claims,

including, to the fullest extent permitted by law, Channeled Asbestos Claims, and liabilities that arose before the Confirmation Date and all debts of the kind specified in sections 502(g), 502(h) and 502(i) of the Bankruptcy Code whether or not: (a) a Proof of Claim based on such Claim was filed under section 501 of the Bankruptcy Code, or such Claim was listed on any of Hopeman's Schedules; (b) such Claim is or was allowed under section 502 of the Bankruptcy Code; or (c) the holder of such Claim has voted on or accepted the Plan. Except as otherwise specifically provided for in the Plan, as of the Effective Date, the rights provided in the Plan to holders of Claims, Demands and Equity Interests shall be in exchange for and in complete satisfaction, settlement and discharge of all Claims (including, to the fullest extent permitted by law, Asbestos Claims and Demands) against, Liens on, and Equity Interests in Hopeman, Reorganized Hopeman, and all of their respective assets and properties.

Hopeman Discharge Injunction. Except as specifically provided in the Plan 10.2. (including Section 8.12, Section 8.13, Section 8.15, and Section 8.16 hereof), any of the other Plan Documents, or the Confirmation Order, all Entities who have held, hold, or may hold Claims (including, to the fullest extent permitted by law, Asbestos Claims and Demands) against Hopeman are permanently enjoined, on and after the Effective Date, from: (a) commencing or continuing in any manner any action or other proceeding of any kind against Hopeman, Reorganized Hopeman, or their respective property with respect to such Claim or Demand, other than to enforce any right to a Distribution pursuant to the Plan or any other right provided under this Plan; (b) enforcing, attaching, collecting, or recovering by any manner or means of any judgment, award, decree, or order against Hopeman, Reorganized Hopeman, or their respective property with respect to such Claim or Demand; (c) creating, perfecting, or enforcing any Encumbrance of any kind against Hopeman, Reorganized Hopeman, or their respective property with respect to such Claim or Demand; (d) asserting any right of setoff, subrogation, or recoupment of any kind against any obligation due to Hopeman or against the property or interests in property of Hopeman, with respect to such Claim or Demand; and/or (e) commencing or continuing any action, in any manner, against Hopeman, Reorganized Hopeman, or their respective property that does not comply with or is inconsistent with the provisions of the Plan or the Confirmation Order. The foregoing injunction shall extend to the successors of Hopeman (including, without limitation, Reorganized Hopeman) and their respective properties and interests in property. The discharge provided in this provision shall void any judgment obtained against Hopeman at any time, to the extent that such judgment relates to a discharged Claim or Demand.

10.3. <u>Asbestos Permanent Channeling Injunction</u>. Pursuant to sections 105(a) and 524(g) of the Bankruptcy Code, and except as otherwise provided in the Plan (including <u>Article VIII</u> hereof), any of the other Plan Documents, and the Confirmation Order, the Confirmation Order shall provide for the issuance of the following injunction to take effect upon the occurrence of the Effective Date:

(a) <u>Scope of Injunction</u>. All Entities that have held or asserted, or hold or assert, or may hold or assert in the future any Channeled Asbestos Claim shall be permanently stayed, restrained, and enjoined from taking any action for the purpose of directly, indirectly, or derivatively collecting, recovering, or receiving payment, satisfaction, or recovery on account of any such Channeled Asbestos Claim, including:

(i) commencing, conducting, or continuing in any manner, directly, indirectly, or derivatively, any suit, action, or other proceeding of any kind (including a judicial, arbitration, administrative, or other proceeding) in any forum against or affecting any Protected Party or any property or interest in property of any Protected Party;

(ii) enforcing, levying, attaching (including any prejudgment attachment), collecting, or otherwise recovering by any means or in any manner, whether directly or indirectly, any judgment, award, decree, or other order against any Protected Party or any property or interest in property of any Protected Party;

(iii) creating, perfecting, or otherwise enforcing in any manner, directly or indirectly, any Encumbrance against any Protected Party or any property or interest in property of any Protected Party;

(iv) setting off, seeking reimbursement of, contribution from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability owed to any Protected Party or any property or interest in property of any Protected Party; and

(v) proceeding in any manner and in any place with regard to any matter that is within the scope of the matters designated by the Plan to be subject to resolution by the Asbestos Trust, except in conformity and compliance with the Asbestos Trust Agreement and the Asbestos Trust Distribution Procedures.

(b) <u>Reservations</u>. Notwithstanding anything to the contrary above, this Asbestos Permanent Channeling Injunction shall not stay, restrain, bar, or enjoin:

(i) the rights of Entities to the treatment accorded them under <u>Article IV</u> of this Plan, as applicable, including the rights of Channeled Asbestos Claimants to have their Channeled Asbestos Claims resolved in accordance with the Asbestos Trust Distribution Procedures;

(ii) the rights of Entities to assert any Channeled Asbestos Claim against the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures, or any claim for payment of an Asbestos Trust Expense against the Asbestos Trust;

(iii) the rights of Channeled Asbestos Claimants to assert any and all claims or causes of action against any Entities that are not Protected Parties;

(iv) the rights of the Asbestos Trust or, if applicable, Reorganized Hopeman to prosecute a claim or cause of action based on, arising from, or attributable to any of the Asbestos Trust Assets against any Entity that is not a Protected Party; (v) any action or suit against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or Wayne that strictly conforms to the requirements of <u>Section 8.12</u> hereof.

(vi) any action, suit, or Claimant Action permitted or authorized under <u>Section 8.13</u> hereof against any Non-Settling Asbestos Insurer; or

(vii) any claim, crossclaim, or third-party demand in a court of competent jurisdiction (including in any state court lawsuits) against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or Wayne, or any direct action against any Non-Settling Asbestos Insurer of Hopeman or Wayne, in each case that strictly conforms to the requirements of <u>Section 8.15</u> hereof.

10.4. Exculpation. NoneFrom and after the Petition Date through the Effective Date, and with the exception of claims related to any act or omission that is determined in a Final Order to have constituted gross negligence, knowing and intentional fraud, or willful misconduct, none of the Exculpated Parties shall have or incur any liability to any Entity for any act or omission in connection with, related to, or arising out of: (a) the Chapter 11 Case; (b) negotiation, formulation and preparation of the Plan and the other Plan Documents, and any of the terms and/or settlements and compromises reflected in the Plan and the other Plan Documents; (c) pursuit of confirmation of the Plan; (d) consummation of the Plan, or administration of the Plan or the property to be distributed under the Plan or the Asbestos Trust Distribution Procedures; <u>or</u> (e) the releases and injunctions contained in the Plan; or (f) the management or operation of Hopeman during the Chapter 11 Case. Without limiting the generality of the foregoing, the Exculpated Parties shall be entitled to and granted the protections and benefits of section 1125(e) of the Bankruptcy Code.

10.5. Releases by Hopeman and Estate. Except as otherwise expressly provided in the Plan or Confirmation Order, on the Effective Date, for good and valuable consideration, the adequacy of which is confirmed, Hopeman, Reorganized Hopeman, and any Entity seeking to exercise the rights of the Estate, in each case individually and collectively, including any successor to Hopeman or any Estate representative appointed or selected pursuant to the applicable provisions of the Bankruptcy Code, shall, and shall be deemed to, completely and forever release, waive and discharge unconditionally each of the Released Parties, solely in their capacities as such, from any and all claims, obligations, suits, judgments, remedies, damages, Demands, debts, rights, Causes of Action and liabilities that are (or that, at any time on or prior to the Effective Date, were) property of the Estate, whether known or unknown, liquidated or unliquidated, fixed or contingent, foreseen or unforeseen, matured or unmatured, now existing or hereafter arising, in law, equity or otherwise, whether direct, indirect, or derivative, based upon, attributable to, or arising out of, in whole or in part, any act or omission, transaction, or occurrence taking place on or prior to the Effective Date (including prior to the Petition Date); provided, however, that nothing contained in this Section 10.5 is intended to operate as a release of liability based upon gross negligence or willful misconduct as determined by a Final Order.

10.6. <u>Release of Hopeman's Directors and Officers Mutual Releases by Releasing</u> <u>Parties in Addition to Protections Under the Asbestos Permanent Channeling Injunction</u>. Except as provided in <u>Section 8.13(c)</u> and <u>Section 8.15</u> of <u>thisthe</u> Plan and in addition to the protections afforded to the Released Parties as Protected Parties under the Asbestos Permanent Channeling Injunction, and in consideration for the obligations of Hopeman and Reorganized Hopeman under the Plan, each Releasing Party and the Asbestos Trust shall waive and release any and all Causes of Action that such holder, did commence or could have commenced against any such Released Party that is based upon, attributable to, or arising from any acts or omissions of Released Party occurring prior to the Effective Date or in any way attributable to Hopeman, Reorganized Hopeman, the Chapter 11 Case, or the Plan; provided, however, that, notwithstanding the foregoing, no release of the Released Parties shall diminish, reduce, or eliminate the duties of any Asbestos Insurer under any Asbestos Insurance Policy or any Asbestos CIP Agreement.

In exchange for holders of Non-Asbestos Claims or Equity Interests opting in to grant the release provided in this section 10.6 of the Plan, each Released Party shall waive and release any and all Causes of Action that such Released Party did commence or could have commenced against any such holders of Non-Asbestos Claims or Equity Interests that optedin that is based upon, attributable to, or arising from any acts or omissions of Releasing Parties occurring prior to the Effective Date in any way attributable to Hopeman, Reorganized Hopeman, the Chapter 11 Case, or the Plan; provided, however, that, notwithstanding the foregoing, no Released Party is waiving or releasing any rights or claims under any insurance policies that provide any type of coverage to such Released Party.

10.7. <u>Release of Avoidance Actions and Estate Causes of Action Against Vendor</u> <u>Released Parties</u>. As of the Effective Date, each of Hopeman, Reorganized Hopeman and the Asbestos Trust hereby fully, finally, and forever releases, relinquishes, and discharges all Avoidance Actions and all other Claims and Causes of Action that it holds against any Vendor Released Parties.

10.8. <u>No Actions on Account of Released Claims</u>. Except as provided in the Plan, as of the Effective Date, all <u>EntitiesReleasing Parties</u> that have held, currently hold or may hold any claims, commitments, obligations, suits, judgments, damages, demands, debts, Causes of Action or liabilities that are released pursuant to the Plan shall be permanently enjoined from taking any of the following against a <u>released EntityReleased Party</u>, or any of its property, on account of such released claims, commitments, obligations, suits, judgments, damages, demands, debts, Causes of Action or liabilities: (i) commencing or continuing in any manner any action or other proceeding; (ii) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree, or order; (iii) creating, perfecting or enforcing any Encumbrance; (iv) asserting a setoff, right of subrogation or recoupment of any kind against any debt, liability or obligation due to any <u>released EntityReleased Party</u>; and (v) commencing or continuing any action, in any manner, in any place that does not comply with or is inconsistent with the provisions of this Plan.

10.9. Compromise and Settlement of Claims and Interests and Approval of Certain Settling Insurers Agreement. Pursuant to section 1123 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration of the Distributions and other benefits provided under the Plan,

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 49 of 129

the provisions of the Plan shall, on the Effective Date, constitute a compromise and settlement of all claims and controversies relating to the rights that a holder of a Claim or Interest may have against Hopeman with respect to any Claim, Interest, or Distribution on account thereof, as well as all potential Causes of Action against Hopeman. To the extent necessary, the entry of the Confirmation Order shall constitute approval of the Certain Settling Insurers Agreement and the settlement contemplated thereby. The entry of the Confirmation Order will constitute the Bankruptcy Court's approval of such compromises and settlements and the Bankruptcy Court's finding that such compromises and settlements are (i) in the best interest of Hopeman, the Estate, and holders of Claims and Interests; and (ii) fair, equitable, and reasonable. Hopeman or Reorganized Hopeman, as applicable, is authorized to take all actions necessary or appropriate to consummate the transactions and settlements contemplated thereby, including by the Certain Settling Insurers Settlement.

10.10. Disallowed Claims. On and after the Effective Date, Hopeman and the Estate shall be fully and finally discharged from any liability or obligation on a Disallowed Claim, and any order creating a Disallowed Claim that is not a Final Order as of the Effective Date solely because of an Entity's right to move for reconsideration of such order pursuant to section 502 of the Bankruptcy Code or Bankruptcy Rule 3008 shall nevertheless become and be deemed to be a Final Order on the Effective Date. The Confirmation Order, except as otherwise provided therein or herein, shall constitute an order: (a) disallowing all Claims (other than Channeled Asbestos Claims) to the extent such Claims are not allowable under any provision of section 502 of the Bankruptcy Code, including, but not limited to, time-barred Claims, and Claims for unmatured interest, and (b) disallowing or subordinating, as the case may be, any Claims, or portions of Claims, for penalties or non-compensatory damages.

ARTICLE XI

CONDITIONS PRECEDENT TO CONFIRMATION AND CONSUMMATION OF THE PLAN

11.1. Conditions Precedent to Confirmation of the Plan. The following are conditions precedent to confirmation of the Plan that (i) are designed, among other things, to ensure that the injunctions, releases, and discharges set forth in <u>Article X</u> shall be effective, binding, and enforceable, and (ii) must be satisfied, unless waived in accordance with <u>Section 11.3</u> below:

(a) the Plan and each of the other Plan Documents shall be in form and substance acceptable in all respects to each of the Plan Proponents;

(b) the Bankruptcy Court shall have entered an order, acceptable in form and substance to each of the Plan Proponents, approving the Disclosure Statement as containing adequate information within the meaning of section 1125 of the Bankruptcy Code;

(c) at least two-thirds (2/3) in amount and seventy-five percent (75%) in number of holders of Asbestos Personal Injury Claims actually voting on the Plan shall have voted in favor of the Plan;

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 50 of 129

(d) the Confirmation Order shall be in form and substance acceptable to each of the Plan Proponents and shall have been entered by (i) the Bankruptcy Court and the District Court acting jointly, (ii) the District Court, or (iii) the Bankruptcy Court and affirmed by the District Court;

(e) (i) the Plan, each of the other Plan Documents, and the Confirmation Order shall be in form and substance acceptable to HII only to the extent that the Plan, each of the other Plan Documents, and the Confirmation Order may affect and/or alter HII's rights under the terms set forth in <u>Section 8.15</u> of this Plan (such acceptance not to be unreasonably withheld), and (ii) the Plan Proponents shall have consulted with HII regarding the final forms of the Plan, each of the other Plan Documents, and the Confirmation Order in all other respects;

(f) the Confirmation Order shall, among other things:

(i) issue the Asbestos Permanent Channeling Injunction in accordance with the Plan and section 524(g) of the Bankruptcy Code

(ii) provide that the Confirmation Order supersedes any Bankruptcy Court orders issued prior to the Confirmation Date that may be inconsistent with the Confirmation Order;

(iii) provide that, except with respect to obligations specifically preserved in the Plan, Hopeman is discharged effective on the Effective Date (in accordance with the Plan and section 1141(d) of the Bankruptcy Code) from any Claims (including, to the fullest extent permitted by law, Asbestos Claims and Demands);

(iv) authorize the implementation of the Plan in accordance with its terms;

(v) provide that, as of the Effective Date, all Equity Interests are cancelled, annulled, and extinguished and that the Reorganized Hopeman Common Stock is issued to, and received by, the Asbestos Trust;

(vi) authorize the making of the Asbestos Trust Contribution, which shall be, on the Effective Date and without further notice, action, or deed, transferred to and indefeasibly vested in the Asbestos Trust free and clear of all Claims and Encumbrances against or on such Asbestos Trust Contribution;

(vii) provide that all transfers of assets of Hopeman contemplated under the Plan shall be free and clear of all Claims and Encumbrances against or on such assets;

(viii) except as otherwise provided in the Plan or Confirmation Order, provide that the assets revesting in Reorganized Hopeman shall be free and clear of all Claims and Encumbrances; (ix) provide that any transfers effected or entered into, or to be effected or entered into, under the Plan shall be and are exempt under section 1146(a) of the Bankruptcy Code from any state, city, or other municipal transfer taxes, mortgage recording taxes and any other stamp or similar tax;

(x) provide that the transfers of property by Hopeman to Reorganized Hopeman (A) are or will be legal, valid, and effective transfers of property; (B) vest or will vest Reorganized Hopeman with good title to such property; (C) do not and will not constitute avoidable transfers under the Bankruptcy Code or under other applicable bankruptcy or non-bankruptcy law; and (D) do not and will not subject Reorganized Hopeman to any liability by reason of such transfer under the Bankruptcy Code or applicable nonbankruptcy law, including any laws affecting or effecting successor or transferee liability;

(xi) provide that all Executory Contracts assumed or assumed and assigned by Hopeman during the Chapter 11 Case or under the Plan, if any, shall remain in full force and effect for the benefit of Reorganized Hopeman or the assignee thereof, notwithstanding any provision in such contract (including those provisions described in sections 365(b)(2) and (f) of the Bankruptcy Code) that prohibits such assignment or transfer or that enables or requires termination of such contract or lease;

(xii) require that as a condition to receiving any distributions of any kind from the Asbestos Trust, each holder of an Asbestos Personal Injury Claim execute the Asbestos Personal Injury Claimant Release;

(xiii) approve in all respects the other settlements, transactions, and agreements to be effected pursuant to the Plan, including the Asbestos Trust Agreement, the Asbestos Trust Distribution Procedures, and the other Asbestos Trust Documents, and the releases herein of the Released Parties;

(xiv) authorize the Plan Proponents and the Future Claimants' Representative to take all actions necessary or appropriate to effectuate, implement, and consummate the Plan, including the execution, delivery, and performance of contracts, instruments, releases, and other agreements or documents created in connection with the Plan; and

(xv) incorporate by reference all previously issued orders approving an Asbestos Insurance Settlement and provide that nothing in the Plan or the Confirmation Order is intended to modify or limit such orders.

(g) In addition to the foregoing, the Confirmation Order shall contain the following findings of fact and conclusions of law, among others:

(i) the Plan complies with all applicable provisions of the Bankruptcy Code, including, without limitation, those requiring that the Plan was proposed in good faith and that the Confirmation Order was not procured by fraud;

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 52 of 129

(ii) the Plan and its acceptance otherwise comply with sections 524(g) and 1126 of the Bankruptcy Code, and confirmation of the Plan is in the best interests of all creditors;

(iii) the Plan does not provide for the liquidation of all or substantially all of the property of Hopeman, and Reorganized Hopeman will continue in business as a reorganized debtor, and confirmation of the Plan is not likely to be followed by the liquidation of Reorganized Hopeman or the need for further financial reorganization;

(iv) the Plan Documents comply with section 524(g) of the Bankruptcy Code and therefore satisfy the prerequisites thereunder for issuing the Asbestos Permanent Channeling Injunction;

(v) as of the Petition Date, Hopeman had been named as a defendant in personal injury, wrongful death, or property damage actions seeking recovery for damages allegedly caused by the presence of, or exposure to, asbestos or asbestoscontaining products;

(vi) the Asbestos Trust, as of the Effective Date, will assume all the liabilities for Hopeman and Reorganized Hopeman for all Channeled Asbestos Claims, except as provided in <u>Section 8.12</u>, <u>Section 8.13</u>, and <u>Section 8.15</u> of the Plan;

(vii) the Asbestos Trust will to be funded by Reorganized Hopeman's contribution of the Asbestos Trust Assets, including the Asbestos Trust Contribution and the Asbestos Insurance Rights;

(viii) the assignment of the Asbestos Insurance Rights is valid and enforceable under sections 524(g), 541(e), 1123(a)(5)(B), and 1129(a)(1) of the Bankruptcy Code, and the Bankruptcy Code preempts any anti-assignment contractual provisions and applicable state law.

(ix) the Asbestos Trust will on the Effective Date own one hundred percent (100%) of the Reorganized Hopeman Common Stock and all rights to receive dividends or other distributions on account of such stock;

(x) the Asbestos Trust is to use its assets and income to pay eligible Channeled Asbestos Claims and Asbestos Trust Expenses, in accordance with the Asbestos Trust Documents, except as provided in <u>Section 8.12</u>, <u>Section 8.13</u>, and <u>Section 8.15</u> of the Plan;

(xi) Hopeman is likely to be subject to substantial future Demands for payment arising out of the same or similar conduct or events that gave rise to the Claims that are addressed by the Asbestos Permanent Channeling Injunction, and all such Demands are subject to the Asbestos Permanent Channeling Injunction;

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 53 of 129

(xii) the actual amounts, numbers, and timing of such Demands cannot be determined;

(xiii) pursuit of such Demands outside the procedures prescribed by the Plan and the Asbestos Trust Distribution Procedures is likely to threaten the Plan's purpose to deal equitably with Asbestos Claims and Demands;

(xiv) the Asbestos Permanent Channeling Injunction is essential to the Plan and Hopeman's reorganization efforts;

(xv) the terms of the Asbestos Permanent Channeling Injunction, including any provisions barring actions against third parties pursuant to section 524(g)(4)(A) of the Bankruptcy Code, are set out in the Plan and Disclosure Statement;

(xvi) the Plan establishes in Class 4 (Channeled Asbestos Claims) a separate class of the claimants whose Claims are to be addressed by the Asbestos Trust;

(xvii) Class 4 (Channeled Asbestos Claims) has voted, by at least seventyfive percent (75%) of the holders of Asbestos Personal Injury Claims and Asbestos Indirect Claims voting on the Plan, in favor of the Plan;

(xviii) Class 4 (Channeled Asbestos Claims) has voted, by at least twothirds (2/3) of the allowed amount of Class 4 Claims, to accept the Plan;

(xix) the Future Claimants' Representative was appointed as part of the proceedings leading to the issuance of the Asbestos Permanent Channeling Injunction for the purpose of protecting the rights of persons that might subsequently assert unknown Asbestos Claims and Demands that are addressed in the Asbestos Permanent Channeling Injunction and channeled to the Asbestos Trust;

(xx) each Protected Party is identifiable from the terms of the Asbestos Permanent Channeling Injunction by name or as part of an identifiable group;

(xxi) applying the Asbestos Permanent Channeling Injunction to each Protected Party is fair and equitable with respect to persons that might subsequently assert Demands against each such Protected Party, in light of the benefits provided, or to be provided, to the Asbestos Trust, including the Asbestos Trust Contribution, by or on behalf of any such Protected Party;

(xxii) the Asbestos Trust Contribution constitutes a sufficient basis upon which to provide the Protected Parties with the protections afforded to them under the Plan, the Plan Documents, and the Confirmation Order;

(xxiii) pursuant to courts orders, the Asbestos Trust Distribution Procedures, or otherwise, the Asbestos Trust will operate through mechanisms,

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 54 of 129

such as structured, periodic, or supplemental payments, pro rata distributions, matrices, or periodic review of estimates of the numbers and values of Asbestos Claims and Demands, or other comparable mechanisms, that provide reasonable assurance that the Asbestos Trust will value, and be in a financial position to pay, eligible Asbestos Claims and Demands that involve similar Claims in substantially the same manner;

(xxiv) the Plan and the other Plan Documents are a fair, equitable, and reasonable resolution of the liabilities of Hopeman for Channeled Asbestos Claims;

(xxv) the Future Claimants' Representative has adequately and completely fulfilled her duties, responsibilities, and obligations as the legal representative for unknown Demand holders in accordance with section 524(g) of the Bankruptcy Code;

(xxvi) adequate and sufficient notice of the Plan and the Confirmation Hearing, as well as all deadlines for objecting to the Plan, was given to (A) all known creditors and holders of Equity Interests, (B) parties that requested notice in accordance with Bankruptcy Rule 2002 (including the Committee and the Future Claimants' Representative), (C) all parties to Executory Contracts, (D) all taxing authorities listed on Hopeman's Schedules, (E) the United States Trustee, and (F) all Asbestos Insurers, in accordance with the solicitation procedures governing such service and in substantial compliance with Bankruptcy Rules 2002(b), 3017, and 3020(b); and such transmittal and service were adequate and sufficient to bind, among other parties, any holder of an Asbestos Claim, and no other or further notice is or shall be required; and

(xxvii) Hopeman's conduct in connection with and throughout the Chapter 11 Case, including its negotiations with the Committee and the Future Claimants' Representative, Hopeman's commencement of this Chapter 11 Case, and the drafting, negotiation, proposing, confirmation, and consummation of the Plan, does not and has not violated any Asbestos Insurance Cooperation Obligations, nor were such events or conduct a breach of any express or implied covenant of good faith and fair dealing.

11.2. Conditions Precedent to the Effective Date. The following are conditions precedent to occurrence of the Effective Date of the Plan that must be satisfied, unless waived in accordance with <u>Section 11.3</u> below:

(a) all conditions precedent to the Confirmation Date shall have been satisfied or waived and shall continue to be satisfied or waived;

Order;

(b) the Confirmation Order shall have been entered and shall have become a Final

(c) the Bankruptcy Court or the District Court, as required, shall have entered or affirmed the Asbestos Permanent Channeling Injunction (which may be included in the Confirmation Order that shall contain terms satisfactory and acceptable to each of the Plan Proponents);

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 55 of 129

(d) no request for revocation of the Confirmation Order under section 1144 of the Bankruptcy Code shall have been made or, if made, shall be pending;

(e) no fact or circumstance shall exist that would prevent the Asbestos Permanent Channeling Injunction from coming into full force and effect immediately upon the occurrence of the Effective Date;

(f) no fact or circumstance shall be preventing the Asbestos Trust from receiving the Asbestos Trust Contribution upon occurrence of the Effective Date;

(g) all Plan Documents and related agreements or instruments that require execution shall have been executed and delivered, or held in escrow to be released therefrom and promptly delivered to the applicable parties upon the occurrence of the Effective Date;

(h) certificates representing the Reorganized Hopeman Common Stock shall have been executed and delivered, or held in escrow to be released therefrom and promptly delivered to the Asbestos Trust upon the occurrence of the Effective Date;

(i) each of the Plan Documents and agreements or instruments contemplated by the Plan Documents shall be in form and substance acceptable to each of the Plan Proponents in all respects and, upon the occurrence of the Effective Date, shall be fully enforceable in accordance with their terms;

(j) all other actions, documents, and agreements that the Plan Proponents deem necessary to implement those provisions of the Plan to be effectuated on or prior to the Effective Date, in form and substance satisfactory to the Plan Proponents, shall have been effected or executed and delivered; and

(k) all conditions to closing set forth in any of the Plan Documents shall have been fulfilled to the reasonable satisfaction of the Plan Proponents.

11.3. Waiver of Conditions Precedent. To the fullest extent practicable and legally permissible, any of the conditions precedent set forth in <u>Section 11.1</u> or <u>Section 11.2</u> above may be waived or modified, in whole or in part, by the Plan Proponents, acting jointly; *provided, however*, that the condition precedent set forth in <u>Section 11.1(e)</u> may not be waived or modified, in whole or in part, without the written consent of HII. Any such waiver or modification may be effected at any time in a writing executed by, or on behalf of, each of the Plan Proponents and without leave or an order of the Bankruptcy Court or the District Court. If any Plan Proponent desires to waive a condition precedent to facilitate confirmation of the Plan or the occurrence of the Effective Date, the other Plan Proponent (and, if applicable, HII) shall promptly confer with it as to whether or not the suggested waiver should be given, in recognition that time is of the essence.

11.4. Failure to Achieve the Effective Date. If each of the conditions to the Effective Date is not met or duly waived in accordance with <u>Section 11.3</u>, then upon joint motion by the Plan Proponents and notice to such parties in interest as the Bankruptcy Court may direct, the Confirmation Order shall be vacated by the Bankruptcy Court and/or the District Court, as the case may be. If (a) the Plan Proponents revoke or withdraw the Plan, (b) the Confirmation Order is vacated, (c) the Plan is otherwise not confirmed by a Final Order, or (d) the Plan is confirmed and

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 56 of 129

does not become effective for any other reason, then the rights of all parties in interest in the Chapter 11 Case are and shall be reserved in full. In any such event, the Plan shall become null and void in all respects; any settlement or compromise embodied in the Plan, any assumption or rejection of Executory Contracts effected by the Plan, and any document or agreement executed pursuant to the Plan, shall be deemed null and void; and nothing contained in the Plan or the Confirmation Order, if previously entered, and no acts taken in preparation for consummation of the Plan, shall: (a) constitute or be deemed to constitute a waiver, release or settlement of any Claims by or against, or any Equity Interests in, Hopeman or any other Entity, (b) prejudice in any manner the rights of Hopeman or any Entity in any further proceedings involving Hopeman; or (c) constitute an admission of any sort by Hopeman or any other Entity.

ARTICLE XII

RETAINED JURISDICTION OF BANKRUPTCY COURT

12.1. Retention of Jurisdiction. Pursuant to sections 105(a) and 1142 of the Bankruptcy Code, the Bankruptcy Court and, to the extent applicable, the District Court, shall, to the fullest extent permitted by law, retain and have exclusive jurisdiction over all matters arising out of and related to the Chapter 11 Case and the Plan, including, among other things, jurisdiction to:

(a) hear and determine any and all objections to and proceedings involving:
(i) the allowance, estimation, classification, and subordination of Claims (other than Channeled Asbestos Claims) that have been or properly should have been brought in the Bankruptcy Court; or (ii)Equity Interests;

(b) hear and determine all objections to the termination of the Asbestos Trust;

(c) hear and determine such other matters that may be set forth in or arise in connection with the Plan, the Confirmation Order, the Asbestos Permanent Channeling Injunction, or the Asbestos Trust Agreement;

(d) hear and determine any proceeding that involves the validity, application, construction, enforceability, or modification of the Asbestos Permanent Channeling Injunction;

(e) hear and determine any conflict or other issues that may arise in the Chapter 11 Case, including the implementation and enforcement of all orders entered by the Bankruptcy Court in the Chapter 11 Case, and in the administration of the Asbestos Trust;

(f) enter such orders as are necessary to implement and enforce the releases and injunctions described herein, including, if necessary, in connection with application of the protections afforded by section 524 of the Bankruptcy Code and/or the Plan to the Protected Parties;

(g) hear and determine any and all applications for allowance of Professional Fee Claims and any other fees and expenses authorized to be paid or reimbursed under the Bankruptcy Code or the Plan;

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 57 of 129

(h) enter such orders authorizing non-material modifications to the Plan or any other Plan Document, as may be necessary to comply with section 468B of the Internal Revenue Code;

(i) hear and determine any applications pending on the Effective Date for the assumption, assumption and assignment, or rejection, as the case may be, of Executory Contracts to which Hopeman is a party, and to hear and determine and, if necessary, liquidate any and all Claims arising therefrom;

(j) hear and determine any and all applications, Claims, causes of action, adversary proceedings, and contested or litigated matters that may be pending in the Chapter 11 Case on the Effective Date;

(k) consider any modifications of the Plan, and remedy any defect or omission or reconcile any inconsistency or make any other necessary modifications in or to the Plan, the Asbestos Trust Documents, or any order of the Bankruptcy Court, including the Confirmation Order, as may be necessary to carry out the purposes and intent of the Plan, to the extent authorized by the Bankruptcy Code and the Bankruptcy Rules;

(1) hear and determine all controversies, suits, and disputes that may arise in connection with the interpretation, enforcement, or consummation of the Plan or any Entity's obligations hereunder and issue orders in aid of confirmation, consummation and execution of the Plan to the extent authorized by section 1142 of the Bankruptcy Code, including, but not limited to, compelling the conveyance of property and other performance contemplated under the Plan and documents executed in connection herewith;

(m) hear and determine all questions and disputes and enter such orders or judgments, including injunctions, as are necessary to (i) enforce the title, rights and powers of Reorganized Hopeman and the Asbestos Trust, and (ii) enable holders of Claims to pursue their rights against any Entity that may be liable therefor pursuant to applicable law or otherwise;

(n) hear and determine any proposed compromise and settlement of any Claim against or cause of action by or against Hopeman that has been or properly should have been brought in the Bankruptcy Court to the extent requested by Reorganized Hopeman;

(o) hear and determine any timely objections to Administrative Expense Claims asserted or to Proofs of Claim filed, both before and after the Confirmation Date, including any objections to the classification of any Claim, and to allow or disallow any Disputed Claim, in whole or in part;

(p) hear and determine matters concerning state, local and federal taxes, tax refunds, tax attributes, tax benefits, and similar or related matters with respect to Hopeman, Reorganized Hopeman, or the Asbestos Trust arising on or prior to the Effective Date, arising on account of transactions contemplated by the Plan Documents, or relating to the period of administration of the Chapter 11 Case;

(q) hear and determine such other matters as may be set forth in the Confirmation Order or other orders of the Bankruptcy Court, or which may arise in connection

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 58 of 129

with the Plan, the Confirmation Order, or the Effective Date, as may be authorized under the provisions of the Bankruptcy Code or any other applicable law;

(r) hear and determine all controversies, suits, and disputes that may arise in connection with the interpretation, enforcement, or consummation of the Plan or any Entity's obligations hereunder, including performance of Hopeman's duties under the Plan;

(s) hear and determine any motions or applications seeking approval of any Asbestos Insurance Settlement whether filed before, on, or after the Effective Date.

(t) enforce remedies upon any default under the Plan or any of the other Plan Documents;

Code;

(u) hear and determine any other matter not inconsistent with the Bankruptcy

(v) hear and determine any claim that in any way challenges or is based on any provision in the Confirmation Order; and

(w) enter a final decree closing the Chapter 11 Case.

If and to the extent that the Bankruptcy Court is not permitted under applicable law to exercise jurisdiction over any of the matters specified above, including the matter described in <u>Section 12.1(d)</u>, the reference to the "Bankruptcy Court" in the preamble to this <u>Section 12.1</u> shall be deemed to be a reference to the "District Court." Notwithstanding anything in this <u>Section 12.1</u> to the contrary, and except as set forth in <u>Article VIII</u>, the Asbestos Trust Agreement and the Asbestos Trust Distribution Procedures shall govern the processing, liquidation, or satisfaction of Channeled Asbestos Claims and the forum in which Channeled Asbestos Claims shall be determined.

12.2. Other Proceedings. Notwithstanding any term or provision herein to the contrary, nothing in the Plan Documents shall detract from or contravene any jurisdictional or other provisions, including Sections 8.12, 8.13, and 8.15 hereof, that permit or require legal actions or proceedings to be brought in a court that is not the Bankruptcy Court or the District Court. Notwithstanding anything in Section 12.1 hereof to the contrary, the Asbestos Trust may initiate, prosecute, defend, and resolve all legal actions and other proceedings related to or arising from any asset, liability, or responsibility of the Asbestos Trust, including any Causes of Action arising from or related to the Asbestos Insurance Rights and Channeled Asbestos Claims, outside of the Bankruptcy Court and in any court of competent jurisdiction, in accordance with Section 8.3. In addition, notwithstanding anything in Section 12.1 hereof to the contrary, Asbestos Claimants may bring Extracontractual Claims in accordance with Section 8.13(e) outside of the Bankruptcy Court and in any court of competent jurisdiction.

12.3. Modification of Plan. Any Plan Proponent, with the consent of the other Plan Proponent, which consent may not be unreasonably withheld, may alter, amend, or modify the Plan or any Exhibits thereto under section 1127(a) of the Bankruptcy Code at any time prior to the Confirmation Date and may include any such amended Exhibits in the Plan, provided that the Plan, as modified, meets the requirements of sections 1122 and 1123 of the Bankruptcy Code, and the Plan

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 59 of 129

Proponents shall have complied with section 1125 of the Bankruptcy Code, to the extent necessary. Further, any Plan Proponent, with the consent of the other Plan Proponents, which consent may not be unreasonably withheld, may alter, amend, or modify the Plan or any Exhibits thereto at any time after entry of the Confirmation Order and before the Plan's substantial consummation, provided that: (a) the Plan, as modified, altered, or amended, meets the requirements of sections 1122 and 1123 of the Bankruptcy Code and (b) the Bankruptcy Court, after notice and a hearing, confirms the Plan as modified, under section 1129 of the Bankruptcy Code, and finds that the circumstances warrant such modification. A holder of a Claim that has accepted or rejected the Plan shall be deemed to have accepted or rejected, as the case may be, the Plan as modified, unless, within the time fixed by the Bankruptcy Court, if any, such holder changes its previous acceptance or rejection, to the extent such holder is afforded the opportunity to do so under section 1127(d) of the Bankruptcy Code.

On and after the Effective Date, Reorganized Hopeman or the Asbestos Trust, as applicable, may remedy any defects or omissions or reconcile any inconsistencies in the Plan Documents for the purpose of implementing the Plan in such manner as may be necessary to carry out the purposes and intent of the Plan, so long as the interests of the holders of Allowed Claims and other applicable parties-in-interest are not adversely affected thereby.

Notwithstanding anything in this <u>Section 12.3</u>, there shall be no modification to this Plan made at any time (whether before or after entry of the Confirmation Order) that would materially reduce or eliminate any of the protections provided herein to the Protected Parties, Released Parties, or Exculpated Parties. Notwithstanding anything in this <u>Section 12.3</u>, there shall be no modification to this Plan made at any time (whether before or after entry of the Confirmation Order) that would materially reduce or eliminate any of the rights or protections provided under <u>Section 8.15</u>, <u>Section 8.17</u>, <u>Section 10.3(b)</u>, <u>Section 11.1(e)(i)</u>, or <u>Section 11.3</u> to HII without the prior written consent of HII.

12.4. Consent to Jurisdiction. Upon default under the Plan, Hopeman, Reorganized Hopeman, the Asbestos Trust, the Asbestos Trustee, the Asbestos Trust Advisory Committee, and the Future Claimants' Representative, respectively, consent to the jurisdiction of the Bankruptcy Court, and agree that it shall be the preferred forum for all proceedings relating to any such default.

ARTICLE XIII

MISCELLANEOUS PROVISIONS

13.1. Governing Law. Unless a rule of law or procedure is supplied by federal law (including the Bankruptcy Code and Bankruptcy Rules), or a Schedule or Exhibit hereto or instrument, agreement or other document executed under the Plan provides otherwise, the rights, duties and obligations arising under the Plan, and the instruments, agreements and other documents executed in connection with the Plan, shall be governed by, and construed and enforced in accordance with, the internal laws of the Commonwealth of Virginia without giving effect to the principles of conflicts of law thereof.

13.2. Notices. To be effective, all notices, requests and demands to or upon the Plan Proponents, or, as applicable, upon the Future Claimants' Representative and the Committee, shall be in writing and, unless otherwise expressly provided herein, shall be addressed as follows and

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 60 of 129

mailed by registered or certified mail, return receipt requested, and shall be deemed to have been duly given or made when actually delivered:

If to Hopeman:

Hopeman Brothers, Inc. 6 Auburn Ct., Unit 3 Brookline, Massachusetts 02446 Attention: Christopher Lascell, President

with a copy (which alone will not constitute notice) to:

Hunton Andrews Kurth LLP 600 Travis Street, Suite 4200 Houston, Texas 77002 Attention: Joseph P. Rovira, Esq. (josephrovira@HuntonAK.com) Catherine A. Rankin, Esq. (crankin@HuntonAK.com)

and

Hunton Andrews Kurth LLP Riverfront Plaza, East Tower 951 East Byrd Street Richmond, Virginia 23219 Attention: Tyler P. Brown, Esq. (tpbrown@HuntonAK.com) Henry P. (Toby) Long, III (hlong@HuntonAK.com)

If to the Committee:

Caplin & Drysdale, Chartered 1200 New Hampshire Avenue, NW, 8th Floor Washington, DC 20036 Attention: Kevin C. Maclay, Esq. (kmaclay@capdale.com) Todd E. Phillips, Esq. (tphillips@capdale.com) Jeffrey A. Liesemer, Esq. (jliesemer@capdale.com) Nathaniel R. Miller, Esq. (nmiller@capdale.com)

If to the Future Claimants' Representative:

Campbell & Levine, LLC 222 Delaware Avenue, Suite 1620 Wilmington, Delaware 19801 Attention: Marla Rosoff Eskin, Esq. (meskin@camlev.com)

13.3. Plan Supplement. Any Exhibits, lists, or schedules referred to herein or in the Disclosure Statement but not filed with the Plan shall be contained in the Plan Supplement and filed with the Clerk of the Bankruptcy Court at least five (5) Business Days prior to the deadline established by the Bankruptcy Court for the filing and service of objections to the Plan.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 61 of 129

13.4. Inconsistencies. To the extent the Plan is inconsistent with either the Disclosure Statement or the other Plan Documents, the provisions of the Plan shall be controlling. To the extent the Plan is inconsistent with the Confirmation Order, the provisions of the Confirmation Order shall be controlling.

13.5. Tax Reporting and Compliance. In connection with the Plan and all instruments issued in connection therewith and Distributions thereon, Hopeman and Reorganized Hopeman shall comply with all withholding and reporting requirements imposed by any federal, state, local, or foreign taxing authority, and all Distributions hereunder shall be subject to any such withholding and reporting requirements. No holder of an Allowed Claim against Hopeman shall effectuate any withholding with respect to the cancellation or satisfaction of such Allowed Claim under the Plan. Reorganized Hopeman is hereby authorized to request an expedited determination of taxes under section 505(b) of the Bankruptcy Code for all taxable periods of Hopeman ending after the Petition Date through and including the Effective Date of the Plan.

13.6. Payment of Statutory Fees. All fees due and payable pursuant to section 1930 of Title 28 of the U.S. Code prior to the Effective Date shall be paid by Hopeman on the Effective Date. After the Effective Date, Reorganized Hopeman shall pay any and all such fees from the Net Reserve Funds when due and payable, and shall file with the Bankruptcy Court quarterly reports in a form reasonably acceptable to the United States Trustee. Hopeman and Reorganized Hopeman shall remain obligated to pay quarterly fees to the Office of the United States Trustee until Hopeman's case is converted, dismissed, or closed, whichever occurs first.

13.7. Exemption from Transfer Taxes. Pursuant to section 1146(a) of the Bankruptcy Code, the issuance, transfer, or exchange of notes or equity securities under the Plan, the creation of any mortgage, deed of trust, pledge or other security interest, the making or assignment of any lease or sublease, or the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with the Plan shall be exempt from all stamp, transfer, or similar taxes, as provided in section 1146(a).

13.8. Recordable Order. Upon Confirmation of the Plan, the Confirmation Order shall be deemed to be in recordable form, and shall be accepted by any recording officer for filing and recording purposes without further or additional orders, certifications, or other supporting documents.

13.9. Binding Effect. The rights, benefits and obligations of any Entity named or referred to in the Plan, or whose actions may be required to effectuate the terms of the Plan, shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor or assign of such Entity (including any trustee appointed for Hopeman under chapters 7 or 11 of the Bankruptcy Code). The Confirmation Order shall provide that the terms and provisions of the Plan and the Confirmation Order shall survive and remain effective after entry of any order which may be entered converting the Chapter 11 Case to a case under chapter 7 of the Bankruptcy Code, and the terms and provisions of the Plan shall continue to be effective in this or any superseding case under the Bankruptcy Code.

13.10. Severability. After the Effective Date, any provision of the Plan, the Plan Documents, the Confirmation Order, or any of the Exhibits to the Plan that is determined to be

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 62 of 129

prohibited, unenforceable, or invalid by a court of competent jurisdiction or any other governmental Entity with appropriate jurisdiction may be deemed ineffective as to any jurisdiction in which such provision is prohibited, unenforceable, or invalidated to the extent of such prohibition, unenforceability, or invalidation, without invalidating the effectiveness of the remaining provisions of the Plan, the Plan Documents, the Confirmation Order, and the Exhibits to the Plan or affecting the validity or enforceability of such provision and such remaining provisions in any other jurisdiction.

13.11. Further Authorizations. The Plan Proponents, as applicable, and, after the Effective Date, the Asbestos Trust, if and to the extent necessary, may seek such orders, judgments, injunctions, and rulings as each deems necessary to carry out the intentions and purposes of, and to give full effect to the provisions of, the Plan.

13.12. General Statements. Statements of a general nature set forth in the Plan shall not be construed to limit or restrict the specific provisions herein.

13.13. Entire Agreement. The Plan Documents set forth the entire agreement and undertakings relating to the subject matter thereof and supersede all prior discussions, negotiations, understandings, and documents. No Entity shall be bound by any terms, conditions, definitions, warranties, understandings, or representations with respect to the subject matter of the Plan Documents other than as expressly provided for in the Plan or the other Plan Documents or as may hereafter be agreed to by the affected parties in writing.

IN WITNESS WHEREOF, the undersigned has duly executed the Plan as of the date first above written.

Respectfully submitted,

Richmond, Virginia Dated: April 29<u>May 21</u>, 2025 Hopeman Brothers, Inc.

By: <u>/s/ Christopher Lascell</u> Name: Christopher Lascell Title: President

The Official Committee of Unsecured Creditors

By: <u>/s/Lisa Nathanson Busch</u> Name: Lisa Nathanson Busch, as specifically authorized by Committee Co-Chairperson, Darrell Kitchen

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 63 of 129

Summary report:		
Litera Compare for Word 11.9.1.1 Document compared	rison done on	
5/21/2025 9:28:11 AM		
Style name: Firm Standard		
Intelligent Table Comparison: Active		
Original DMS: iw://hunton.cloudimanage.com/DMS/351056	400/2	
Modified DMS: iw://hunton.cloudimanage.com/DMS/351309905/3		
Changes:		
Add	21	
Delete	21	
Move From	0	
Move To	0	
Table Insert	0	
Table Delete	0	
Table moves to	0	
Table moves from	0	
Embedded Graphics (Visio, ChemDraw, Images etc.)	0	
Embedded Excel	0	
Format changes	0	
Total Changes:	42	

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 64 of 129

EXHIBIT B

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 65 of 129

HUNTON ANDREWS KURTH LLP

Joseph P. Rovira (admitted *pro hac vice*) Catherine A. Rankin (admitted *pro hac vice*) 600 Travis Street, Suite 4200 Houston, Texas 77002 Telephone: (713) 220-4200

HUNTON ANDREWS KURTH LLP

Tyler P. Brown (VSB No. 28072) Henry P. (Toby) Long, III (VSB No. 75134) Riverfront Plaza, East Tower 951 East Byrd Street Richmond, Virginia 23219 Telephone: (804) 788-8200

Attorneys for Debtor and Debtor in Possession

UNITED STATES BANKRUPTCY COURT EASTERN DISTRICT OF VIRGINIA RICHMOND DIVISION

:

:

:

In re:

HOPEMAN BROTHERS, INC.,

Debtor.

Chapter 11

Case No. 24-32428 (KLP)

DISCLOSURE STATEMENT WITH RESPECT TO THE <u>AMENDED</u> PLAN OF REORGANIZATION OF HOPEMAN BROTHERS, INC. UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

THIS DISCLOSURE STATEMENT HAS NOT BEEN APPROVED BY THE BANKRUPTCY COURT AND NO ONE MAY SOLICIT ACCEPTANCES OR REJECTIONS OF THE PLAN UNTIL THE DISCLOSURE STATEMENT HAS BEEN APPROVED BY THE BANKRUPTCY COURT AS CONTAINING ADEQUATE INFORMATION. THIS DISCLOSURE STATEMENT IS SUBJECT TO FURTHER MODIFICATION PRIOR TO BANKRUPTCY COURT APPROVAL.

The Plan of Reorganization, attached hereto as <u>Exhibit A</u>, provides for an "Asbestos Permanent Channeling Injunction" in accordance with sections 105(a) and 524(g) of the Bankruptcy Code. For a description of the claims and causes of action to be enjoined and the identities of entities that would be subject to the Asbestos Permanent Channeling Injunction, see Section IV.G of this Disclosure Statement and Article X of the Plan.

Dated: April 29 May 21, 2025

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 66 of 129

Hopeman Brothers, Inc. ("<u>Hopeman</u>"), the debtor and debtor-in-possession in the abovecaptioned Chapter 11 Case, and the Official Committee of Unsecured Creditors (the "<u>Committee</u>", and collectively with Hopeman, the "<u>Plan Proponents</u>") are soliciting votes of creditors in favor of their jointly proposed *Plan of Reorganization of Hopeman Brothers, Inc. Under Chapter 11 of the Bankruptcy Code* (the "<u>Plan</u>"), attached hereto as <u>Exhibit A</u>. Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Plan.

Please read this Disclosure Statement, the Plan, the exhibits, other supporting materials, and any appropriate ballot carefully and follow the instructions set forth below to vote on the Plan. The Plan Proponents believe that the Plan provides the best method of maximizing recoveries for the holders of Claims against Hopeman. Therefore, the Plan Proponents recommend that all creditors who are entitled to vote should vote in favor of the Plan.

The Plan provides, among other things, for the issuance of the Asbestos Permanent Channeling Injunction under sections 105(a) and 524(g) of the Bankruptcy Code that will, among other things, result in the channeling of certain asbestos-related liabilities of Hopeman (excluding liabilities for worker compensation claims) to a newly created Asbestos Trust, as more fully described in the Plan. The terms, conditions, and Entities subject to the Asbestos Permanent Channeling Injunction are set forth in <u>Section 10.3</u> of the Plan. Except as otherwise permitted under the terms of the Plan, the Asbestos Permanent Channeling Injunction will prohibit any acts to collect, recover, or offset any asbestos-related liabilities of Hopeman against Hopeman, Reorganized Hopeman, certain third parties, and various persons or entities related to those parties. Under section 524(g), all distributions paid under the Plan to the holders of Asbestos Claims and Demands will be the responsibility of the Asbestos Trust, and Hopeman and Reorganized Hopeman shall have no further personal liability for such claims.

In addition, under the terms of <u>Section 8.12</u> of the Plan, holders of Asbestos Claims and Demands may bring actions against Reorganized Hopeman and, to the extent they have obtained a judgment against Reorganized Hopeman or have the right to pursue direct actions under applicable nonbankruptcy law, such holders may bring judgment-enforcement or direct actions against Non-Settling Asbestos Insurers, in accordance with <u>Section 8.13</u> of the Plan, to obtain the benefits of Asbestos Insurance Coverage. All other claims against Hopeman will receive payments and distributions in accordance with the Plan and the Asbestos Trust Documents.

The following description of the Plan is qualified in its entirety by the terms, conditions, and definitions set forth in the Plan.

THIS DISCLOSURE STATEMENT CONTAINS SUMMARIES OF CERTAIN PROVISIONS OF THE PLAN, CERTAIN STATUTORY PROVISIONS, CERTAIN DOCUMENTS RELATED TO THE PLAN. CERTAIN ANTICIPATED EVENTS IN THE **CHAPTER 11 CASE AND CERTAIN FINANCIAL INFORMATION. ALTHOUGH THE** PLAN PROPONENTS BELIEVE THAT THE DISCLOSURE STATEMENT, AND RELATED DOCUMENT SUMMARIES, ARE FAIR AND ACCURATE, THEY ARE OUALIFIED TO THE EXTENT THEY DO NOT SET FORTH THE ENTIRE TEXT OF THE PLAN, RELATED DOCUMENTS, OR ANY STATUTORY PROVISIONS. THE TERMS OF THE PLAN GOVERN IN THE EVENT OF ANY INCONSISTENCY WITH THIS DISCLOSURE STATEMENT. ALL EXHIBITS TO THE DISCLOSURE STATEMENT ARE INCORPORATED INTO AND ARE A PART OF THIS DISCLOSURE STATEMENT AS IF SET FORTH IN FULL HEREIN. THE STATEMENTS CONTAINED IN THIS DISCLOSURE STATEMENT ARE MADE AS OF THE DATE HEREOF, UNLESS OTHERWISE SPECIFIED, AND THE PLAN PROPONENTS DISCLAIM ANY **OBLIGATION TO UPDATE ANY SUCH STATEMENTS AFTER THE HEARING ON** APPROVAL OF THE DISCLOSURE STATEMENT. ALL PARTIES ENTITLED TO VOTE ARE ADVISED AND ENCOURAGED TO READ THE DISCLOSURE STATEMENT AND THE PLAN IN THEIR ENTIRETY BEFORE VOTING TO ACCEPT **OR REJECT THE PLAN.**

NO INDEPENDENT AUDITOR OR ACCOUNTANT HAS REVIEWED OR APPROVED THE LIQUIDATION ANALYSIS HEREIN.

IN CONNECTION WITH HOPEMAN'S SOLICITATION OF ACCEPTANCES OF THE PLAN PURSUANT TO SECTION 1126(B) OF THE BANKRUPTCY CODE, THE PLAN PROPONENTS ARE FURNISHING A SOLICITATION PACKAGE, CONSISTING OF THE DISCLOSURE STATEMENT, THE ENCLOSURES HERETO, AND A BALLOT OR MASTER BALLOT, AS APPLICABLE, TO EACH RECORD HOLDER OF CLAIMS ELIGIBLE TO VOTE OR ITS COUNSEL. THIS DISCLOSURE STATEMENT IS TO BE USED BY EACH SUCH ELIGIBLE HOLDER SOLELY IN CONNECTION WITH ITS EVALUATION OF THE PLAN AND ITS DECISION FOR VOTING PURPOSES WHETHER TO ACCEPT OR REJECT THE PLAN; USE OF THIS DISCLOSURE STATEMENT FOR ANY OTHER PURPOSE IS NOT AUTHORIZED. THIS DISCLOSURE STATEMENT MAY NOT BE REPRODUCED OR PROVIDED TO ANYONE OTHER THAN ADVISORS TO THE RECIPIENT WITHOUT THE PRIOR WRITTEN CONSENT OF HOPEMAN.

THIS DISCLOSURE STATEMENT HAS BEEN PREPARED IN ACCORDANCE WITH SECTION 1125 OF THE BANKRUPTCY CODE AND RULE 3016(C) OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE, AND NOT NECESSARILY IN ACCORDANCE WITH FEDERAL OR STATE SECURITIES LAWS OR OTHER NON-BANKRUPTCY LAW. THIS DISCLOSURE STATEMENT WAS PREPARED WITH THE INTENT TO PROVIDE "ADEQUATE INFORMATION" (AS DEFINED IN THE BANKRUPTCY CODE) TO ENABLE HOLDERS OF CLAIMS AGAINST AND INTERESTS IN HOPEMAN TO MAKE INFORMED JUDGMENTS ABOUT THE PLAN.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 68 of 129

TABLE OF CONTENTS

Page

I.	Intro	duction1	
II.	The Disclosure Statement and Plan2		
III.			
IV.	Description of Hopeman's History, Claims, and Insurance5		
	A.	Hopeman's Corporate History and Formation	
	B.	Hopeman's Prepetition Business	
	C.	Board of Directors7	
	D.	Historical Asbestos Claims Against Hopeman7	
	E.	Hopeman's Prepetition Claims Process8	
	F.	Hopeman's Insurance Coverage and Prepetition Funding of Defense and Reconciliation Costs Concerning Asbestos Claims	
V.	The Chapter 11 Case10		
	A.	Reasons for the Chapter 11 Filing10	
VI.	Insurance Settlement Motions, Plan Negotiations and the 524(g) Settlement1		
	A.	Insurance Settlement Motions10	
	В.	Prosecution of the Insurer Settlement Motions and Negotiations that Led to the 524(g) Settlement	
	C.	Unresolved Contentions	
VII.	Other	r Events During the Chapter 11 Case14	
	A.	First Day Motions and Orders14	
	В.	Schedules and Statements and the Bar Date for Non-Asbestos Claims15	
	C.	Retention of Hopeman's Chapter 11 Professionals16	
	D.	Formation of the Official Committee of Unsecured Creditors and Retention of Committee Professionals	
	E.	Exclusivity17	
	F.	Appointment of Future Claimants' Representative17	
	G.	U.S17	
VIII.	Summ	nary of the Plan17	
	A.	Treatment of Holders of Claims and Interests Under the Plan18	
	В.	Administrative Expense Claims18	
	C.	Professional Fees	

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 69 of 129

D.	D. Priority Tax Claims		
E.	. Treatment of Classified Claims and Interests		20
	1.	Class 1 – Priority Non-Tax Claims	20
	2.	Class 2 – Secured Claims	20
	3.	Class 3 – General Unsecured Claims	20
	4.	Class 4 – Channeled Asbestos Claims	21
	5.	Class 5 – Equity Interests	21
F.	Mean	s for Implementation of the Plan	21
	1.	Generally	21
	2.	The Asbestos Trust	22
	3.	Net Reserve Funds	26
	4.	Actions Against Reorganized Hopeman or Wayne to Obtain Benefits of Asbestos Insurance Coverage	27
	5.	Actions Against Non-Settling Asbestos Insurers to Obtain Benefits of Asbestos Insurance Coverage	27
	6.	No Actions Against Released Parties	30
	7.	Special Provisions Pertaining to HII	30
	8.	Payments or Distributions from the Asbestos Trust	31
	9.	Settlements with Non-Settling Asbestos Insurers After the Effective Date	31
	10.	Insurance Neutrality	32
	11.	Reorganized Hopeman Common Stock	32
	12.	Corporate Governance of Reorganized Hopeman	32
	13.	Restructuring Transactions and Continued Corporate Existence of Reorganized Hopeman	32
G.	Effect	t of Confirmation	33
	1.	Dissolution of Committee; Discharge of the Future Claimants' Representative	33
	2.	Vesting of Assets	33
	3.	Preservation of Certain Causes of Action, Rights to Settle Claims and Compromise Controversies, and Defenses	33
	4.	Terms of Injunction and Automatic Stay	34
	5.	No Liability for Certain Tax Claims	34
	6.	No Successor Liability	34
H.	Excul	pation, Injunctions, Releases and Settlement	35
	1.	Exculpation	35

		2.	Discharge of Hopeman and Reorganized Hopeman	35
		3.	Hopeman Discharge Injunction	35
		4.	Asbestos Permanent Channeling Injunction	
		5.	Releases by Hopeman and Estate	
		6.	Release of Hopeman's Directors and Officers by Releasing Parties.	
		7.	Releases of Avoidance Actions and Estate Causes of Action Agai Vendor Released Parties	
		8.	No Actions on Account of Released Claims	
		9.	Disallowed Claims	
		10.	Compromise and Settlement of Claims and Interests and Approval Certain Settling Insurers Agreement	
	I.	Exec	cutory Contracts and Unexpired Leases	40
		1.	General Treatment	40
		2.	Asbestos Insurance Agreements	40
		3.	Bar Date to Rejection Damages	41
J. Procedures for Resolving and Treated Disputed Claims		edures for Resolving and Treated Disputed Claims Other than Asbes ms and Professional Fee Claims		
		1.	Disputed Claims	41
		2.	Objection to Claims	41
		3.	Payments and Distributions with Respect to Disputed Claims	41
	K.	Misc	cellaneous Provisions	42
IX.	Alter	native	s To The Plan	42
	A.	Disn	nissal of Hopeman's Chapter 11 Case	42
	B.	Liqu	idation under Chapter 7 of the Bankruptcy Code	42
	C.	Alternate Chapter 11 Plan		
X.	Risk	Factor	۲۶	44
	A.	Certa	ain Bankruptcy Considerations	44
	B.	Proje	ected Financial Information	44
	C.	Distributions Under the Asbestos Trust Distribution Procedures44		
	D.	The	Asbestos Permanent Channeling Injunction	45
XI.	Tax (Conseq	uences of the Plan	45
	A.		tment of Transfer of Asbestos Trust Assets and Taxation of the Asbes	
	В.	Cons	sequences to Holders of Channeled Asbestos Claims	47
	C.	Info	rmation Reporting and Withholding	48

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 71 of 129

XII.	Plan Voting Instructions, Procedures and Deadline4			
	A.	Claim Holders Entitled to Vote	48	
	B. Voting Instructions and Voting Deadline			
	C.	Summary of Voting Procedures for Holders of Asbestos Claims	49	
XIII.	Confi	rmation of the Plan	50	
	A.	Confirmation Hearing	50	
	B.	General Requirements of Section 1129 of the Bankruptcy Code	50	
	C.	Best Interests Test	51	
	D.	Feasibility	51	
	E.	Cramdown	52	
XIV.	Recon	nmendation	53	

EXHIBITS TO DISCLOSURE STATEMENT

EXHIBIT A	JOINT PLAN OF REORGANIZATION
EXHIBIT B	LIQUIDATION ANALYSIS
EXHIBIT C	REORGANIZED HOPEMAN PROJECTIONS

I.

Introduction

The Plan Proponents submit this Disclosure Statement in connection with the Plan. This Disclosure Statement is being distributed to all holders of Claims and Equity Interests in accordance with section 1125(b) of the Bankruptcy Code, and Rules 2002, 3016, and 3017 of the Federal Rules of Bankruptcy Procedure. The Plan is attached hereto as <u>Exhibit A.</u>

This Disclosure Statement sets forth certain information regarding Hopeman's prepetition history, significant events during this Chapter 11 Case, and the contemplated post-confirmation reorganization of Hopeman. This Disclosure Statement also describes the Plan, certain effects of confirmation of the Plan, and the manner in which claims will be addressed and certain distributions will be made under the Plan. Finally, this Disclosure Statement discusses the confirmation process and the procedures that holders of Claims or Equity Interests in impaired Classes must follow to object to confirmation of the Plan. This Disclosure Statement was prepared with the intent to provide "adequate information" (as defined in the Bankruptcy Code) to enable holders of Claims against and Equity Interests in Hopeman to make informed judgments about the Plan.

As set forth in detail below, this Chapter 11 Case was commenced as a result of Hopeman's dwindling cash position coupled with over 2,700 Asbestos Claims asserted and remaining unresolved against Hopeman as of June 2024. These Asbestos Claims generally are based on allegations of personal injury as a result of alleged exposure to asbestos fibers contained in marine interior materials included with joiner packages provided by Hopeman.

The Plan is intended to resolve the Asbestos Claims against Hopeman in an efficient and equitable manner. As discussed in more detail below, the Plan incorporates a settlement (the "<u>524(g) Settlement</u>") between Hopeman, the Committee, and Huntington Ingalls Industries, Inc. ("<u>HII</u>"). The basic terms of the Settlement are set forth in a term sheet filed with the Bankruptcy Court on March 7, 2025. *See* Docket No. 609, <u>Exhibit B</u>. The centerpiece of the 524(g) Settlement and Plan is the establishment of a trust under section 524(g) of the Bankruptcy Code (as defined in the Plan, the "<u>Asbestos Trust</u>") and an injunction (as defined in the Plan, the "<u>Asbestos Claims</u>") to the Asbestos Trust. As agreed in the 524(g) Settlement, and as provided for in the Plan, Hopeman will transfer its remaining insurance coverage and cash to that trust to allow for resolution of the Plan.

Except as provided in <u>Section 8.12</u>, <u>Section 8.13</u> and <u>Section 8.15</u> of the Plan, the Asbestos Trust will assume liability for the Channeled Asbestos Claims and use the assets conveyed to the Asbestos Trust to resolve the Channeled Asbestos Claims and to compensate eligible holders of the Asbestos Claims in a fair and efficient manner. <u>Section 8.12</u> and <u>Section 8.13</u> of the Plan preserves for holders of Channeled Asbestos Claims their respective rights to obtain the benefits of Asbestos Insurance Coverage in the tort system, in addition to potential recoveries from the Asbestos Trust.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 73 of 129

The Asbestos Permanent Channeling Injunction to be issued as part of the Plan will stay, restrain, bar and enjoin Channeled Asbestos Claims against Hopeman, Reorganized Hopeman, current and former directors, officers, or employees of Hopeman, or any past or present Affiliate of the Hopeman, solely in their respective capacities as such, or Settled Asbestos Insurers, except as provided section 10.3(b)(v) (reserving the right of holders to pursue Channeled Asbestos Claims against Reorganized Hopeman that strictly conforms with the requirements of <u>Section 8.12</u> of the Plan to obtain the benefits of Asbestos Insurance Coverage).

The effect of "channeling" Asbestos Claims to the Asbestos Trust is that they may only be pursued against, and paid from, the Asbestos Trust, except as provided in <u>Section 8.12</u>, <u>Section 8.13</u>, and <u>Section 8.15</u> of the Plan.

For the reasons detailed in this Disclosure Statement, the Plan Proponents believe that there will be substantially more assets available to resolve Channeled Asbestos Claims under the Plan than would be the case if Hopeman were to be liquidated under chapter 7 because Hopeman is contributing substantial assets to the Asbestos Trust on behalf of themselves and the Protected Parties in exchange for the protections provided to those parties under the Plan, which would not be contributed otherwise, and Reorganized Hopeman will continue in existence to allow claimants to attempt to recover on their Channeled Asbestos Claims from the Asbestos Insurance Coverage that might not be available absent the reorganization.

Accordingly, the Plan Proponents believe that the Plan provides a fair and reasonable recovery to current claimants and future Demand holders and that acceptance of the Plan is in the best interests of all claimants and Demand holders. The Plan Proponents urge you to vote to accept the Plan.

II.

The Disclosure Statement and Plan

Chapter 11 is the principal reorganization chapter of the Bankruptcy Code. It allows a debtor to remain in operation and work out its financial difficulties. In a chapter 11 bankruptcy case, the debtor continues to manage its affairs as a debtor-in-possession and as a fiduciary to the creditors of its estate.

Formulation and confirmation of a chapter 11 plan are the principal goals of a chapter 11 case. A plan is the vehicle for satisfying claims against a debtor. After a plan has been filed, the holders of claims that will be impaired are permitted to vote to accept or reject the plan. In connection with filing a proposed plan, section 1125 of the Bankruptcy Code requires the plan proponent to prepare a disclosure statement containing adequate information of such kind, and in such detail, as to enable a hypothetical reasonable investor to make an informed judgment about the plan.

Chapter 11 does not require that each holder of a Claim against Hopeman vote in favor of the Plan in order for the Bankruptcy Court to confirm the Plan. At a minimum, however, the Bankruptcy Code requires that the Plan be accepted by at least one Class of Claims Impaired under the Plan. In order for a Class of Claims to accept the Plan, a majority in number and at least two-

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 74 of 129

thirds in amount of those Claims that vote must vote to accept the Plan. Additionally, to satisfy the requirements of section 524(g) of the Bankruptcy Code, 75% in number and at least two-thirds in amount of the Asbestos Claims in Class 4 (Channeled Asbestos Claims) that vote must vote in favor of the Plan.

Even though a holder of a Claim may choose either not to vote or to vote against the Plan, the holder will be bound by the terms and treatment set forth in the Plan if the Plan is confirmed by the Bankruptcy Court. Creditors who fail to vote will not be counted in determining acceptance or rejection of the Plan.

Confirmation of a chapter 11 plan by a bankruptcy court makes the plan binding upon the debtor, any issuer of securities under the plan, any person or entity acquiring property under the plan, and any creditor of or equity security holder in the debtor, whether or not such creditor or equity security holder (i) is impaired under or has accepted the plan, or (ii) receives or retains any property under the plan. Subject to certain limited exceptions and other than as provided in the plan itself or the confirmation order, confirmation orders often limit the rights of creditors against a debtor with respect to any debt that arose prior to the date of confirmation of the plan and substitute therefor the obligations specified under the confirmed plan and terminate all rights and interests of equity security holders.

III.

Summary of Classification of Treatment of Claims Against and Equity Interests in Hopeman

Claims and Priority Tax Claims, are classified for purposes of voting and making Distributions under the Plan. A summary of the classification of these Claims and Equity Interests, the proposed treatment of each Class of Claims or Equity Interests, and the voting status of each Class of Claims or Interests follows:

Class and	Treatment	Impairment and	Estimated Allowed
Claim or Equity		Entitlement to	Amount and
Interest		Vote	Percentage Recovery
	Except to the extent that the holder of an		
	Allowed Priority Non-Tax Claim agrees	Unimpaired	Estimated Allowed
	to less favorable treatment, each holder of		Amount: [TBD]
	an Allowed Priority Non-Tax Claim shall	Not Entitled to	
1	receive, in full and complete settlement,	Vote (presumed	Estimated Recovery
	release, and discharge of, and in exchange	to accept)	Percentage: 100%
Priority Non-Tax	for, such Allowed Priority Non-Tax		
Claims	Claim, Cash to be paid from the Net		
	Reserve Funds in an amount equal to the		
	Allowed amount of such Claim on the		
	later of (i) the Effective Date, and (ii) the		
	date on which such Claim becomes		
	Allowed, or, in each case, as soon as		
	reasonably practicable thereafter.		

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 75 of 129

[
2 Secured Claims	Except to the extent that the holder of an Allowed Secured Claim agrees to less favorable treatment, on the Effective Date or as soon as reasonably practicable thereafter, each holder of an Allowed Secured Claim shall receive, at the option of Reorganized Hopeman and in full and complete settlement, release, and discharge of, and in exchange for, such Claim (i) payment in full in Cash to be paid from the Net Reserve Funds; (ii) the collateral securing such Allowed Secured Claim; or (iii) other treatment rendering such Claim Unimpaired.	Unimpaired Not Entitled to Vote (presumed to accept)	Estimated Allowed Amount: [TBD] Estimated Recovery Percentage: 100%
3 General Unsecured Claims	Except to the extent that the holder of an Allowed General Unsecured Claim agrees to less favorable treatment, each holder of an Allowed General Unsecured Claim shall receive, in full and complete settlement, release, and discharge of, and in exchange for, such Allowed General Unsecured Claim, Cash in an amount equal to its Pro Rata share of the General Unsecured Recovery Pool on the later of (i) the Effective Date, and (ii) the date on which such Claim becomes Allowed, or, in each case, as soon as reasonably practicable thereafter. Solely for purposes of calculating Distributions to holders of Allowed General Unsecured Claims, on the Effective Date, all Disputed General Unsecured Claims will be treated as though they are Allowed in the amounts asserted or as estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code, and a reserve will be set aside for such Disputed General Unsecured Claims.	Impaired Entitled to Vote	Estimated Allowed Amount: \$69,000 Estimated Recovery Percentage: [TBD]
4 Channeled Asbestos Claims	As of the Effective Date, liability for all Channeled Asbestos Claims shall automatically, and without further act, deed, or court order, be channeled exclusively to and assumed by the Asbestos Trust in accordance with, and to the extent set forth in, <u>Articles VIII</u> and <u>X</u> of the Plan, the applicable Plan Documents, and the Confirmation Order. All Channeled Asbestos Claims shall be treated pursuant to the terms of <u>Article</u> <u>VIII</u> , the Asbestos Trust Agreement, and the Asbestos Trust Distribution Procedures. Except as provided in <u>Article</u> <u>VIII</u> , pursuant to section 524(g) of the	Impaired Entitled to Vote	Estimated Allowed Amount: Unknown Estimated Recovery Percentage: Unknown

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 76 of 129

	Bankruptcy Code, the Plan, and the Confirmation Order, Channeled Asbestos Claims shall be subject to the Asbestos Permanent Channeling Injunction.		
5	On the Effective Date, the Equity Interests in Hopeman shall be cancelled, annulled and extinguished.	Impaired Not Entitled to	Estimated Allowed Amount: N/A
Equity Interests		Vote (deemed to reject)	Estimated Recovery Percentage: 0%

IV.

Description of Hopeman's History, Claims, and Insurance

A. Hopeman's Corporate History and Formation

Hopeman's origins date from 1869, when Arendt Willem Hopeman, a native of Holland, established himself as a building contractor in Rochester, New York. In 1908, Mr. Hopeman's two sons joined him in the building contractor business, and they incorporated in New York as A.W. Hopeman & Sons Company ("<u>Hopeman & Sons</u>") and added cabinet-making to their business operations. In 1917, Hopeman & Sons diversified its business into marine joiner work, *i.e.*, the assemblage, furnishing, and installation of bulkhead panels, ceilings and other interior components of ships. In 1930, the marine business segment of Hopeman & Sons' operations was separately incorporated in Delaware as Hopeman, the debtor in this chapter 11 case.¹

After its formation, Hopeman continued in the exclusive business of marine joiner work for over 50 years. More specifically, Hopeman was a joiner subcontractor that would acquire materials from manufacturers, make modifications to those products to meet shipbuilder specifications, and deliver the resulting "joiner packages" to various shipyards for installation by either Hopeman or shipyard employees. Hopeman provided these joiner packages and/or related services for over 3,000 ships in at least 49 shipyards in over 19 states, including Virginia. Hopeman did not at any time own or operate any shipyard and did not own ships for which it provided joiner packages and/or services.

The burning of the S.S. Morro Castle off of the coast of New Jersey in 1934, which resulted in the loss of 124 crew and passengers, led to changes in regulations requiring the use of fireproof materials in ships. In 1939, the United States Coast Guard (the "Coast Guard") approved of asbestos-containing wall and ceiling panels, which later became required non-combustible materials in ship construction. To comply with these regulations for its subcontracting work, Hopeman purchased "Marinite" fireproof core panels (which contained asbestos) primarily from Johns-Manville Corporation ("Johns-Manville"). The facing materials for the panels included a

¹ Hopeman moved its shipyard and accounting operations from New York to Virginia in 1971, and by 1976, Hopeman was headquartered in and maintained its principal place of business in Virginia. In February 2007, Hopeman reincorporated in Virginia and, in 2015, AWH Corporation merged into Hopeman, leaving Hopeman as the surviving entity. Hopeman later closed its office in Virginia, but it continues to store business records in a warehouse in Virginia and maintains its active corporate status in Virginia with its registered agent located in Glen Allen, Virginia.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 77 of 129

Johns-Manville product called "Marine Veneer" (which also contained asbestos), and "Micarta," a Westinghouse Electric Corporation laminate (which sometimes had an asbestos backing), that Hopeman purchased from U.S. Plywood Corporation. Installation of the wall panels required cutting and drilling at the shipyard and/or aboard the ship being constructed.

Wayne Manufacturing Corporation ("<u>Wayne</u>"),² a now-dissolved former subsidiary of Hopeman, would glue the laminate to the Marinite core panels for the lining and divisional wall panels. Marine Veneer also was used for ceiling panels and was typically purchased by Hopeman and shipped directly to the shipyard.

On April 18, 1975, the Coast Guard approved Hopeman's asbestos-free "Beta 100" panel, for which Hopeman later received a patent. Following the Coast Guard's approval of Beta 100, Hopeman believes that it began using this asbestos-free panel. However, Hopeman had certain pre-existing contracts for which asbestos-containing panels were still specified. Hopeman believes that the majority of such contracts were completed by the end of 1976, and the remaining few contracts in 1977. Hopeman believes that the use of any asbestos-containing bulkhead panels (and Marine Veneer) was discontinued in 1977 once the pre-Beta 100 contracts were satisfied. During the 1980s, Hopeman transitioned its business away from ship joining and into manufacturing check-out counters used in commercial retail stores such as Walmart. In 2002, Hopeman spun off its cabinet-making business into Cinnabar Solutions, Inc. ("<u>Cinnabar</u>").³ In 2003, Hopeman sold substantially all of its remaining shipbuilding-related assets to an unrelated party, US Joiner LLC (the "<u>Asset Sale</u>"), pursuant to an Asset Purchase Agreement, dated as of December 23, 2003. Hopeman's liabilities for any Asbestos Claims were excluded and not assumed by US Joiner in the Asset Sale and, as such, remained the obligations of Hopeman post-sale closing.

B. Hopeman's Prepetition Business

Since the above-referenced Asset Sale in 2003, Hopeman has existed solely to defend and, when appropriate, settle the Asbestos Claims. Generally, the prepetition administration of the Asbestos Claims involved Hopeman defending and attempting to resolve lawsuits filed against it in federal and state courts across the country, reviewing and attempting to resolve pre-litigation claims submitted pursuant to Administrative Agreements (defined below), paying settled claims, and managing Hopeman's insurance program including by collecting insurance proceeds to reimburse it for defense and claims administration costs and indemnify it for payments Hopeman made to satisfy resolved Asbestos Claims.

Hopeman has no employees. Aside from its remaining cash and business records, Hopeman's only other assets are its interests in the remaining limits of its insurance policies. As

² Wayne, formerly known as Wayne Lumber and Manufacturing Corporation, was a Virginia corporation that became a wholly-owned subsidiary of Hopeman and was dissolved in 1985. Wayne primarily was in the sheet metal business, manufacturing furniture, doors, window casings, trim and stairs. Wayne, however, performed the job of gluing the Micarta laminate to the Marinite core panels Hopeman supplied to Wayne because Wayne was located in Waynesboro, Virginia, in a more suitable climate for gluing the boards together than in many of the shipyards where Hopeman was furnishing the finished panels. Hopeman did not continue the general business operations of Wayne after Wayne dissolved in 1985.

³ Cinnabar was subsequently sold in 2007.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 78 of 129

set forth below, Hopeman's only material unpaid liabilities are the Asbestos Claims, most of which currently are unresolved.

C. Board of Directors

Hopeman's board of directors is comprised of three members (the "<u>Board Members</u>"), each of whom are executive officers of Hopeman, as summarized in the chart below:

Name	Position
Christopher Lascell	President
Daniel Lascell	Secretary
Carrie Lascell Brown	Vice President

None of the above-listed Board Members had any affiliation with Hopeman's business operations prior to its cessation of operations in 2003, and each only became affiliated with Hopeman upon the passing of their father, David M. Lascell, in 2016. The Board Members' father previously served as general counsel to Hopeman and, following Hopeman ceasing its business operations in 2003, he became the sole officer and majority owner of Hopeman. In such capacities, the Board Members' father oversaw the administration and resolution of the Asbestos Claims.

In 2016, the Board Members acquired their father's shares, as well as the minority shares, in Hopeman. Since that time, the Board Members have been the 100% equity holders in Hopeman and have served as Hopeman's Board Members and officers. As such, in light of their lack of involvement in Hopeman's business operations prior to its cessation of operations in 2003, the Board Members have attempted to serve as stewards of Hopeman for the last eight years, overseeing the work of outside professionals in the administration and resolution of the Asbestos Claims.

As set forth in <u>Section 8.7</u> of the Plan, on the Effective Date, (a) the current officers and directors of Hopeman shall be deemed to resign from their respective positions by operation of the Plan and (b) the individual(s) identified in a notice to be filed jointly by the Committee and the Future Claimants' Representative no later than two (2) days prior to the deadline established to accept or reject the Plan shall be appointed to serve as the officers and as the director of Reorganized Hopeman.

D. Historical Asbestos Claims Against Hopeman

In 1979, Hopeman received its first Asbestos Claim alleging personal injury as a result of alleged exposure to asbestos fibers contained in marine interior materials included within the joiner packages provided by Hopeman. Since that time, claimants have asserted more than 126,000 Asbestos Claims against Hopeman. The Asbestos Claims include both lawsuits and out-of-court claims asserted through agreed procedures put in place with several law firms pursuant to Administrative Agreements (as defined below). While the pace of the filing of claims has slowed, the Asbestos Claims filed against Hopeman from January 2015 through April 2024 still exceeded

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 79 of 129

5,000 claims. Claimants continued to assert Asbestos Claims against Hopeman during the weeks leading up to the Petition Date and such Asbestos Claims are expected to continue for years to come.

The Asbestos Claims asserted against Hopeman typically involve allegations of the manifestation of one or more of four general types of asbestos-related diseases: (i) mesothelioma; (ii) lung cancer; (iii) cancers other than lung cancer, such as esophageal cancer or colon cancer; and (iv) asbestosis, or a disease that was not articulated on the face of the pleading or claim.

E. Hopeman's Prepetition Claims Process

Prepetition, the Asbestos Claims were asserted against Hopeman by two methods. The first method was pursuant to an agreed out-of-court claims process pursuant to administrative agreements Hopeman entered into with various personal injury law firms (collectively, the "<u>Administrative Agreements</u>"). The purpose of the Administrative Agreements was to resolve substantiated Asbestos Claims without the need for litigation, with each Disease Level pursuant to most agreements having been assigned a scheduled settlement amount. To this end, pursuant to the terms of the applicable Administrative Agreements, each claimant was required to submit evidence of his or her asbestos-related disease diagnosis and exposure to asbestos that was connected to Hopeman's business operations.

Upon confirmation that an Asbestos Claim was valid, Hopeman's third-party prepetition claims administrator, Special Claims Services, Inc. ("<u>SCS</u>"), would pay the agreed upon scheduled amount prescribed for the substantiated Disease Level pursuant to the applicable Administrative Agreement, or the parties would negotiate a settlement amount if there was no predetermined amount scheduled for the disease level with the counterparty. If the claims were not resolved through the administrative process, litigation of the claims might proceed in court.

Of the 1,318 Asbestos Claims Hopeman resolved in the 2023 calendar year, collectively through litigation or the administrative process, 224 of these claims were accepted as substantiated claims and settled by Hopeman. Of these settled claims, 66.47% were for mesothelioma, 27.84% were for lung cancer, 2.15% were for other types of cancer (such as esophageal cancer or colon cancer), and 3.54% were for asbestosis or another type of disease.

The second prepetition method for claim assertion was through a claimant naming Hopeman as a defendant in federal or state court litigation, typically along with multiple other defendants, including other parties that provided products or services in the construction or repair of ships, manufacturers and sellers of products, shipyards, and ship owners, among others, and at times, former directors and officers of Hopeman. The former directors and officers of Hopeman who have been named in asbestos-related lawsuits have indemnification rights against Hopeman, and Hopeman traditionally has paid the defense costs associated with defending the claims asserted against such directors and officers, subject to reimbursement for a portion of those defense costs by insurers.

As of June 23, 2024, over 2,700 unresolved Asbestos Claims have been asserted against Hopeman. These claims are filed in courts in, or submitted pursuant to the Administrative Agreements and assigned by SCS as if filed in, California (35), Illinois (32), Louisiana (88),

Maryland (1,097), Pennsylvania (41), and Virginia (1,376), with the remaining 47 claims being filed in, or assigned as if filed in, fourteen other states.⁴

F. Hopeman's Insurance Coverage and Prepetition Funding of Defense and Reconciliation Costs Concerning Asbestos Claims

Prior to the Petition Date, Hopeman primarily funded its defense and resolution of the Asbestos Claims by drawing upon available coverage from its liability insurance program and with cash on hand. Hopeman's asbestos-related liability insurance program consists of primary-layer insurance policies and multilayer excess general liability insurance policies issued by various insurers (collectively, the "<u>Insurers</u>"), the last coverage period for which ended December 31, 1984. The primary-layer policies Hopeman purchased from 1937 through 1984 were all issued by Liberty Mutual Insurance Company ("<u>LMIC</u>"). The excess insurance policies in the program were issued by LMIC and various other Insurers from 1965 through 1984.

Historically, pursuant to Hopeman's various insurance policies, solvent Insurers, within their applicable policy limits, would reimburse Hopeman for portions of its defense costs (including claims administration costs) and for portions of the liability payments it made to resolve Asbestos Claims. Additionally, prior to the Petition Date, Hopeman entered into various agreements with certain Insurers to address the Asbestos Claims. Specifically, in June 1985, Hopeman and certain of its Insurers, as well as other asbestos claim defendants and their respective insurers, entered into an Agreement Concerning Asbestos Claims (commonly known as the "<u>Wellington Agreement</u>"). Pursuant to the Wellington Agreement, participating insurers' obligations for asbestos-related claims, including for payment of defense costs and indemnification of liability payments incurred by Hopeman in connection with Asbestos Claims, were spread prorata across all insurance policies from a claimant's date of first exposure across a "coverage block" which, in Hopeman's case, eventually extended to 1984.

A number of Insurers who were not signatories to the Wellington Agreement entered into bilateral insurance settlement agreements, called "coverage-in-place" agreements, with Hopeman (collectively, the "<u>Asbestos CIP Agreements</u>"). Pursuant to each Asbestos CIP Agreement, the applicable insurance policy remained in place and the agreements obligated those Insurers to pay portions of Hopeman's defense costs and liability indemnification amounts for Asbestos Claims on terms identical to or substantially similar to those of the Wellington Agreement. As a result of such agreements and payments, all of the primary layer and excess insurance that Hopeman purchased from LMIC was released by Hopeman.⁵

⁴ Some of these claims may also be the subject of filed complaints that the parties to Administrative Agreements agree are not to proceed while the administrative process continues.

For claims submitted via Administrative Agreements, the above-referenced jurisdictions historically were assigned by SCS based upon several criteria, including (i) the jurisdiction in which the claimant's counsel typically filed lawsuits pertaining to the Asbestos Claims asserted against Hopeman; or (ii) the jurisdiction in which a claimant initially filed a lawsuit against Hopeman pertaining to his or her Asbestos Claim but subsequently pursued the claim instead through an Administrative Agreement.

⁵ As discussed below, the holders of Asbestos Claims assert that Hopeman's release of LMIC does not extinguish any rights to or interests in the coverage these claimants have against LMIC.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 81 of 129

Hopeman's historical spend to pay claims and fund defense costs in connection with the Asbestos Claims has far exceeded the amounts reimbursed by the Insurers. For the four-year period of 2020 through 2023, Hopeman's claim payments and defense costs totaled over \$52 million, with payments to claimants totaling \$30 million and defense costs totaling \$22 million. In 2023 alone, Hopeman spent over \$12 million in combined claim payments and defense costs, while being reimbursed only \$6.6 million of this amount by Insurers. As such, for year 2023, net of insurance recoveries, Hopeman used its own cash to pay approximately 35.12% of claim payments and 57.33% of defense costs, resulting in an annual cash burn of approximately \$5.5 million.

Pursuant to Asbestos CIP Agreements in effect in 2023, of the \$6,362,200 Hopeman paid to claimants in 2023, Insurers were responsible to reimburse Hopeman for 64.88% of those amounts and 42.67% of the \$5,946,060 in defense costs.

V.

The Chapter 11 Case

A. Reasons for the Chapter 11 Filing

After years of Hopeman covering the shortfall in insurance proceeds for its defense costs and claims payments in connection with the Asbestos Claims, Hopeman's cash reserves have dwindled. If allowed to continue on the current pace, Hopeman would deplete its remaining cash within 12 months of the Petition Date. Upon such depletion, only the coverage remaining from unexhausted insurance policies would be available to cover the costs and liability associated with the Asbestos Claims. With Hopeman unable to continue managing the defense and resolution of the Asbestos Claims upon exhausting its available cash, it would create the classic "race to the courthouse" for claimants to recover remaining insurance proceeds. As a result, in consultation with its restructuring counsel and other advisors, Hopeman determined that it was in its best interest, as well as in the best interest of holders of Asbestos Claims, to commence this chapter 11 proceeding to seek approval and implementation of an efficient, value maximizing process to monetize the remaining available insurance and distribute those proceeds equitably to valid holders of Asbestos Claims.

VI.

Insurance Settlement Motions, Plan Negotiations and the 524(g) Settlement

A. Insurance Settlement Motions

To accomplish Hopeman's primary goal for this Chapter 11 Case to establish an efficient and fair process to utilize Hopeman's remaining cash and its insurance policies to address the thousands of asbestos-related claims asserted against Hopeman, Hopeman promptly sought the Bankruptcy Court's approval of two insurance settlements – the "Chubb Insurer Settlement" and the "Certain Settling Insurers Settlement" – that would generate nearly \$50 million in cash and filed a proposed chapter 11 plan at Docket No. 84 (the "<u>Original Plan</u>") seeking to establish a

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 82 of 129

liquidation trust to which Hopeman would transfer the settlement proceeds and its other assets and have the trust address the remaining asbestos-related claims asserted against it.

Specifically, Hopeman believes that, prior to the Petition Date, Hopeman and the Chubb Insurers conducted extensive, good faith negotiations in the months leading up to the commencement of this Chapter 11 Case. These prepetition negotiations were initiated and conducted with the purpose of resolving Hopeman's unexhausted insurance coverage under policies issued by the Chubb Insurers and resulted in a prepetition settlement agreement, dated as of June 27, 2024, that monetizes the applicable insurance policies in the amount of \$31,500,000 (the "Chubb Insurer Settlement").

On the Petition Date, Hopeman filed the motion seeking approval of the Chubb Insurer Settlement, that was filed [Docket No. 9] (the "<u>Chubb Insurer Settlement Motion</u>").

Similarly, prior to the Petition Date, Hopeman also began arm's length, good faith negotiations with the following liability insurers: Continental Casualty Company, Fidelity & Casualty Company, Lexington Insurance Company, Granite State Insurance Company, the Insurance Company of the State of Pennsylvania, National Union fire Insurance Company of Pittsburgh, PA, and General Reinsurance Corporation (collectively, the "<u>Certain Settling Insurers</u>"), for the purpose of resolving Hopeman's unexhausted insurance policies issued by the Certain Settling Insurers that, subject to their respective terms, conditions, and exclusions cover or allegedly cover Asbestos Claims asserted against Hopeman. These negotiations resulted in a settlement between Hopeman and the Certain Settling Insurers, dated July 10, 2024 (the "<u>Certain Insurer Settlement Agreement</u>"), that monetizes the policies issued to Hopeman by the Certain Settling Insurers in the aggregate amount of \$18,395,011. By motion filed on July 10, 2024, Hopeman sought Bankruptcy Court approval of the Certain Insurer Settlement Agreement [Docket No. 53] (the "<u>Certain Insurer Settlement Approval Motion</u>"; together with the Chubb Insurer Settlement Motion, the "<u>Insurer Settlement Motions</u>").

On the same day Hopeman filed the Certain Insurer Settlement Approval Motion, Hopeman also sought approval of proposed procedures for providing notice of the insurance settlement motions and scheduling a hearing on the Insurer Settlement Motions. *See* Docket No. 54. On September 12, 2024, the Bankruptcy Court entered its order approving the proposed notice procedures and setting the hearing for approval of the Insurer Settlement Motions for November 12, 2024. *See* Docket No. 204.

B. Prosecution of the Insurer Settlement Motions and Negotiations that Led to the 524(g) Settlement

During the course of this case, Hopeman has made good faith efforts to obtain the Bankruptcy Court's approval of the Insurer Settlement Motions on an expeditious basis, while also allowing the Committee and creditors time to conduct discovery on and consider the merits of those proposed settlements.

Before completing that discovery, and following a Bankruptcy Court-approved agreement by Hopeman and the Committee to adjourn the hearing on the Insurer Settlement Motions to December 16 [Docket No. 376], Hopeman and the Committee executed a settlement term sheet,

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 83 of 129

effective as of November 29, 2024 (the "<u>November 29 Term Sheet</u>"), through which, among other things, (a) Hopeman agreed to adjourn the hearing on the Chubb Insurer Settlement Motion, (b) Hopeman and the Committee agreed to participate in judicial mediation concerning the Chubb Insurer Settlement Motion, and (c) the Committee agreed not to oppose approval of the Certain Settling Insurers Settlement Motion at the December 16 hearing. The November 29 Term Sheet also provided that Hopeman and the Committee agreed to negotiate in good faith over the terms of a chapter 11 plan that would propose to create a trust pursuant to section 524(g) of the Bankruptcy Code.

Hopeman successfully prosecuted the Certain Settling Insurers Motion at the December 16 hearing. On December 19, 2024, the Bankruptcy Court entered its Order approving the Certain Settling Insurer Settlement Motion and overruling four objections to that settlement. *See* Docket No. 442 (the "<u>Settlement Approval Order</u>"). Subsequently, two parties – including HII – appealed the Settlement Approval Order. While HII dismissed its appeal as part of the 524(g) Settlement discussed below, as of the date hereof, the other party has not yet dismissed its appeal, and the sale approved by the Settlement Approval Order has not been consummated. If the appeal of the Settlement Approval Order is not resolved earlier and as provided in <u>Section 10.9</u> of the Plan, the entry of the Confirmation Order shall constitute the Bankruptcy Court's or District Court's approval, as applicable, of the Certain Settling Insurers Agreement pursuant to Bankruptcy Rule 9019 and section 524(g) of the Bankruptcy Code.

On December 17, 2024, the Bankruptcy Court entered the *Agreed Order Continuing Hearing and Deadlines Solely as to Chubb Insurers Settlement Motion* [Docket No. 437], that continued the hearing as to the Chubb Insurers Settlement Motion to the omnibus hearing scheduled on March 20, 2025, and also adjourned indefinitely certain of the discovery/briefing deadlines related to the Chubb Insurers Settlement Motion that, as of November 29, 2024, had not yet expired. A copy of the November 29 Term Sheet was annexed to that agreed order. Hopeman subsequently adjourned the Chubb Insurers Motion to the omnibus hearing scheduled for June 18, 2025, after entering into the Settlement.

On December 20, 2024, the Court entered its Order approving the joint motion by Hopeman and the Committee to authorize judicial mediation (the "<u>Mediation</u>") of the Chubb Insurer Settlement Motion [Docket No. 443] (the "<u>Mediation Order</u>"), among other things, (i) appointing the Honorable Kevin R. Huennekens as judicial mediator (the "<u>Mediator</u>") and (ii) directing Hopeman, the Committee, and the related Chubb insurers (Century Indemnity Company and Westchester Fire Insurance Company) to mediate the relief sought in the Chubb Insurers Settlement Motion. Hopeman and the Committee consented to a request from HII to participate in the Mediation.

The Mediation resulted in the 524(g) Settlement -- an agreement between Hopeman, the Committee and HII but not an agreement with the Chubb Insurers. The 524(g) Settlement, *inter alia*, sets forth the essential terms on which Hopeman and the Committee agreed to resolve Hopeman's liability for Channeled Asbestos Claims, including that Hopeman and the Committee have agreed to jointly prosecute a chapter 11 plan that would create a trust pursuant to section 524(g) of the Bankruptcy Code, and Hopeman would transfer its remaining insurance coverage and cash to that trust to allow for resolution of the thousands of Channeled Asbestos Claims against Hopeman after the effective date of the contemplated plan. Among other terms, the Committee

also agreed to and subsequently dismissed its appeal of the a second interim order granting the Stay Extension Motion, which is described part VII below.

The Plan incorporates the terms of the 524(g) Term Sheet.

C. Unresolved Contentions

(a) <u>The Chubb Insurers</u>. The Chubb Insurers have contended that (a) Hopeman has failed to make a good faith effort to obtain Bankruptcy Court approval of the Chubb Insurers Settlement Motion in accordance with Hopeman's prepetition proposed settlement agreement with them, (b) the Plan contemplated by the 524(g) Settlement Term Sheet is not confirmable (*see* Docket No. 595), and (c) the Committee is not fulfilling its fiduciary obligations to its creditors (*see id.*), a contention the Committee vehemently denies. Each of these contentions are addressed below:

(i) Good Faith Effort. Hopeman contends that it has complied with its obligations by promptly seeking Bankruptcy Court approval of the proposed settlement through a motion filed on the Petition Date, through multiple conversations with the Committee and creditors to attempt to convince them to support the motion, through cooperation in extensive discovery by the Committee and other creditors about the proposed settlement, through Hopeman jointly moving with the Committee to conduct a judicial mediation concerning the Chubb Insurers Settlement Motion, through participating in the Mediation, and through post-Mediation conversations to attempt to get the Chubb Insurers to engage with the Committee on negotiating a final resolution of the Chubb Insurer Motion.

(ii) Confirmable 524(g) Plan. Hopeman expects that the Chubb Insurers will oppose confirmation of the Plan. Hopeman and the Committee contend that the Plan is confirmable under section 524(g) and will present evidence proving the Plan complies with all the requirements for confirmation of the Plan under the Bankruptcy Code at the confirmation hearing. <u>The Chubb Insurers are reserving</u> and not waiving any of their arguments as to the confirmability of the Plan.

(iii) Fulfilling Fiduciary Obligations. Hopeman and the Committee have jointly applied to the Court to appoint a Future Claimants' Representative to serve as the legal representative for parties making "demands" upon Hopeman or Reorganized Hopeman in accordance with section 524(g)(4)(B)(i).

(iv) Adversary Proceeding. The Chubb Insurers have filed a Complaint against the Debtor asserting claims for breach of contract and anticipatory breach of contract in adversary proceeding number 25-03015-KLP. In their Complaint, the Chubb Insurers claim that the Debtor has breached a prepetition settlement agreement it reached with the Chubb Insurers. The Chubb Insurers assert that they have a binding agreement with the Debtor that would prohibit the Debtor from prosecuting the proposed Plan, which the Chubb Insurers allege is inconsistent with the settlement agreement. The Chubb Insurers seek damages and administrative priority treatment for those damages, among other relief. The Debtor refers all parties to the Complaint itself for the particulars of the Chubb Insurers' allegations. The Debtor, on the other hand, contends that the Debtor has not breached the settlement agreement, that the settlement agreement is subject to Bankruptcy Court approval and has not yet been approved, that it is an executory contract that can be rejected in the bankruptcy with Court approval, and that any claim the Chubb Insurers may have for rejection or other breach of the agreement is a pre-petition general unsecured claim that must be resolved in the Court-approved claims process, rather than through an adversary proceeding. The Debtor intends to vigorously defend against the <u>Complaint.</u>

(v) Insurance Neutrality. The Chubb Insurers also assert that the Plan imposes burdens and duties upon the Chubb Insurers that are not required by the policies the Chubb Insurers issued to the Debtor. The Debtor contends the Plan is "insurance neutral" and does not impose any burdens or duties on the Chubb Insurers that they do not have already under the policies and that the obligations and defenses of the Chubb Insurers are dictated by applicable non-bankruptcy law, not the Plan

(b) <u>LMIC</u>. LMIC was Hopeman's primary liability insurance carrier. LMIC paid out millions of dollars on asbestos-related claims asserted against Hopeman and for Hopeman's costs. In various settlements Hopeman reached with LMIC, Hopeman released LMIC and sold back the primary liability insurance coverage to LMIC. Under those settlement agreements, Hopeman agreed to indemnify LMIC against certain claims and from certain sources. LMIC contends that Hopeman has failed to indemnify it in accordance with its obligations and has filed a proof of claim in this case, asserting Hopeman owes LMIC a minimum of \$317,254.89 in unpaid indemnity obligations. LMIC reserves the right in that proof of claim to assert additional sums are owed to it by Hopeman.

<u>LMIC also contends that its insurance coverage is unavailable to any Asbestos</u> <u>Claimant as a result of the settlements reached with Hopeman, and that Hopeman's</u> <u>prosecution of the Plan breaches Hopeman's contractual obligations to Liberty. Liberty</u> <u>argues that the Plan cannot transfer Hopeman's rights in the LMIC policies because</u> <u>Hopeman has no such rights.</u>

Hopeman contends that, under unambiguous provisions of its settlement agreements, LMIC's sole source of recovery for any breach by Hopeman of its indemnity or defense obligations to LMIC are funds that were held in a qualified settlement trust established under those settlement agreements. Those funds were depleted many years ago. Accordingly, even if Hopeman breached its obligations to LMIC, it has no liability to LMIC that would be a claim against Hopeman's bankruptcy estate. Hopeman intends to object to LMIC's proof of claim and to vigorously oppose any attempt by LMIC to seek a claim in this case. In Hopeman's view, LMIC is not a creditor with standing in this case. <u>LMIC, furthermore, lacks standing as an</u> <u>insurer of the Debtor to object to confirmation of the Plan and will not be aggrieved by the</u> <u>Plan, if confirmed.</u> (c) <u>Roussel & Clement</u>. Roussel & Clement ("<u>Roussel</u>") represents persons who have asserted asbestos-related claims against Hopeman. Roussel has opposed Hopeman's Stay Extension Motion and opposed various other motions of Hopeman, including the Certain Insurer Settlement Approval Motion. *See* Docket Nos. 86, 137, 405, 406, 423 and 592. After the Court entered the Settlement Approval Order over its objection, Roussel appealed that Order to the District Court. The appeal remains pending.

Hopeman contends that the evidence it offered in support of the Certain Insurer Settlement Agreement was substantial and unrebutted and that the Court was well within its authority to enter the Settlement Approval Order. The Bankruptcy Court entered a memorandum opinion denying a motion for stay pending appeal filed by HII, in which Roussel joined, and which Hopeman opposed. *See* Docket No. 526. Hopeman opposes the Roussel appeal and, for many of the reasons recited by the Bankruptey Court in its memorandum opinion denying the stay pending appeal, Hopeman believes it will prevail on appeal.

Furthermore, as explained above and set forth in <u>Section 10.9</u> of the Plan, confirmation of the Plan will obviate the need for the appeal because, if necessary, entry of the Confirmation Order shall constitute the Bankruptey Court's or District Court's approval, as applicable, of the Certain Settling Insurers Agreement pursuant to Bankruptey Rule 9019 and section 524(g) of the Bankruptey Code.

(c) (d) Other Parties' Contentions. Other parties have opposed certain relief sought by Hopeman in this case, including the Committee, HII, the Office of the U.S. Trustee, and certain other law firms representing asbestos-related claimants. The Committee and HII both agreed to the Section 524(g) Term Sheet and should support the Plan as long as it is consistent with that term sheet. The Plan Proponents do not know whether other parties will oppose confirmation of the Plan, but any such party with standing will have the opportunity to file an objection to confirmation of the Plan in accordance with the deadlines set forth in the Solicitation Procedures Order

VII.

Other Events During the Chapter 11 Case

A. First Day Motions and Orders

On the Petition Date, Hopeman filed motions seeking various forms of relief from the Bankruptcy Court to enable Hopeman to facilitate a smooth transition into chapter 11 (the "<u>First Day Motions</u>"). Hopeman asked that the Bankruptcy Court consider and approve the relief requested therein on an expedited basis at the hearing held on July 2, 2024 at 11:00 a.m. (prevailing Eastern Time). Each of the First Day Motions' requested relief were granted by the Bankruptcy Court and are briefly summarized below:

 continuing the use of Hopeman's cash management system, bank account, and business forms [Docket No. 5] (the "<u>Cash Management Motion</u>");

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 87 of 129

- listing addresses of legal counsel for personal injury claimants in the creditor matrix in lieu of such claimants' addresses and approving notice procedures for such claimants [Docket No. 6];
- waiving the requirement to submit a formatted mailing matrix and approving the form and manner of the Notice of Commencement of the Chapter 11 Case [Docket No. 4]; and
- extending the automatic stay to enjoin claims against certain non-debtor defendants in asbestos-related lawsuits filed, or to-be-filed, as applicable, in "direct action" states that otherwise would drain coverage available for Asbestos Claims asserted or to be asserted against Hopeman [Docket No. 7] (the "<u>Stay Extension Motion</u>").

Further, while not a motion, Hopeman also filed an application seeking authority to retain Kurtzman Carson Consultants, LLC dba Verita Global ("<u>Verita</u>") as Hopeman's claims, noticing, and administrative agent [Docket No. 3] (the "<u>Verita Retention Application</u>") to be considered by the Bankruptcy Court simultaneously with the First Day Motions.

The docket numbers for the orders granting the relief requested in each of the abovereferenced First Day Motions and the Verita Retention Application are 32, 33, 34, and 35, respectively. The Bankruptcy Court has entered two additional orders granting the relief requested in Stay Extension Motions on an interim basis and the docket numbers for such orders are 245 and 622. The third interim stay order, unless extended, expires June 30, 2025, or upon the effective date of the Plan should that occur earlier than June 30 or any extended expiration date.

B. Schedules and Statements and the Bar Date for Non-Asbestos Claims

On July 15, 2024, Hopeman timely filed its schedules of assets and liabilities ("<u>Schedules</u>") and statements of financial affairs ("<u>Statements</u>"). *See* Docket Nos. 59 and 60.

In an order dated September 12, 2024 (Docket No. 193), the Bankruptcy Court established, November 4, 2024, as the general bar date for the filing of Proofs of Claim for Claims *other than Asbestos Claims* and those belonging to governmental entities. This order also required that governmental units that are creditors must file proofs of claim by December 27, 2024. Under this order, holders of Asbestos Claims were not required to file proofs of claims against Hopeman. Likewise, any creditor whose Claim is set forth in the Schedules and is not identified therein as being contingent, unliquidated, or disputed, was permitted (but was not required) to file a Proof of Claim against Hopeman in order to participate in the Chapter 11 Case or receive a Distribution under the Plan.

C. Retention of Hopeman's Chapter 11 Professionals

The Bankruptcy Court has entered orders approving Hopeman's retention of the following professionals to assist Hopeman in carrying out its duties under the Bankruptcy Code during the Chapter 11 Case:

- Hunton Andrews Kurth, LLP, as Hopeman's Counsel;
- Stout Risius Ross, LLC, as financial advisor to Hopeman;

- Blank Rome, LLP, as Hopeman's special insurance counsel;
- Courington, Kiefer, Sommers, Marullo & Matherne, L.L.C., as Hopeman's special litigation counsel; and
- Kutak Rock LLP, as Hopeman's special conflicts counsel.

The docket numbers for the orders approving the retention of such professionals are 163, 164, 165, 187, and 509.

D. Formation of the Official Committee of Unsecured Creditors and Retention of Committee Professionals

On July 22, 2024, the United States Trustee for the Eastern District of Virginia appointed the Committee. *See* Docket No. 69. The Committee consists of the following five (5) individuals who assert Asbestos Personal Injury Claims against Hopeman: Nancy McComas-Doiron, represented by Peter Angelos Law; Veronica Miller, represented by MRHFM Law Firm; Darrell Kitchen, represented by Simmons Hanly Conroy; Melissa <u>BermanBeerman</u>, represented by Dean Omar Branham Shirley, LLP; and Donald M. Hoffman, Jr., represented by Stephen J. Austin, LLC.

The Bankruptcy Court has entered orders approving the Committee's retention of the following professionals to assist the Committee in carrying out its duties under the Bankruptcy Code during the Chapter 11 Case:

- Caplin & Drysdale, Chartered, as Committee's counsel;
- Morgan, Lewis & Bockius LLP, as Committee's special insurance counsel;
- FTI Consulting, as financial advisor to the Committee; and
- NERA Economic Consulting, Inc., as Committee's insurance allocation expert

The docket numbers for the orders approving the retention of such professionals are 202, 269, 270, and 416.

E. Exclusivity

Section 1121(b) of the Bankruptcy Code provides that, for a period of 120 days after the commencement of a chapter 11 case, a debtor has the exclusive right to file a plan (the "<u>Exclusive</u> <u>Plan Period</u>"). In addition, section 1121(c)(3) of the Bankruptcy Code provides that if a debtor files a plan within the Exclusive Plan Period, it has a period of 180 days after commencement of the chapter 11 case to obtain acceptances of such plan (the "<u>Exclusive Solicitation Period</u>" and, together with the Exclusive Plan Period, the "<u>Exclusive Periods</u>").

Pursuant to a second motion of Hopeman to extend the Exclusive Periods, by an order dated March 11, 2025 (Docket No. 523), the Bankruptcy Court extended the (i) Exclusive Filing Period through and including May 25, 2025, and (ii) the Exclusive Solicitation period through and including July 25, 2025

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 89 of 129

Further, as it pertains to the Exclusive Solicitation Period, Hopeman anticipates filing, contemporaneously with the filing of this Disclosure Statement or shortly thereafter, a motion seeking a combined hearing on approval of the adequacy of this Disclosure Statement and confirmation of the Plan and the solicitation procedures regarding the Plan, among other forms of relief requested therein (the "<u>Solicitation Motion</u>"). The Solicitation Motion will aid in ensuring the Plan Proponents work to obtain votes on the Plan, in accordance with the Bankruptcy Code, as soon as reasonably possible and within the Exclusive Solicitation Period.

F. Appointment of Future Claimants' Representative

On May <u>14</u>, 2025, the Bankruptcy Court approved a joint motion filed by Hopeman and Committee at Docket No. <u>732</u> to appoint Marla Rosoff Eskin, Esq. to act as legal representative to protect the shared interests of individuals who might assert asbestos-related demands against Hopeman in the future (the "Future Claimants' Representative"). <u>After the Court entered the order appointing the Future Claimants' Representative over their objection, the Chubb Insurers appealed that order to the District Court. The appeal remains pending. The Debtor and Committee will vigorously oppose the appeal and believe they will prevail.</u>

Hopeman anticipates that the Future Claimants' Representative will engage Professionals, subject to Bankruptcy Court approval, to assist the Future Claimants' Representative in carrying out her duties under the Bankruptcy Code.

G. U.S. Trustee Matters and Reports

Hopeman has filed, and will continue to file, reports including, among other things, information regarding its deposits, expenditures, and other relevant financial information in monthly operating reports prior to the Effective Date. After the Effective Date, Reorganized Hopeman will file quarterly post-confirmation reports with the Bankruptcy Court until the Chapter 11 Case is converted, dismissed, or closed, whichever occurs first.

VIII.

Summary of the Plan

This Section of the Disclosure Statement summarizes the treatment of Claims and Interests under the Plan and structure and means for implementation of the Plan and is qualified entirely by reference to the Plan and the definitions contained therein (and to any exhibits attached to the Plan) as this is only a summary of the Plan and therefore is not inclusive of each Article or Section of the Plan.

A. Treatment of Holders of Claims and Interests Under the Plan

Section 1123 of the Bankruptcy Code provides that a chapter 11 plan must classify the claims and interests of a debtor's creditors and equity interest holders. In accordance with section 1123 of the Bankruptcy Code, the Plan divides Claims and Interests into Classes and sets forth the treatment for each Class (other than Administrative Expense Claims and Priority Tax Claims, which, pursuant to section 1123(a)(1) of the Bankruptcy Code, need not be and have not been classified). Hopeman is also required, under section 1122 of the Bankruptcy Code, to classify

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 90 of 129

Claims against and Equity Interests in Hopeman into Classes that contain Claims and Equity Interests that are substantially similar to the other claims and Interests in such Class.

The classification of Claims and Equity Interests and the nature of Distributions to members of each Class are summarized in the table at the beginning of this Disclosure Statement and more fully below. The Plan Proponents believe that the consideration provided under the Plan to holders of Claims and Interests reflects an appropriate resolution of the Claims and Equity Interests. The Bankruptcy Court must find, however, that a number of statutory tests are met before the Plan may be confirmed. Many of these tests are designed to protect the interests of holders of Claims or Interests who will be bound by the provisions of the Plan if it is confirmed.

B. Administrative Expense Claims

Filing Administrative Expense Claims. The holder of an Administrative (a) Expense Claim, other than (a) a Professional Fee Claim covered by Section 2.2 of the Plan, (b) a liability incurred and payable in the ordinary course of business by Hopeman after the Petition Date that has been paid by Hopeman, or (c) an Administrative Expense Claim that has been Allowed and/or paid in full on or before the Effective Date, must file and serve on Reorganized Hopeman a request for payment of such Administrative Expense Claim pursuant to section 503(a) of the Bankruptcy Code so that it is received no later than the Administrative Expense Claim Bar Date. Holders required to file and serve but who fail to file and serve a request for payment of Administrative Expense Claims by the Administrative Expense Claim Bar Date shall be forever barred from asserting such Administrative Expense Claims against Hopeman, Reorganized Hopeman, and their property, and such Administrative Expense Claims shall be deemed waived and released as of the Effective Date. Notwithstanding the foregoing, pursuant to section 503(b)(1)(D) of the Bankruptcy Code, no Governmental Unit shall be required to file a request for payment of any Administrative Expense Claim of a type described in sections 503(b)(1)(B) or 503(b)(1)(C) of the Bankruptcy Code as a condition to such Claim being Allowed.

(b) <u>Allowance of Administrative Expense Claims</u>. An Administrative Expense Claim, with respect to which a request for payment has been properly and timely filed pursuant to <u>Section 2.1(a)</u> of the Plan, shall become an Allowed Administrative Expense Claim if no objection to such request is filed with the Bankruptcy Court and served on Reorganized Hopeman and the requesting party on or before the thirtieth (30th) calendar day after the Administrative Expense Claim Bar Date, as the same may be modified or extended from time to time by order of the Bankruptcy Court. If an objection is timely filed, the Administrative Expense Claim shall become an Allowed Administrative Expense Claim shall become an Allowed Administrative Expense Claim shall become an Allowed Administrative Expense Claim only to the extent Allowed by a Final Order or as such Claim is settled, compromised, or otherwise resolved pursuant to <u>Article VII</u>.

(c) <u>Payment of Allowed Administrative Expense Claims</u>. Except to the extent that an Administrative Expense Claim already has been paid during the Chapter 11 Case or the holder of an Allowed Administrative Expense Claim agrees to less favorable treatment, and except as provided for Professional Fee Claims in <u>Section 2.2</u> of the Plan, each holder of an Allowed Administrative Expense Claim against Hopeman shall receive, in full and complete settlement, release, and discharge of such Claim, Cash, to be paid from the Net Reserve Funds, equal to the unpaid amount of such Allowed Administrative Expense Claim on the latest of (i) the Effective Date or as soon thereafter as reasonably practicable; (ii) the first Business Day that is at least thirty

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 91 of 129

(30) calendar days after the date on which such Administrative Expense Claim becomes Allowed; and (iii) such other date as may be agreed to by such holder and Reorganized Hopeman or as otherwise ordered by the Bankruptcy Court; *provided*, *however*, that Allowed Administrative Expense Claims (other than a Professional Fee Claim covered by <u>Section 2.2</u>) representing liabilities incurred in the ordinary course of business by Hopeman, as Debtor in Possession, may be paid by Hopeman, as applicable, in the ordinary course of business, consistent with past practice and in accordance with the terms and subject to the conditions of any agreements governing, instruments evidencing, or other documents relating to such transactions.

C. Professional Fees

Each Professional requesting compensation pursuant to section(s) 327, 328, 330, 331, 363, 503(b), or 1103 of the Bankruptcy Code for services rendered in connection with the Chapter 11 Case before the Effective Date shall (a) file with the Bankruptcy Court, and serve on counsel to Reorganized Hopeman, an application for allowance of final compensation and reimbursement of expenses in the Chapter 11 Case on or before the date that is forty-five (45) calendar days after the Effective Date, and (b) after notice and a hearing in accordance with the procedures established by the Bankruptcy Code and the Bankruptcy Rules and any prior orders of the Bankruptcy Court in the Chapter 11 Case, be paid by Reorganized Hopeman, in Cash from the Net Reserve Funds, in such amounts as are Allowed by the Bankruptcy Court (i) no later than thirty (30) calendar days after the date upon which the order relating to any such Allowed Claim is entered or (ii) upon such other terms as may be mutually agreed upon between the holder of such an Allowed Claim and Reorganized Hopeman or as otherwise ordered by the Bankruptcy Court.

D. Priority Tax Claims

Except to the extent that the holder of an Allowed Priority Tax Claim has been paid by Hopeman prior to the Effective Date or agrees to less favorable treatment, each holder of an Allowed Priority Tax Claim shall receive, in full and complete settlement, release, and discharge of such Claim, Cash from the Net Reserve Funds in an amount equal to the unpaid portion of such Allowed Priority Tax Claim, on the latest of (i) the Effective Date, (ii) thirty (30) days after the date such Priority Tax Claim becomes an Allowed Claim, or as soon thereafter as is practicable, and (iii) the date such Allowed Priority Tax Claim becomes due and payable under applicable nonbankruptcy law.

E. Treatment of Classified Claims and Interests

The proposed treatment of all other Allowed Claims and Equity Interests is set forth in <u>Article IV</u> of the Plan and is summarized as follows:

1. Class 1 – Priority Non-Tax Claims

(a) <u>Classification</u>. Class 1 consists of all Priority Non-Tax Claims.

(b) <u>Treatment</u>. Except to the extent that the holder of an Allowed Priority Non-Tax Claim agrees to less favorable treatment, each holder of an Allowed Priority Non-Tax Claim shall receive, in full and complete settlement, release, and discharge of, and in exchange for, such Allowed Priority Non-Tax Claim, Cash to be paid from the Net Reserve Funds in an amount equal

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 92 of 129

to the Allowed Amount of such Claim on the later of (i) the Effective Date, and (ii) the date on which such Claim becomes Allowed, or, in each case, as soon as reasonably practicable thereafter.

(c) <u>Impairment and Voting</u>. Class 1 is Unimpaired under the Plan. Each holder of a Priority Non-Tax Claim is conclusively presumed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code and, accordingly, is not entitled to vote to accept or reject the Plan.

2. Class 2 – Secured Claims

(a) <u>Classification</u>. Class 2 consists of all Secured Claims.

(b) <u>Treatment</u>. Except to the extent that the holder of an Allowed Secured Claim agrees to less favorable treatment, on the Effective Date or as soon as reasonably practicable thereafter, each holder of an Allowed Secured Claim shall receive, at the option of Reorganized Hopeman and in full and complete settlement, release, and discharge of, and in exchange for, such Claim (i) payment in full in Cash to be paid from the Net Reserve Funds; (ii) the collateral securing such Allowed Secured Claim; or (iii) other treatment rendering such Claim Unimpaired.

(c) <u>Impairment and Voting</u>. Class 2 is Unimpaired under the Plan. Each holder of a Secured Claim is conclusively presumed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code and, accordingly, is not entitled to vote to accept or reject the Plan.

3. Class 3 – General Unsecured Claims

(a) <u>Classification</u>. Class 3 consists of all General Unsecured Claims.

(b) <u>Treatment</u>. Except to the extent that the holder of an Allowed General Unsecured Claim agrees to less favorable treatment, each holder of an Allowed General Unsecured Claim shall receive, in full and complete settlement, release, and discharge of, and in exchange for, such Allowed General Unsecured Claim, Cash in an amount equal to its Pro Rata share of the General Unsecured Recovery Pool on the later of (i) the Effective Date, and (ii) the date on which such Claim becomes Allowed, or, in each case, as soon as reasonably practicable thereafter. Solely for purposes of calculating Distributions to holders of Allowed General Unsecured Claims, on the Effective Date, all Disputed General Unsecured Claims will be treated as though they are Allowed in the amounts asserted or as estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code, and a reserve will be set aside for such Disputed General Unsecured Claims.

(c) <u>Impairment and Voting</u>. Class 3 is Impaired under the Plan. Each holder of a General Unsecured Claim is entitled to vote to accept or reject the Plan.

4. Class 4 – Channeled Asbestos Claims

(a) <u>Classification</u>. Class 4 consists of all Channeled Asbestos Claims.

(b) <u>Treatment</u>. As of the Effective Date, liability for all Channeled Asbestos Claims shall automatically, and without further act, deed, or court order, be channeled exclusively to and assumed by the Asbestos Trust in accordance with, and to the extent set forth in, <u>Articles VIII</u> and <u>X</u> of the Plan, the applicable Plan Documents, and the Confirmation Order. All Channeled

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 93 of 129

Asbestos Claims shall be treated pursuant to the terms of <u>Article VIII</u> of the Plan, the Asbestos Trust Agreement, and the Asbestos Trust Distribution Procedures. Except as provided in <u>Article VIII</u> of the Plan, pursuant to section 524(g) of the Bankruptcy Code, the Plan, and the Confirmation Order, Channeled Asbestos Claims shall be subject to the Asbestos Permanent Channeling Injunction.

(c) <u>Impairment and Voting</u>. Class 4 is Impaired under the Plan. Each holder of an Asbestos Claim is entitled to vote to accept or reject the Plan.

5. Class 5 – Equity Interests

(a) <u>Classification</u>. Class 5 consists of all Equity Interests.

(b) <u>Treatment</u>. On the Effective Date, the Equity Interests in Hopeman shall be cancelled, annulled, and extinguished.

(c) <u>Impairment and Voting</u>. Class 5 is Impaired under the Plan. Holders of Interests are deemed to reject the Plan pursuant to section 1126(g) of the Bankruptcy Code and, accordingly, are not entitled to vote to accept or reject the Plan.

F. Means for Implementation of the Plan

1. Generally

On and after the Confirmation Date, the Plan Proponents shall be empowered and authorized to take or cause to be taken, prior to the Effective Date, all actions necessary to enable them to implement the provisions of the Plan, including the creation of the Asbestos Trust and the preparations for the transfer of the Asbestos Trust Assets to the Asbestos Trust.⁶ On and after the Effective Date, Reorganized Hopeman shall be governed pursuant to the Amended Certificate of Incorporation and Amended By-Laws of Reorganized Hopeman.

2. The Asbestos Trust

(a) <u>Creation of the Asbestos Trust</u>. On the Effective Date, the Asbestos Trust shall be created in accordance with the Plan Documents, the Asbestos Trust Documents, and section 524(g) of the Bankruptcy Code. Subject to the provisions of the Plan, and in consideration of the transfer of the Asbestos Trust Assets to the Asbestos Trust, the Asbestos Trust shall assume all liabilities and responsibility for all Channeled Asbestos Claims, and, among other things, to: (1) direct the processing, liquidation, and payment of all compensable Channeled Asbestos Claims in accordance with the Plan, the Asbestos Trust Documents, and the Confirmation Order; (2) preserve, hold, manage, and maximize the assets of the Asbestos Trust for use in paying and satisfying Channeled Asbestos Claims; and (3) qualify at all times as a qualified settlement fund. The Asbestos Trust shall use the Asbestos Trust's assets and income to resolve Channeled Asbestos

⁶ "Asbestos Trust Assets" is defined in the Plan to mean (a) the Asbestos Trust Contribution; (b) the Asbestos Insurance Rights; (c) all other assets, rights (including Causes of Action), and benefits assigned, transferred or conveyed to the Asbestos Trust in connection with the Plan or any Plan Documents; and (d) all proceeds of the foregoing.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 94 of 129

Claims in accordance with the Asbestos Trust Agreement and the Asbestos Trust Distribution Procedures in such a way that holders of Channeled Asbestos Claims are treated fairly, equitably, and reasonably in light of the finite assets available to satisfy such claims, and shall otherwise comply in all respects with the requirements of a trust set forth in section 524(g)(2)(B) of the Bankruptcy Code. On the Effective Date, all right, title, and interest in and to the Asbestos Trust Assets, and any proceeds thereof, will be transferred to, and indefeasibly vested in, the Asbestos Trust, free and clear of all Claims, Demands, Equity Interests, Encumbrances, and other interests of any Entity, without any further action of the Bankruptcy Court or any Entity, but subject to Section 8.5 of the Plan and the remaining provisions of Section 8.3 of the Plan.

(b) <u>Transfer of the Asbestos Insurance Rights</u>.⁷ On the Effective Date, by virtue of Confirmation, without further notice, action, or deed, the Asbestos Insurance Rights shall be automatically transferred to, and indefeasibly vested in, the Asbestos Trust, and the Asbestos Trust shall thereby become the estate representative pursuant to sections 1123(a)(5) and 1123(b)(3)(B) of the Bankruptcy Code, with the exclusive right to enforce any and all of the Asbestos Insurance Rights against any Entity, subject to the provisions of <u>Section 8.13</u> and <u>Section 8.15</u> of the Plan, and the Proceeds of the recoveries of any such Asbestos Insurance Rights shall be the property of, and shall be deposited in, the Asbestos Trust. The Asbestos Insurance Rights shall be indefeasibly

⁷ "Asbestos Insurance Rights" is defined in the Plan to mean any and all of Hopeman's rights, title, privileges, interests, claims, demands, or entitlements in or to any insurance coverage, defense, indemnity, proceeds, payments, escrowed funds, initial or supplemental dividends, scheme payments, supplemental scheme payments, state guaranty fund payments, causes of action, and choses in action under, for, or related to (i) the Asbestos Insurance Settlements, (ii) the Asbestos Insurance Policies, or (iii) the Asbestos CIP Agreements, whether now existing or hereafter arising, accrued or unaccrued, liquidated or unliquidated, matured or unmatured, disputed or undisputed, fixed or contingent, including:

⁽a) any and all rights of Hopeman to pursue or receive payment, reimbursement, or proceeds under any Asbestos Insurance Policy or any Asbestos CIP Agreement, whether for indemnity, liability, defense costs, or otherwise;

⁽b) the remaining balance of any proceeds of any and all payments received by Hopeman on account of an Asbestos Insurance Policy or Asbestos CIP Agreement at any time on, before, or after the Effective Date, together with all interest earned on such proceeds;

⁽c) any and all rights of Hopeman to pursue or receive payments from any insolvent Asbestos Insurer, whether in receivership, liquidation, rehabilitation, runoff, or scheme of arrangement, or any other form of proceeding, or from any insolvent Asbestos Insurer's estate, and the remaining balance of any proceeds of all payments received by Hopeman from any such Asbestos Insurer or such insolvent insurer's estate, at any time on, before, or after the Effective Date, together with all interest earned on such proceeds;

⁽d) any and all rights of Hopeman to pursue or receive payments with respect to Asbestos Personal Injury Claims from any insurance guaranty association;

⁽e) any and all rights of Hopeman to pursue or receive payment to any exception to a workers' compensation exclusion in any Asbestos Insurance Policy; provided, however, that Asbestos Insurance Rights shall not include any rights or obligations under any insurance policy, settlement agreement, or coverage in-place agreement to which any Asbestos Insurer is a party to the extent, but only to the extent, that such rights or obligations pertain solely to coverage for Worker Compensation Claims; and

⁽f) any and all Extracontractual Claims, and any and all rights of Hopeman to pursue or receive payments or recoveries on account thereof.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 95 of 129

vested in the Asbestos Trust free and clear of all Claims, Demands, Equity Interests, Encumbrances, and other interests of any Entity.

(c) <u>Authority of the Asbestos Trust</u>. As of the Effective Date, without any further action of the Bankruptcy Court or any Entity, except as otherwise expressly set forth in the Plan, including, without limitation, the rights reserved to HII under <u>Section 8.15</u>, the Asbestos Trust shall be empowered to initiate, prosecute, enforce, sue on, defend, settle, compromise, and resolve (or decline to do any of the foregoing) all claims, rights, Causes of Action, suits and proceedings, whether in law or in equity, whether known or unknown, related to or arising from any asset, liability, or responsibility of the Asbestos Trust, including any actions arising from or related to the Asbestos Insurance Rights, in any court of competent jurisdiction consistent with applicable law.

(d) <u>Appointment of Asbestos Trustee</u>. On the Confirmation Date, the Bankruptcy Court shall appoint an individual designated by the <u>Asbestos Claimants</u> Committee and the Future Claimants' Representative to serve as the initial Asbestos Trustee. The designated individual will be identified in the Asbestos Trust Agreement and appointed pursuant to the Confirmation Order. Such appointment shall be effective as of the Effective Date. The individual to serve as the initial Asbestos Trustee will be identified in the Plan Supplement. All subsequent Asbestos Trustees shall be appointed in accordance with the terms of the Asbestos Trust Agreement. For purposes of performing the duties and fulfilling the obligations under the Asbestos Trust Agreement and the Plan, the Asbestos Trustee shall be deemed to be a party in interest within the meaning of section 1109(b) of the Bankruptcy Code.

(e) <u>Appointment of Delaware Trustee</u>. [Wilmington Trust, N.A.] has been selected by agreement of the <u>Asbestos Claimants</u> Committee and the Future Claimants' Representative to serve as the initial Delaware Trustee and will be identified in the Asbestos Trust Agreement and appointed pursuant to the Confirmation Order. All subsequent Delaware Trustees shall be appointed in accordance with the terms of the Asbestos Trust Agreement.

(f) <u>Appointment of Post-Effective Date Future Claimants' Representative</u>. On the Effective Date, the individual to be identified in the Plan Supplement shall be appointed, pursuant to the Plan, the Confirmation Order, and the Asbestos Trust Agreement, to serve as the Future Claimants' Representative on and after the Effective Date. Such Future Claimants' Representative shall have the functions, duties, and rights provided in, and shall serve in accordance with, the Asbestos Trust Agreement. In addition to the foregoing, such Future Claimants' Representative also may, at her option, participate in any: (1) appeal of the Confirmation Order; (2) hearing on a Professional Fee Claim; and (3) adversary proceeding pending on the Effective Date to which the Future Claimants' Representative was a party. Successor Future Claimants' Representatives will be appointed as provided in the Asbestos Trust Agreement.

(g) <u>Appointment of Asbestos Trust Advisory Committee</u>. Not later than ten (10) calendar days prior to the Confirmation Hearing, the <u>Asbestos Claimants</u> Committee shall nominate six (6) individuals to serve as the initial members of the Asbestos Trust Advisory Committee. The Confirmation Order shall constitute an order of the Bankruptcy Court appointing the initial members of the Asbestos Trust Advisory Committee. The Asbestos Trust Advisory

Committee shall have the functions, duties, and rights provided in, and shall serve in accordance with, the Asbestos Trust Agreement. Successor members of the Asbestos Trust Advisory Committee will be appointed as provided in the Asbestos Trust Agreement.

Transfer of Claims and Demands to the Asbestos Trust. On the Effective (h) Date, in consideration for the property transferred to the Asbestos Trust, and except as provided in Section 8.12, Section 8.13, and Section 8.15 of the Plan, all Channeled Asbestos Claims shall be transferred and channeled to, and assumed by, the Asbestos Trust pursuant to the Asbestos Permanent Channeling Injunction, and shall be resolved, liquidated, and (if eligible for payment) paid in accordance with the Asbestos Trust Agreement, the Asbestos Trust Distribution Procedures, and any other Asbestos Trust Document. The Asbestos Trust shall have no liability for any Claims other than Channeled Asbestos Claims and Asbestos Trust Expenses, and no Claims other than Channeled Asbestos Claims and Asbestos Trust Expenses shall be transferred and channeled to, or assumed by, the Asbestos Trust. Notwithstanding the Asbestos Trust's assumption of liability and responsibility for Channeled Asbestos Claims, such assumption shall not itself operate or be construed as a release, accord and satisfaction, mutual rescission, or novation of Hopeman's obligations on account of such Claims for purposes of any Asbestos Insurance Rights solely to the extent of actions or suits against Reorganized Hopeman directly in accordance with Section 8.12 of the Plan (subject, however, to the discharge of any "personal liability" of Hopeman as that term is used in section 524(a) of the Bankruptcy Code and as provided in Article X of the Plan).

(i) <u>Transfer of Rights and Defenses Related to Channeled Asbestos Claims</u>. On the Effective Date, all claims, defenses, rights and Causes of Action of Hopeman arising from or related to Channeled Asbestos Claims shall be transferred and assigned to the Asbestos Trust. In accordance with section 1123(b) of the Bankruptcy Code, the Asbestos Trust shall retain and may enforce such claims, defenses, rights, and Causes of Action relating to Channeled Asbestos Claims in any court of competent jurisdiction against any Entity other than a Protected Party, and shall retain and may enforce all defenses and counterclaims to all Asbestos Claims or Demands asserted against the Asbestos Trust, including setoff, recoupment, and any rights under section 502(d) of the Bankruptcy Code. The Asbestos Trust shall be deemed to be the appointed representative of Hopeman and Reorganized Hopeman, and may, pursue, litigate, compromise, and settle any rights, claims, or Causes of Action transferred to it, as appropriate.

(j) <u>Release</u>. As a condition to making any payment to a Channeled Asbestos Claimant, the Asbestos Trust shall obtain from that holder a release in the form of the Asbestos Personal Injury Claimant Release.

(k) <u>Consideration for Asbestos Permanent Channeling Injunction</u>. The assignment, transfer, and conveyance of the Asbestos Trust Assets to the Asbestos Trust on the Effective Date supports the imposition of the Asbestos Permanent Channeling Injunction in favor of all Protected Parties as of the Effective Date.

(1) <u>Books and Records</u>. On the Effective Date, Hopeman shall transfer to Reorganized Hopeman all of Hopeman's books and records (including electronic records) necessary for the Asbestos Trust to investigate and resolve Channeled Asbestos Claims in accordance with <u>Section 8.3</u> and <u>Section 8.16</u> of the Plan, the Asbestos Trust Agreement and

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 97 of 129

Asbestos Trust Distribution Procedures, including the books and records presently stored in Hopeman's warehouse in Waynesboro, Virginia, and in or in storage near the offices of Hopeman's prepetition claims administrator Special Claim Services, Inc. Notwithstanding anything to the contrary herein, holders of Channeled Asbestos Claims may pursue and obtain information stored in Hopeman's books and records (including electronic records) through discovery to the full extent permitted by applicable law. For the avoidance of doubt, privileges belonging to Hopeman on the Petition Date in such books and records shall belong to Reorganized Hopeman as of the Effective Date, and the Asbestos Trust's access to such books and records shall not result in the destruction or waiver of any applicable privileges pertaining to such books and records.

(m) <u>Institution and Maintenance of Legal and Other Proceedings</u>. From and after the Effective Date, the Asbestos Trust shall be empowered and entitled, in its sole and absolute discretion and at its own expense, to pursue, compromise, or settle all legal actions and other proceedings related to any asset, liability, or responsibility of the Asbestos Trust that is not released pursuant to the Plan.

(n) <u>Asbestos Trust Expenses</u>. The Asbestos Trust shall pay all Asbestos Trust Expenses from the Asbestos Trust Assets. The Plan Proponents, Hopeman's Estate, Reorganized Hopeman, the Protected Parties, and any of them, shall not have any obligation to pay any Asbestos Trust Expenses or any other liabilities of the Asbestos Trust. The Asbestos Trust shall promptly pay all Asbestos Trust Expenses incurred by Reorganized Hopeman for any liabilities, costs, or expenses as a result of taking any action on behalf of, and at the direction of, the Asbestos Trust.

(o) <u>Indemnification</u>. The Asbestos Trust shall, pursuant to the terms of the Asbestos Trust Agreement, indemnify and hold harmless the Protected Parties for any liability or alleged liability arising out of, or resulting from, or attributable to, a Channeled Asbestos Claim, including fines and penalties resulting from the Asbestos Trust's failure to comply with <u>Section</u> <u>8.14</u> of the Plan or the Asbestos Trust Agreement. Indemnification claims arising under <u>Section</u> <u>8.3(o)</u> of the Plan will not be subject to the Asbestos Trust Distribution Procedures.

(p) <u>Investment Policy</u>. Pursuant to the Asbestos Trust Agreement, all monies held in the Asbestos Trust shall be invested, subject to the investment limitations and provisions enumerated in the Asbestos Trust Agreement and shall not be limited to the types of investments described in section 345 of the Bankruptcy Code.

(q) <u>Excess Asbestos Trust Assets</u>. To the extent there are any Asbestos Trust Assets remaining at such time as the Asbestos Trust is dissolved, such excess Asbestos Trust Assets shall be transferred to a charity or charities in which the Asbestos Trustee has no financial interest or other connection for such charitable purposes as the Asbestos Trustee, in his or her reasonable discretion, shall determine, provided that, if practicable, the charity or charities to which such excess Asbestos Trust Assets are transferred shall be related to the treatment of, research on, or the relief of suffering of individuals suffering from asbestos-related disorders.

(r) <u>Dissolution of Asbestos Trust</u>. Upon dissolution of the Asbestos Trust: (1) the Asbestos Trustee, members of the Asbestos Trust Advisory Committee and the Future Claimants' Representative shall be released and discharged from all further authority, duties,

responsibilities, and obligations relating to and arising from and in connection with the Chapter 11 Case; and (2) the Asbestos Trust Advisory Committee shall be dissolved.

3. Net Reserve Funds

No later than the Effective Date, Hopeman or Reorganized Hopeman, as applicable, shall either establish a bank account or utilize an existing bank account of Hopeman for the purpose of holding only the Net Reserve Funds. Hopeman, Reorganized Hopeman, or their respective designees, as applicable, shall make any and all disbursements from the Net Reserve Funds that are provided in the Plan. The Net Reserve Funds shall be used for the sole purpose of paying or making Distributions in accordance with the Plan. Any Excess Net Reserve Funds shall be contributed by Reorganized Hopeman to the Asbestos Trust, and that contribution shall be part of the Asbestos Trust Contribution.

Net Reserve Funds shall consist of the lesser of (i) Cash in the amount to be disclosed in the Plan Supplement or (ii) the amount of Cash held by Hopeman as of the Effective Date, to be set aside by Hopeman on the Effective Date for the purpose of paying or making the Distributions contemplated by the Plan with respect to (a) Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Priority Claims, Allowed Secured Claims, and Allowed General Unsecured Claims (including the payment of any interest on any such Claims that may be allowed under the Plan or required to be paid by the Bankruptcy Code), (b) any fees and expenses of Hopeman, Reorganized Hopeman, the Committee and the Future Claimants' Representative that are payable by Hopeman or Reorganized Hopeman, as applicable, pursuant to the Plan, (c) any fees payable pursuant to section 1930 of title 28 of the United States Code (whether those fees pursuant to section 1930 of title 28 are payable before or after the Effective Date, (d) any other amounts that the Plan provides are to be paid from the Net Reserve Funds, and (e) such amounts as Reorganized Hopeman determines, in the reasonable exercise of its discretion, are or will be sufficient to fully satisfy (as and when due) all franchise taxes and other expenditures that are necessary to maintain its corporate existence in good standing under the laws of the state of its formation or that otherwise are necessary for Reorganized Hopeman to conduct its business after the Effective Date.

Any Excess Net Reserve Funds shall be contributed by Reorganized Hopeman to the Asbestos Trust, and that contribution shall be part of the Asbestos Trust Contribution.

4. Actions Against Reorganized Hopeman or Wayne to Obtain Benefits of Asbestos Insurance Coverage

(a) Except as otherwise permitted under <u>Section 8.16</u> of the Plan, on and after the Effective Date, a Channeled Asbestos Claimant shall have the right to initiate, commence, continue, or prosecute an action against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust), and, where permitted by applicable nonbankruptcy law, any Non-Settling Asbestos Insurer for Wayne, in a court of competent jurisdiction to obtain the benefit of Asbestos Insurance Coverage.

(b) If a Channeled Asbestos Claimant commences such an action on account of its Channeled Asbestos Claim, the complaint shall name Reorganized Hopeman (or, if deemed an

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 99 of 129

indispensable party, the Asbestos Trust) or any Non-Settling Asbestos Insurer for Wayne as a defendant and shall be deemed by operation of law to be an action against Reorganized Hopeman or any Non-Settling Asbestos Insurer for Wayne, as applicable. Such an action may be filed in any court where Hopeman was subject to *in personam* jurisdiction as of the Petition Date or any other court of competent jurisdiction. Any such action shall be served on the Asbestos Trust, which shall provide notice of such action, as appropriate, to all Non-Settling Asbestos Insurers. Notwithstanding the foregoing, Reorganized Hopeman, the Asbestos Trust, and Wayne shall have no obligation to answer, appear, or otherwise participate in the action in any respect other than as set forth in the Plan and as may be necessary to comply with applicable Asbestos Insurance Cooperation Obligations.

(c) Any liability of Reorganized Hopeman or Wayne to any Entity, including any Channeled Asbestos Claimant or Asbestos Insurer, that is based on, arises from, or is attributable to any action commenced under <u>Section 8.12</u> of the Plan shall be enforceable only against the Asbestos Insurance Coverage provided by the Non-Settling Asbestos Insurers and not against any other asset, including any other Asbestos Insurance Right, of the Asbestos Trust or Reorganized Hopeman.

5. Actions Against Non-Settling Asbestos Insurers to Obtain Benefits of Asbestos Insurance Coverage

(a) Except as otherwise permitted under <u>Section 8.13</u> or in <u>Section 8.15</u> of the Plan, the Asbestos Trust shall have the exclusive right to pursue, monetize, settle, or otherwise obtain the benefit of the Asbestos Insurance Rights, including with respect to any unpaid insurance Proceeds applicable to a judgment or settlement obtained or entered into by a Channeled Asbestos Claimant in accordance with <u>Section 8.12</u> of the Plan.

(b) If a Channeled Asbestos Claimant has entered into an enforceable settlement agreement with a Non-Settling Asbestos Insurer pertaining to his Channeled Asbestos Claim and such Non-Settling Asbestos Insurer has not timely paid or has refused to pay the amount provided in such settlement, such Channeled Asbestos Claimant may commence a breach-ofcontract action or other form of collection action against such Non-Settling Asbestos Insurer to recover the settlement payment owed.

(c) Any Channeled Asbestos Claimant who (1) has obtained a judgment against Reorganized Hopeman or Wayne in accordance with <u>Section 8.12</u> of the Plan, or (2) has the right under applicable nonbankruptcy law to name, join, or substitute as a defendant an Asbestos Insurer, may, to obtain the benefits of Asbestos Insurance Coverage, commence a judgment-enforcement action or a direct action against the relevant Non-Settling Asbestos Insurer ("**Insurance Policy Action**") in accordance with the terms of <u>Section 8.13</u> of the Plan, subject to the following conditions:

> (i) If any Non-Settling Asbestos Insurer against whom an Insurance Policy Action is brought asserts as a defense that it would have a claim as a result of contribution rights against one or more Settled Asbestos Insurers with respect to the Channeled Asbestos Claimant's claim that it could have asserted but for the Asbestos Permanent Channeling Injunction ("Contribution Claim"), the liability,

if any, of the Non-Settling Asbestos Insurer to the Channeled Asbestos Claimant shall be reduced dollar-for-dollar by the amount, if any, of any judgment establishing the Contribution Claim in accordance with <u>Section 8.13</u> of the Plan.

(ii) In determining the amount of any Contribution Claim that operates to reduce the liability of a Non-Settling Asbestos Insurer in any Insurance Policy Action, the Channeled Asbestos Claimant may assert the legal or equitable rights or defenses, if any, of the Settled Asbestos Insurers with respect to such Contribution Claims, and for purposes of <u>Section 8.13(c)(ii)</u> of the Plan, all Settled Asbestos Insurers, in exchange for their status as a Protected Party and for receiving the benefits of the Asbestos Permanent Channeling Injunction, shall be deemed to have transferred or assigned such legal or equitable rights or defenses to Channeled Asbestos Claimants; *provided* that the Channeled Asbestos Claimant shall not be permitted to argue that any Contribution Claims are not properly asserted against the Channeled Asbestos Claimant or that the Asbestos Permanent Channeling Injunction bars or affects in any way such Contribution Claims in connection with the Channeled Asbestos Claimant's claim against the Non-Settling Asbestos Insurer.

(iii) If a court reduces the amount of liability of a Non-Settling Asbestos Insurer in an Insurance Policy Action based on the share attributable to a Settled Asbestos Insurer consistent with <u>Section 8.13</u> of the Plan, the Channeled Asbestos Claimant whose judgment has been reduced may seek payment from the Asbestos Trust for all or a portion of the amount of the judgment reduction attributable to the Settled Asbestos Insurer's share, but only as permitted by and in accordance with the Asbestos Trust Distribution Procedures.

(iv) If the Asbestos Trust enters into an Asbestos Insurance Settlement with an Asbestos Insurer that is a party to a pending Insurance Policy Action, and if such Asbestos Insurance Settlement is approved by the Bankruptcy Court, the Channeled Asbestos Claimant pursuing such Insurance Policy Action shall (I) be deemed to release any rights transferred to it via the Trust Transfer, without further notice or action by any Entity, and (II) terminate its Insurance Policy Action with respect to that Asbestos Insurer. The Channeled Asbestos Claimant shall have the right to submit an Uninsured Asbestos Claim to the Asbestos Trust for payment in accordance with, and as permitted by, the Asbestos Trust Distribution Procedures, as set forth in <u>Section 8.16</u> of the Plan.

(v) The Asbestos Trust may seek to intervene in any Insurance Policy Action at any time. A Channeled Asbestos Claimant pursuing an Insurance Policy Action shall not object to or oppose any request or motion of the Asbestos Trust to intervene in such Insurance Policy Action. For purposes of establishing the grounds in favor of the Asbestos Trust's request or motion to intervene, the Confirmation Order shall include the following finding of the Bankruptcy Court: The Asbestos Trust shall have, and is deemed to have, an interest relating to the Asbestos Insurance Coverage that is the subject of any Insurance Policy Action, and shall be, and is deemed to be, so situated that disposing of the Insurance Policy Action may,

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 101 of 129

as a practical matter, impair or impede the Asbestos Trust's ability to protect its interest, and no party to the Insurance Policy Action can adequately represent that interest.

(d) In addition to the rights and remedies set forth in <u>Section 8.13</u> of the Plan, on and after the Effective Date, Channeled Asbestos Claimants may, only to the extent permitted or provided under applicable nonbankruptcy law, bring such Insurance Policy Actions against a Non-Settling Asbestos Insurer of Hopeman or Wayne with respect to potential liability of any Designated Person, subject to the terms and conditions set forth in <u>Section 8.13(c)</u> of the Plan. For the avoidance of doubt, no Designated Person shall be named as a defendant in any such Insurance Policy Actions.

(e) If a Channeled Asbestos Claimant intends to pursue an Extracontractual Claim against a Non-Settling Asbestos Insurer, the Channeled Asbestos Claimant shall send written notice to the Asbestos Trust, requesting the Asbestos Trust's leave to pursue such Extracontractual Claim.⁸ Within fourteen (14) calendar days after receiving the Channeled Asbestos Claimant's request, the Asbestos Trust shall respond in writing that it is either granting or denying the Channeled Asbestos Claimant's request.

> (i) The Asbestos Trust's notice to the Channeled Asbestos Claimant of its decision to grant the Channeled Asbestos Claimant's request to pursue an Extracontractual Claim shall legally operate as, or shall effect, to the extent permitted or authorized by applicable nonbankruptcy law and without further notice or action by any Entity, a transfer to the Channeled Asbestos Claimant of the Asbestos Trust's rights to pursue such Extracontractual Claim ("**Trust Transfer**"). The Channeled Asbestos Claimant shall thereupon have standing and authority to pursue, settle, or resolve such Extracontractual Claim within his or her absolute discretion, subject to the terms set forth in <u>Section 8.13</u> of the Plan.

> (ii) The Asbestos Trust may deny a Channeled Asbestos Claimant's request for leave to pursue an Extracontractual Claim if (I) the Asbestos Trust has determined that it is not able or permitted under applicable nonbankruptcy law to transfer or assign the Extracontractual Claim to the requesting Channeled Asbestos Claimant; (II) the Asbestos Trust intends, within ninety (90) calendar days after receiving the Channeled Asbestos Claimant's request, to commence litigation (or arbitration, to the extent required by the applicable Asbestos Insurance Policy or Asbestos CIP Agreement) against the relevant Non-Settling Asbestos Trust has already commenced such litigation or arbitration, which is pending or has been resolved. Such litigation commenced by the Asbestos Trust may include, or pertain to, multiple other Channeled Asbestos Claims or seek a declaration of rights

⁸ "Extracontractual Claim" is defined in the Plan to mean any claim against an Asbestos Insurer for "bad faith," extracontractual, or tort liability that is based on, arises from, or is attributable to an Asbestos Insurance Policy or Asbestos CIP Agreement.

generally with respect to the Non-Settling Asbestos Insurers' obligations in connection with Channeled Asbestos Claims.

6. No Actions Against Released Parties

In any action pursued in accordance with <u>Section 8.12</u> or <u>Section 8.13</u> of the Plan, Channeled Asbestos Claimants shall not name any Released Party as a defendant, or promptly shall dismiss from such action any Released Party previously named as a defendant. The Asbestos Trust shall indemnify any Released Party to the extent such Released Party is named as a defendant in such an action, and the Trust shall take all steps reasonably necessary to have any Released Party named in such an action dismissed from such action, including seeking relief from the Bankruptcy Court to enforce the terms of the Plan.

7. Special Provisions Pertaining to HII

Notwithstanding any provision of the Plan or any other Plan Document to the contrary, on and after the Effective Date, HII may, (a) to the extent it is the holder of one or more Asbestos Indirect Claims and (b) only to the extent permitted or provided under applicable nonbankruptcy law, file claims, crossclaims, or third-party demands in a court of competent jurisdiction (including in any state court lawsuits) against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or Wayne, or pursue direct actions against any Non-Settling Asbestos Insurer of Hopeman or Wayne, including to bring an action against a Non-Settling Asbestos Insurer of Hopeman or Wayne with respect to the potential liability of any Designated Person. For the avoidance of doubt, such Designated Persons will not be named as defendants in any such actions. Reorganized Hopeman, the Asbestos Trust, and Wayne shall have no obligation to answer, reply, appear, or otherwise participate in any action in which HII has filed a claim, crossclaim, thirdparty demand, or in any such direct action, other than as necessary to maintain coverage under the Asbestos Insurance Policies. Any judgment that may be obtained in connection with such a claim, crossclaim, third-party demand, or direct action cannot be enforced against the assets of Reorganized Hopeman or the Asbestos Trust, other than from the Asbestos Insurance Coverage. To the extent Hopeman's Asbestos Insurance Rights become subject to Asbestos Insurance Settlements and such Asbestos Insurance Rights are liquidated by the Trust, HII may submit Asbestos Indirect Claims to the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures. For the avoidance of doubt, the foregoing in no way limits HII's ability to seek payment or any other form of relief that may be made available according to the provisions contained within the Asbestos Trust Distribution Procedures, the Plan, or any other Plan Document. For the further avoidance of doubt, HII need not execute an Asbestos Personal Injury Claimant Release to obtain the benefits of Section 8.15 of the Plan. Notwithstanding any provision of the Asbestos Trust Distribution Procedures, the Plan or any other Plan Document to the contrary, the rights contained within this Section 8.15 of the Plan may not be impaired, impeded, abridged, or otherwise modified at any time (whether before or after entry of the Confirmation Order) without the prior written consent of HII; provided, however, that the foregoing shall not be construed to vest HII with consent or veto rights as to a proposed Asbestos Insurance Settlement described in Section 8.17 of the Plan, or to override, expand, or otherwise modify HII's consultation rights granted under Section 8.17 of the Plan.

8. Payments or Distributions from the Asbestos Trust

The sole and exclusive source of payment or recovery of a Channeled Asbestos Claimant on account of his Channeled Asbestos Claim shall be the Asbestos Insurance Coverage applicable to such Channeled Asbestos Claim, as provided in <u>Section 8.12</u>, <u>Section 8.13</u>, and Section 8.15 of the Plan, unless the Channeled Asbestos Claim (a) is an Uninsured Asbestos Claim, or (b) becomes an Uninsured Asbestos Claim before the Channeled Asbestos Claimant receives payment in full of any judgment obtained against Reorganized Hopeman or the Non-Settling Asbestos Insurer, or settlement reached with a Non-Settling Asbestos Insurer, in accordance with Section 8.12, Section 8.13, or Section 8.15 of the Plan. A Channeled Asbestos Claim shall become an Uninsured Asbestos Claim when (i) the Asbestos Trust has settled, in accordance with an Asbestos Insurance Settlement, all rights to the Asbestos Insurance Coverage applicable to the Channeled Asbestos Claim, or (ii) any Asbestos Insurance Coverage that otherwise may be applicable to such Channeled Asbestos Claim becomes unavailable due to exhaustion of the relevant Asbestos Insurance Coverage or due to a Final Order ruling on a coverage issue or defense, in which event such Channeled Asbestos Claimant may seek payment or distribution on account of his Channeled Asbestos Claim from the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures. Any treatment of a Channeled Asbestos Claim that is partially an Insured Asbestos Claim and partially an Uninsured Asbestos Claim shall be set forth in the Asbestos Trust Distribution Procedures. Notwithstanding any provision in the Plan or any other Plan Document to the contrary, a Channeled Asbestos Claimant shall not be entitled to receive a payment or distribution on account of his Channeled Asbestos Claim from the Asbestos Trust unless and until his Channeled Asbestos Claim is eligible for payment or distribution from the Asbestos Trust under the Asbestos Trust Distribution Procedures.

9. Settlements with Non-Settling Asbestos Insurers After the Effective Date

The Asbestos Trust shall have authority to enter into an Asbestos Insurance Settlement with any Asbestos Insurer on or after the Effective Date, but such Asbestos Insurance Settlement shall not be valid or enforceable unless and until (a) the Asbestos Trust Advisory Committee and the Future Claimants' Representative have consented to such Asbestos Insurance Settlement and determined in writing that such Asbestos Insurance Settlement is fair, reasonable, and sufficiently comprehensive to warrant that such Asbestos Insurer receive the protections of a Settled Asbestos Insurer under the Asbestos Permanent Channeling Injunction, any other applicable Plan Injunction, and section 524(g) of the Bankruptcy Code; (b) the Asbestos Trust has consulted with HII about the terms of such Asbestos Insurance Settlement prior to its submission to the Bankruptcy Court for approval, as provided herein; and (c) the Bankruptcy Court approves such Asbestos Insurance Settlement after notice to all parties directly affected by such Asbestos Insurance Settlement and a hearing thereon. To the extent necessary, the Asbestos Trust shall be allowed to seek to reopen the Chapter 11 Case to obtain the Bankruptcy Court's approval of any such settlement for good cause shown in accordance with section 350(b) of the Bankruptcy Code.

10. Insurance Neutrality

Nothing in the Plan, the Plan Documents, the Confirmation Order, any finding of fact and/or conclusion of law with respect to the confirmation of the Plan, or any order or opinion entered on appeal from the Confirmation Order shall limit the right of any insurer to assert any

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 104 of 129

coverage defense; *provided, however*, that (a) the transfer of rights in and under the Asbestos Insurance Rights to the Asbestos Trust is valid and enforceable and transfers such rights under the Asbestos Insurance Rights as Hopeman or Reorganized Hopeman may have, and that such transfer shall not affect the liability of any insurer, and (b) the discharge and release of Hopeman and Reorganized Hopeman, from all Claims and the injunctive protection provided to Hopeman, Reorganized Hopeman, and the Protected Parties with respect to Claims as provided herein shall not affect the liability of any insurer, except to the extent that any such insurer is a Settled Asbestos Insurer. Notwithstanding anything in Section 8.18 of the Plan to the contrary, nothing in Section 8.18 shall affect or limit, or be construed as affecting or limiting, (1) the binding effect of the Plan and the Confirmation Order on Hopeman, Reorganized Hopeman, the Asbestos Trust, or the beneficiaries of the Asbestos Trust or (2) the protection afforded to any Settled Asbestos Insurer by the Asbestos Permanent Channeling Injunction. Further, nothing in <u>Section 8.18</u> is intended or shall be construed to preclude otherwise applicable principles of res judicata or collateral estoppel from being applied against any insurer with respect to any issue that is actually litigated by such insurer as part of its objections to confirmation of the Plan.

11. Reorganized Hopeman Common Stock

On the Effective Date, all of the existing Equity Interests in Hopeman shall be cancelled, annulled, and extinguished, and 100% of the Reorganized Hopeman Common Stock shall be authorized and issued to the Asbestos Trust.

12. Corporate Governance of Reorganized Hopeman

On the Effective Date, (a) the current officers and directors of Hopeman shall be deemed to resign from their respective positions by operation of the Plan and (b) the individual(s) identified in a notice to be filed jointly by the Committee and the Future Claimants' Representative no later than two (2) days prior to the deadline established to accept or reject the Plan shall be appointed to serve as the officers and as the director of Reorganized Hopeman.

13. Restructuring Transactions and Continued Corporate Existence of Reorganized Hopeman

On or after the Confirmation Date, the Hopeman or Reorganized Hopeman may take such actions as Hopeman or Reorganized Hopeman determines to be necessary or appropriate to effectuate, implement and consummate the Restructuring Transactions set forth on Exhibit F to the Plan.

Except as otherwise provided in the Plan (and subject to the Restructuring Transactions), Hopeman will, as a reorganized debtor, continue to exist after the Effective Date as a separate corporate or other legal Entity, with all powers of a corporation or other legal Entity under applicable law and without prejudice to any right to alter or terminate such existence (whether by merger, dissolution or otherwise) under applicable state law. On and after the Effective Date, Reorganized Hopeman may operate its business and may use, acquire and dispose of property and compromise or settle any Claims without supervision or approval of the Bankruptcy Court and free of any restrictions of the Bankruptcy Code or Bankruptcy Rules, other than those restrictions expressly imposed by the Plan or Confirmation Order.

G. Effect of Confirmation

1. Dissolution of Committee; Discharge of the Future Claimants' Representative

Effective on the Effective Date, the Committee shall be dissolved automatically, whereupon its members, Professionals, and agents shall be released and discharged from any further authority, duties, obligations and responsibilities in the Chapter 11 Case and under the Bankruptcy Code.

Effective as of the dissolution of the Committee, the Asbestos Trust Advisory Committee shall succeed to, and exclusively hold, the attorney-client privilege and any other privilege held by the Committee and shall enjoy the work product protections that were applicable or available to the Committee before its dissolution.

Effective on the Effective Date, the Future Claimants' Representative shall be discharged from her duties in such capacity, whereupon the Future Claimants' Representative and her Professionals and agents shall be released and discharged from any further authority, duties, obligations and responsibilities in the Chapter 11 Case and under the Bankruptcy Code.

2. Vesting of Assets

Pursuant to section 1141(b) of the Bankruptcy Code, except as otherwise provided in the Plan, the Plan Documents or the Confirmation Order, the property of the Estate of Hopeman (except for the Asbestos Trust Contribution and any other property of Hopeman distributed pursuant to the Plan) shall vest in Reorganized Hopeman on the Effective Date free and clear of any and all Liens, Claims, Encumbrances and other interests of any Entity. From and after the Effective Date, Reorganized Hopeman may operate its business and may use, acquire, and dispose of property free of any restrictions imposed under or by the Bankruptcy Code, the Bankruptcy Rules, or the Bankruptcy Court.

3. Preservation of Certain Causes of Action, Rights to Settle Claims and Compromise Controversies, and Defenses

With the exception of those claims and rights transferred to the Asbestos Trust, including Causes of Action, pursuant to <u>Section 8.3</u> of the Plan or released pursuant to <u>Article X</u> of the Plan, in accordance with section 1123(b) of the Bankruptcy Code, Reorganized Hopeman, as successor in interest to Hopeman and its Estate, shall retain and may initiate, prosecute, enforce, sue on, settle, compromise, or resolve (or decline to do any of the foregoing), all claims, rights, causes of action, suits and proceedings, whether in law or in equity, whether known or unknown, accruing to, or that are property of, Hopeman and its Estate, pursuant to the Bankruptcy Code or any statute or legal theory, including any rights to, claims, or causes of action for, recovery under any policies of insurance issued to or on behalf of Hopeman other than the Asbestos Insurance Rights, in any court of competent jurisdiction consistent with applicable law.

4. Terms of Injunction and Automatic Stay

With the exception of the stay provided in the Bankruptcy Court's *Third Interim Order* Extending the Automatic Stay to Asbestos-Related Actions Against Non-Debtor Defendants

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 106 of 129

[Docket No. 622] that will (unless extended) expire on the earlier of June 30, 2025, or the Effective Date, all of the injunctions and/or stays in existence immediately prior to the Confirmation Date provided for in or in connection with the Chapter 11 Case, whether pursuant to section 105, 362, or any other provision of the Bankruptcy Code, the Bankruptcy Rules or other applicable law, shall remain in full force and effect until the injunctions set forth in the Plan become effective pursuant to a Final Order, and shall continue to remain in full force and effect thereafter as and to the extent provided by the Plan, the Confirmation Order, or by their own terms. For the avoidance of doubt, upon effectiveness of the injunctions set forth in the Plan, the automatic stay imposed by section 362 of the Bankruptcy Code shall be terminated. In addition, on and after the Confirmation Date, Reorganized Hopeman may seek such further orders as it may deem necessary or appropriate to preserve the status quo during the time between the Confirmation Date and the Effective Date.

Each of the injunctions contained in the Plan or the Confirmation Order shall become effective on the Effective Date and shall continue in effect thereafter unless otherwise provided in the Plan or the Confirmation Order.

5. No Liability for Certain Tax Claims

Unless a taxing authority has properly asserted a Claim against Hopeman on or before the Prepetition Non-Asbestos Claims Bar Dates, no Claim of that taxing authority shall be Allowed against Hopeman or Reorganized Hopeman for taxes, penalties, interest, additions to tax, or other charges arising out of the failure, if any, of Hopeman or any other Entity to have paid tax or to have filed any tax return (including, but not limited to, any income tax return or franchise tax return) in or for any prior year or arising out of an audit of any return for a period before the Petition Date.

6. No Successor Liability

Except as otherwise expressly provided in the Plan, Hopeman, Reorganized Hopeman, the other Protected Parties, and the Asbestos Trust do not, nor shall they be deemed to, assume, agree to perform, pay, or indemnify creditors for any liabilities or obligations of Hopeman relating to or arising out of the operations of, or assets of, Hopeman whether arising prior to or resulting from actions, events, or circumstances occurring or existing at any time prior to the Confirmation Date. None of Reorganized Hopeman, the other Protected Parties, or the Asbestos Trust is, or shall be, a successor to Hopeman by reason of any theory of law or equity, and none shall have any successor or transfere liability of any kind or character, except that Reorganized Hopeman and the Asbestos Trust shall assume the obligations specified expressly in the Plan and the Confirmation Order.

H. Exculpation, Injunctions, Releases and Settlement

1. Exculpation

None<u>From and after the Petition Date through the Effective Date, and with the</u> <u>exception of claims related to any act or omission that is determined in a Final Order to have</u> <u>constituted gross negligence, knowing and intentional fraud, or willful misconduct, none</u> of the Exculpated Parties shall have or incur any liability to any Entity for any act or omission in connection with, related to, or arising out of: (a) the Chapter 11 Case; (b) negotiation, formulation and preparation of the Plan and the other Plan Documents, and any of the terms and/or settlements and compromises reflected in the Plan and the other Plan Documents; (c) pursuit of confirmation of the Plan; (d) consummation of the Plan, or administration of the Plan or the property to be distributed under the Plan or the Asbestos Trust Distribution Procedures; <u>or (e)</u> the releases and injunctions contained in the Plan; or (f) the management or operation of Hopeman during the Chapter 11 Case. Without limiting the generality of the foregoing, the Exculpated Parties shall be entitled to and granted the protections and benefits of section 1125(e) of the Bankruptcy Code.

2. Discharge of Hopeman and Reorganized Hopeman

Except as specifically provided in the Plan, any of the other Plan Documents, or the Confirmation Order, pursuant to sections 524 and 1141(d)(1)(A) of the Bankruptcy Code, confirmation of the Plan shall discharge Hopeman and Reorganized Hopeman on the Effective Date from any and all Claims and Demands of any nature whatsoever, including, without limitation, all Claims, including, to the fullest extent permitted by law, Channeled Asbestos Claims, and liabilities that arose before the Confirmation Date and all debts of the kind specified in sections 502(g), 502(h) and 502(i) of the Bankruptcy Code whether or not: (a) a Proof of Claim based on such Claim was filed under section 501 of the Bankruptcy Code, or such Claim was listed on any of Hopeman's Schedules; (b) such Claim is or was allowed under section 502 of the Bankruptcy Code; or (c) the holder of such Claim has voted on or accepted the Plan. Except as otherwise specifically provided for in the Plan, as of the Effective Date, the rights provided in the Plan to holders of Claims, Demands and Equity Interests shall be in exchange for and in complete satisfaction, settlement and discharge of all Claims (including, to the fullest extent permitted by law, Asbestos Claims and Demands) against, Liens on, and Equity Interests in Hopeman, Reorganized Hopeman, and all of their respective assets and properties.

3. Hopeman Discharge Injunction

Except as specifically provided in the Plan (including Section 8.12, Section 8.13, Section 8.15, and Section 8.16 of the Plan), any of the other Plan Documents, or the Confirmation Order, all Entities who have held, hold, or may hold Claims (including, to the fullest extent permitted by law, Asbestos Claims and Demands) against Hopeman are permanently enjoined, on and after the Effective Date, from: (a) commencing or continuing in any manner any action or other proceeding of any kind against Hopeman, Reorganized Hopeman, or their respective property with respect to such Claim or Demand, other than to enforce any right to a Distribution pursuant to the Plan or any other right provided under the Plan; (b) enforcing, attaching, collecting, or recovering by any manner or means of any judgment, award, decree, or order against Hopeman, Reorganized Hopeman, or their respective property with respect to such Claim or Demand; (c) creating, perfecting, or enforcing any Encumbrance of any kind against Hopeman, Reorganized Hopeman, or their respective property with respect to such Claim or Demand; (d) asserting any right of setoff, subrogation, or recoupment of any kind against any obligation due to Hopeman or against the property or interests in property of Hopeman, with respect to such Claim or Demand; and/or (e) commencing or continuing any action, in any manner, against Hopeman, Reorganized Hopeman, or their respective property that does not comply with or is inconsistent with the provisions of the Plan or the Confirmation Order. The foregoing injunction shall extend to the successors of Hopeman (including, without limitation, Reorganized Hopeman) and their respective properties

and interests in property. The discharge provided in this provision shall void any judgment obtained against Hopeman at any time, to the extent that such judgment relates to a discharged Claim or Demand.

4. Asbestos Permanent Channeling Injunction

Pursuant to sections 105(a) and 524(g) of the Bankruptcy Code, and except as otherwise provided in the Plan (including <u>Article VIII</u> of the Plan), any of the other Plan Documents, and the Confirmation Order, the Confirmation Order shall provide for the issuance of the following injunction to take effect upon the occurrence of the Effective Date:

(a) <u>Scope of Injunction</u>. All Entities that have held or asserted, or hold or assert, or may hold or assert in the future any Channeled Asbestos Claim shall be permanently stayed, restrained, and enjoined from taking any action for the purpose of directly, indirectly, or derivatively collecting, recovering, or receiving payment, satisfaction, or recovery on account of any such Channeled Asbestos Claim, including:

(i) commencing, conducting, or continuing in any manner, directly, indirectly, or derivatively, any suit, action, or other proceeding of any kind (including a judicial, arbitration, administrative, or other proceeding) in any forum against or affecting any Protected Party or any property or interest in property of any Protected Party;

(ii) enforcing, levying, attaching (including any prejudgment attachment), collecting, or otherwise recovering by any means or in any manner, whether directly or indirectly, any judgment, award, decree, or other order against any Protected Party or any property or interest in property of any Protected Party;

(iii) creating, perfecting, or otherwise enforcing in any manner, directly or indirectly, any Encumbrance against any Protected Party or any property or interest in property of any Protected Party;

(iv) setting off, seeking reimbursement of, contribution from, or subrogation against, or otherwise recouping in any manner, directly or indirectly, any amount against any liability owed to any Protected Party or any property or interest in property of any Protected Party; and

(v) proceeding in any manner and in any place with regard to any matter that is within the scope of the matters designated by the Plan to be subject to resolution by the Asbestos Trust, except in conformity and compliance with the Asbestos Trust Agreement and the Asbestos Trust Distribution Procedures.

(b) <u>Reservations</u>. Notwithstanding anything to the contrary above, this Asbestos Permanent Channeling Injunction shall not stay, restrain, bar, or enjoin:

(i) the rights of Entities to the treatment accorded them under <u>Article</u> <u>IV</u> of the Plan, as applicable, including the rights of Channeled Asbestos Claimants

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 109 of 129

to have their Channeled Asbestos Claims resolved in accordance with the Asbestos Trust Distribution Procedures;

(ii) the rights of Entities to assert any Channeled Asbestos Claim against the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures, or any claim for payment of an Asbestos Trust Expense against the Asbestos Trust;

(iii) the rights of Channeled Asbestos Claimants to assert any and all claims or causes of action against any Entities that are not Protected Parties;

(iv) the rights of the Asbestos Trust or, if applicable, Reorganized Hopeman to prosecute a claim or cause of action based on, arising from, or attributable to any of the Asbestos Trust Assets against any Entity that is not a Protected Party;

(v) any action or suit against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or Wayne that strictly conforms to the requirements of <u>Section 8.12</u> of the Plan.

(vi) any action, suit, or Claimant Action permitted or authorized under <u>Section 8.13</u> of the Plan against any Non-Settling Asbestos Insurer; or

(vii) any claim, crossclaim, or third-party demand in a court of competent jurisdiction (including in any state court lawsuits) against Reorganized Hopeman (or, if deemed an indispensable party, the Asbestos Trust) or Wayne, or any direct action against any Non-Settling Asbestos Insurer of Hopeman or Wayne, in each case that strictly conforms to the requirements of <u>Section 8.15</u> of the Plan.

5. Releases by Hopeman and Estate

Except as otherwise expressly provided in the Plan or Confirmation Order, on the Effective Date, for good and valuable consideration, the adequacy of which is confirmed, Hopeman, Reorganized Hopeman, and any Entity seeking to exercise the rights of the Estate, in each case individually and collectively, including any successor to Hopeman or any Estate representative appointed or selected pursuant to the applicable provisions of the Bankruptcy Code, shall, and shall be deemed to, completely and forever release, waive and discharge unconditionally each of the Released Parties, solely in their capacities as such, from any and all claims, obligations, suits, judgments, remedies, damages, Demands, debts, rights, Causes of Action and liabilities that are (or that, at any time on or prior to the Effective Date, were) property of the Estate, whether known or unknown, liquidated or unliquidated, fixed or contingent, foreseen or unforeseen, matured or unmatured, now existing or hereafter arising, in law, equity or otherwise, whether direct, indirect, or derivative, based upon, attributable to, or arising out of, in whole or in part, any act or omission, transaction, or occurrence taking place on or prior to the Effective Date (including prior to the Petition Date); provided, however, that nothing contained in Section 10.5 of the Plan is intended to operate as a release of liability based upon gross negligence or willful misconduct as determined by a Final Order.

6. <u>Release of Hopeman's Directors and OfficersMutual Releases</u> by Releasing Parties<u>in Addition to Protections under the Asbestos Permanent Channeling</u> <u>Injunction</u>

Except as provided in <u>Section 8.13(c</u>) and <u>Section 8.15</u> of the Plan and in addition to the protections afforded to the Released Parties as Protected Parties under the Asbestos Permanent Channeling Injunction, and in consideration for the obligations of Hopeman and Reorganized Hopeman under the Plan, each Releasing Party and the Asbestos Trust shall waive and release any and all Causes of Action that such holder,- did commence or could have commenced against any such Released Party that is based upon, attributable to, or arising from any acts or omissions of Released Party occurring prior to the Effective Date or in any way attributable to Hopeman, Reorganized Hopeman, the Chapter 11 Case, or the Plan; provided, however, that, notwithstanding the foregoing, no release of the Released Parties shall diminish, reduce, or eliminate the duties of any Asbestos Insurer under any Asbestos Insurance Policy or any Asbestos CIP Agreement.

In exchange for holders of Non-Asbestos Claims or Equity Interests opting in to grant the release provided in Section 10.6 of the Plan, each Released Party shall waive and release any and all Causes of Action that such Released Party did commence or could have commenced against any such holders of Non-Asbestos Claims or Equity Interests that optedin that is based upon, attributable to, or arising from any acts or omissions of Releasing Parties occurring prior to the Effective Date in any way attributable to Hopeman, Reorganized Hopeman, the Chapter 11 Case, or the Plan; provided, however, that, notwithstanding the foregoing, no Released Party is waiving or releasing any rights or claims under any insurance policies that provide any type of coverage to such Released Party.

Releasing Party means collectively: (a) all holders of <u>Asbestos</u> Claims that vote to accept or are presumed to accept the Plan; and (b) all holders of <u>Claims that abstain from voting on</u> the Plan and who do not<u>Non-Asbestos Claims or Equity Interests who</u> affirmatively opt out of<u>in to</u> the releases provided by the Plan by checking the box on the applicable form indicating that they opt not<u>in</u> to grant the releases provided in the Plan in accordance with the procedures set forth in the Solicitation Procedures Order; (e) all holders of Claims and Interests that vote to reject the Plan or are deemed to reject the Plan and who do not affirmatively opt out of the releases provided by the Plan by checking the box on the applicable form indicating that they opt not to grant the releases provided in the Plan in accordance with the procedures set forth in the Solicitation Procedures Order; and (d) with respect to Hopeman and each of the foregoing Entities in clauses (a) through (c), such Entity and its current and former affiliates, and such Entities' and their current and former affiliates' current and former directors, managers, officers, equity holders (regardless of whether such interests are held directly or indirectly), interest holders, predecessors, successors, and assigns.

7. Releases of Avoidance Actions and Estate Causes of Action Against Vendor Released Parties

As of the Effective Date, each of Hopeman, Reorganized Hopeman and the Asbestos Trust hereby fully, finally, and forever releases, relinquishes, and discharges all Avoidance

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 111 of 129

Actions and all other Claims and Causes of Action that it holds against any Vendor Released Parties.

Vendor Released Parties means those Entities listed on <u>Exhibit G</u> to the Plan, each of which is a vendor, including professionals, that provided services to Hopeman prior to the Petition Date.

8. No Actions on Account of Released Claims

Except as provided in <u>Section 8.12</u> of the Plan, as of the Effective Date, all <u>EntitiesReleasing Parties</u> that have held, currently hold or may hold any claims, commitments, obligations, suits, judgments, damages, demands, debts, Causes of Action or liabilities that are released pursuant to the Plan shall be permanently enjoined from taking any of the following against a <u>released EntityReleased Party</u>, or any of its property, on account of such released claims, commitments, obligations, suits, judgments, damages, demands, debts, Causes of Action or liabilities: (i) commencing or continuing in any manner any action or other proceeding; (ii) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order; (iii) creating, perfecting or enforcing any Encumbrance; (iv) asserting a setoff, right of subrogation or recoupment of any kind against any debt, liability or obligation due to any <u>released EntityReleased Party</u>; and (v) commencing or continuing any action, in any manner, in any place that does not comply with or is inconsistent with the provisions of the Plan.

9. Disallowed Claims

On and after the Effective Date, Hopeman and the Estate shall be fully and finally discharged from any liability or obligation on a Disallowed Claim, and any order creating a Disallowed Claim that is not a Final Order as of the Effective Date solely because of an Entity's right to move for reconsideration of such order pursuant to section 502 of the Bankruptcy Code or Bankruptcy Rule 3008 shall nevertheless become and be deemed to be a Final Order on the Effective Date. The Confirmation Order, except as otherwise provided therein or herein, shall constitute an order: (a) disallowing all Claims (other than Asbestos Claims) to the extent such Claims are not allowable under any provision of section 502 of the Bankruptcy Code, including, but not limited to, time-barred Claims, and Claims for unmatured interest, and (b) disallowing or subordinating, as the case may be, any Claims, or portions of Claims, for penalties or non-compensatory damages.

10. Compromise and Settlement of Claims and Interests and Approval of Certain Settling Insurers Agreement

Pursuant to section 1123 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration of the Distributions and other benefits provided under the Plan, the provisions of the Plan shall, on the Effective Date, constitute a compromise and settlement of all claims and controversies relating to the rights that a holder of a Claim or Interest may have against Hopeman with respect to any Claim, Interest, or Distribution on account thereof, as well as all potential Causes of Action against Hopeman. To the extent necessary, the entry of the Confirmation Order shall constitute approval of the Certain Settling Insurers Agreement and the settlement contemplated thereby. The entry of the Confirmation Order will constitute the Bankruptcy Court's

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 112 of 129

approval of such compromises and settlements and the Bankruptcy Court's finding that such compromises and settlements are (i) in the best interest of Hopeman, the Estate, and holders of Claims and Interests; and (ii) fair, equitable, and reasonable. Hopeman or Reorganized Hopeman, as applicable, is authorized to take all actions necessary or appropriate to consummate the transactions and settlements contemplated thereby, including by the Certain Settling Insurers Settlement.

I. Executory Contracts and Unexpired Leases

1. General Treatment

As of the Effective Date and except as otherwise provided in the Plan (including with respect to <u>Sections 6.2</u> and <u>Section 8.3</u> of the Plan) or in any contract, instrument, release or other agreement or document entered into in connection with the Plan, all Executory Contracts to which Hopeman is a party are hereby rejected, except for any Executory Contract that (a) has previously been rejected pursuant to a Final Order of the Bankruptcy Court, (b) is the subject of a separate rejection motion filed by Hopeman under section 365 of the Bankruptcy Code before the Confirmation Date, (c) is an Asbestos Insurance Policy or Asbestos CIP Agreement, or (d) is a Non-Asbestos Insurance Policy. The Confirmation Order shall constitute the Bankruptcy Court's approval of the rejection of the contracts and leases rejected hereby.

2. Asbestos Insurance Agreements

For the avoidance of doubt, none of the Asbestos Insurance Policies or Asbestos CIP Agreements are being rejected, altered, or otherwise modified pursuant to the Plan, and all parties' respective rights, duties, defenses, obligations, and liabilities thereunder are hereby preserved, except to the extent of an Asbestos Insurance Policy or Asbestos CIP Agreement that is the subject of and only to the extent contemplated by and provided for in an Asbestos Insurance Settlement and only to the extent approved pursuant to the entry of an order by the Bankruptcy Court or the District Court.

Furthermore, Hopeman does not believe that any of the Asbestos Insurance Policies or Asbestos CIP Agreements constitute Executory Contracts. To the extent any of the Asbestos Insurance Policies or Asbestos CIP Agreements are Executory Contracts, then, notwithstanding anything contained in the Plan to the contrary, the Plan will constitute a motion to assume any such Asbestos Insurance Policy or Asbestos CIP Agreement. Subject to the occurrence of the Effective Date, the entry of the Confirmation Order will constitute approval of such assumption pursuant to section 365(a) of the Bankruptcy Code and a finding by the Bankruptcy Court that each such assumption is in the best interest of Hopeman and its Estate. Unless otherwise determined by the Bankruptcy Court pursuant to a Final Order or agreed to by the parties thereto prior to the Effective Date, no payments are required to cure any defaults of Hopeman existing as of the Confirmation Date with respect to each such Asbestos Insurance Policy or Asbestos Insurance Policy or Asbestos Insurance Policy or Asbestos CIP Agreement.

For purposes of clarity, while Hopeman also does not believe that any such agreements constitute Executory Contracts, Hopeman is not assuming any prepetition settlement agreement (or any related indemnity obligations thereunder) that do not currently provide rights in favor of Hopeman to continuing coverage or payment of insurance proceeds.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 113 of 129

In addition, nothing in the Plan, any Plan Supplement, or any other document related to or made as an exhibit to the Plan is intended to or shall limit the right of any Asbestos Insurer or Non-Asbestos Insurer to assert any insurance coverage defense available under the applicable Asbestos Insurance Policy to any Channeled Asbestos Claim, or the applicable Non-Asbestos Insurance Policy to any non-Channeled Asbestos Claim, as appropriate, asserted against Hopeman or administered by the Asbestos Trust.

3. Bar Date to Rejection Damages

In the event that the rejection of an Executory Contract by Hopeman or Reorganized Hopeman pursuant to the Plan results in damages to the nondebtor party or parties to such Executory Contract, a claim for such damages shall be forever barred and shall not be enforceable against Hopeman, Reorganized Hopeman or any of their respective properties or interests in property, and the nondebtor party or parties to such Executory Contract shall be barred from receiving any Distribution under the Plan on account of such Claim, unless a Proof of Claim with respect to such damages is filed with the Bankruptcy Court and served upon counsel for Hopeman or Reorganized Hopeman, as applicable, on or before (a) if such Executory Contract is rejected pursuant to <u>Section 6.1</u> of the Plan, thirty (30) days after entry of the Confirmation Order; or (b) if such Executory Contract is rejected pursuant to a Final Order of the Bankruptcy Court granting a motion filed by Hopeman to reject such Executory Contract, thirty (30) days after entry of such order. The notice of the Effective Date to be delivered pursuant to Bankruptcy Rules 2002 and 3020(c) shall set forth the bar date for filing rejection damages claims under <u>Section 6.2</u> of the Plan and constitute notice thereof.

J. Procedures for Resolving and Treated Disputed Claims Other than Asbestos Claims and Professional Fee Claims

1. Disputed Claims

All Disputed Claims against Hopeman, other than Channeled Asbestos Claims and Professional Fee Claims, shall be subject to the provisions of <u>Article VII</u> of the Plan.

2. Objection to Claims

Except as otherwise provided in the Plan (including Section 4.2 and Section 4.3), Hopeman or Reorganized Hopeman, as the case may be, shall be entitled to file objections to Claims that have been brought in the Bankruptcy Court or should properly have been brought in the Bankruptcy Court but were brought in other forums (other than Channeled Asbestos Claims), on or before the date that is one hundred and twenty (120) days after the Effective Date (unless such day is not a Business Day, in which case such deadline shall be the next Business Day thereafter), as the same may be extended from time to time by the Bankruptcy Court. In addition, Hopeman or Reorganized Hopeman, as the case may be, may before the expiration of such period request that the Bankruptcy Court estimate any contingent or unliquidated Claim (not including any Channeled Asbestos Claim) for any reason pursuant to section 502(c) of the Bankruptcy Code, regardless of whether Hopeman previously objected to such Claim or whether the Bankruptcy Code, such Claim at any time, including during the pendency of litigation concerning any objection to

any Claim or of any appeal relating thereto. Hopeman and Reorganized Hopeman shall be authorized to settle, compromise, withdraw or litigate to judgment such objections without further approval of the Bankruptcy Court.

3. Payments and Distributions with Respect to Disputed Claims

Notwithstanding any other provision of the Plan, no payments or Distributions shall be made with respect to all or any portion of a Disputed Claim unless and until all objections to such Disputed Claim have been settled or withdrawn or have been determined by Final Order and the Disputed Claim, or some portion thereof, has become an Allowed Claim.

K. Miscellaneous Provisions

The Plan contains provisions relating to corporate actions, delivery of Distributions, manner of paying Distributions, conditions precedent to confirmation and consummation of the Plan, retention of jurisdiction by the Bankruptcy Court, modifications of the Plan, payment of statutory fees, binding effect of the Plan, compliance with tax requirements, governing law, and timing, among other things. For more information regarding these items, see the Plan annexed hereto.

IX.

Alternatives To The Plan

If the Plan is not confirmed and consummated, the alternatives to the Plan include either (a) a dismissal of this Chapter 11 Case, (b) liquidation of Hopeman under chapter 7 of the Bankruptcy Code, or (c) an alternative chapter 11 plan in accordance with the requirements of chapter 11 of the Bankruptcy Code. As discussed below, the Plan Proponents do not believe that any of these alternatives, even if viable, would afford holders of Claims or Equity Interests a greater recovery than what is provided by the Plan.

A. Dismissal of Hopeman's Chapter 11 Case.

If the Plan cannot be confirmed, Hopeman could elect to dismiss this Chapter 11 Case. Hopeman anticipates that such a dismissal would trigger a "race to the courthouse" among holders of Asbestos Claims, thereby eliminating any likelihood of an equality of distribution among similarly situated holders of Asbestos Claims.

In addition, the cost of defending against such actions would likely exhaust Hopeman's remaining funds in the next twelve to eighteen months and deplete insurance coverage for defense costs that may remain available under the Asbestos Insurance Policies. At this time, Hopeman has a limited amount of funds on hand and no other significant assets except for its rights under the Asbestos Insurance Policies. Hopeman may lack the funds to fully litigate its Asbestos Insurance Rights and other claims and interests arising in connection with its Asbestos Insurance Policies on a case-by-case basis outside of this Chapter 11 Case.

B. Liquidation under Chapter 7 of the Bankruptcy Code

If no plan can be confirmed, the Chapter 11 Case may be converted to a liquidation case under chapter 7 of the Bankruptcy Code. Following a conversion to chapter 7, a chapter 7 trustee would be appointed to liquidate the assets of Hopeman and make distributions in accordance with the priorities established by the Bankruptcy Code. In a chapter 7 liquidation, there likely would be no framework resembling the Plan for creating a trust that (as is proposed in the Plan) would take responsibility for administering Asbestos Claims through the Asbestos Trust Distribution Procedures and for making distributions to eligible claimholders from assets of the Asbestos Trust. Bankruptcy courts are generally not permitted to adjudicate Asbestos Claims, so, instead, the validity and amount of all such claims would have to be determined in another forum. The Plan Proponents believe that liquidation under chapter 7 would also result in smaller distributions being made to creditors than those provided for them in the Plan because, among other things, in a chapter 7 case, the Estate would bear the costs of the chapter 7 trustee's compensation, pursuant to section 326 of the Bankruptcy Code, and the fees of chapter 7 professionals who would be unfamiliar with events in the Chapter 11 Case.

The Bankruptcy Code requires as a condition to confirmation of a plan of reorganization that the Bankruptcy Court find that, with respect to an Impaired Class (such as Class 3 (General Unsecured Claims) and Class 4 (Channeled Asbestos Claims)), each holder of a claim or interest in such class has accepted the plan or will receive or retain under the plan on account of such claim or interest property of a value, as of the effective date of the plan, that is not less than the amount that such holder would receive if the debtor were liquidated under chapter 7 of the Bankruptcy Code. Attached to this Disclosure Statement as <u>Exhibit B</u> is a liquidation analysis for Hopeman (the "Liquidation Analysis"), which compares the proposal in the Plan with a hypothetical liquidation assuming that (i) a bankruptcy case under chapter 7 of the Bankruptcy Code is commenced immediately, (ii) Hopeman's assets are liquidated by a court-appointed chapter 7 trustee in an orderly liquidation, and (iii) the Asbestos Claims are resolved through liquidation in a non-bankruptcy forum rather than pursuant to the Plan and the Asbestos Trust Distribution Procedures. The Liquidation Analysis demonstrates that holders of Claims in Class 3 (General Unsecured Claims) and Asbestos Claims in Class 4 (Channeled Asbestos Claims) will receive as much value under the Plan, if not more, than they would receive in a chapter 7 liquidation.

The Liquidation Analysis is based on a number of estimates and assumptions that, while considered reasonable, are inherently beyond the control of Hopeman or any chapter 7 trustee. Accordingly, there can be no assurances that the values reflected in the Liquidation Analysis would be realized if Hopeman were to undergo a chapter 7 liquidation. Instead, actual results could vary materially from those shown in the Liquidation Analysis. In addition, any liquidation necessarily would take place in the future, under circumstances that presently cannot be predicted. Accordingly, if the remaining unliquidated assets of the Estate were liquidated at a later date by a chapter 7 trustee, the actual liquidation proceeds could be materially lower or higher than the amounts set forth in <u>Exhibit B</u>, and no representation or warranty can be made with respect to the actual proceeds that may or could be received in a chapter 7 liquidation.

C. Alternate Chapter 11 Plan.

If the Plan is not confirmed, then Hopeman or any other party in interest could attempt to formulate a different chapter 11 plan. However, the Plan has been proposed by the Plan Proponents taking into consideration the competing and conflicting interests held by the holders of Channeled Asbestos Claims, Hopeman's other creditors, the Future Claimants' Representative and the priorities established in the Bankruptcy Code. The Plan is the result of substantial and lengthy negotiations among the Plan Proponents.

The Plan Proponents believe that confirmation and implementation of the Plan is preferable to any of these other alternatives and should provide greater recoveries than those available in a dismissal of this Chapter 11 Case, a liquidation under chapter 7 of the Bankruptcy Code, or an alternative chapter 11 plan.

X.

Risk Factors

Prior to voting to accept or reject the Plan, holders of Claims against and Equity Interests in Hopeman should read and consider carefully the risk factors set forth below, as well as the other information set forth in this Disclosure Statement (and the documents delivered together with and/or referred to in this Disclosure Statement). These risk factors, however, should not be regarded as constituting the only risks involved in connection with the Plan and its implementation.

A. Certain Bankruptcy Considerations

Although the Plan Proponents believe that the Plan will satisfy all requirements necessary for confirmation by the Bankruptcy Court, there can be no guarantee that the Bankruptcy Court will reach the same conclusion, or that the Confirmation Order, if challenged on appeal, will be affirmed. There also can be no assurance that the Plan as proposed will be accepted by the requisite number of holders or amount of Claims, that the Plan will not be modified up to and including the Confirmation Date, or that the Bankruptcy Court will enter an order confirming, or recommending confirmation of, the Plan containing the findings of fact and conclusions of law that are conditions precedent to confirmation of the Plan. There also can be no assurance that the District Court will accept and affirm, or issue, the order confirming the Plan, that such acceptance and affirmance or issuance will become a Final Order, and that the Asbestos Permanent Channeling Injunction will therefore become valid and enforceable.

If the Plan is not confirmed and consummated, there can be no assurance that the Chapter 11 Case will continue rather than be converted to a liquidation, or that any alternative chapter 11 plan would be on terms as favorable to the holders of Claims and Equity Interests as the terms of the Plan. If a liquidation or protracted reorganization were to occur, there is a substantial risk that the value of Hopeman's assets would erode substantially – to the detriment of all stakeholders.

B. Projected Financial Information

The financial projections that accompany this Disclosure Statement are dependent upon numerous assumptions, including confirmation and consummation of the Plan in accordance with

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 117 of 129

its terms, the anticipated future performance of Reorganized Hopeman, general business and economic conditions, and other matters, many of which are beyond Hopeman's control. Accordingly, there can be no assurance that such assumptions will prove to be valid. In addition, unanticipated and unforeseeable events or circumstances occurring subsequent to the preparation of the financial projections may affect the actual financial results of the Reorganized Debtor. Although the Plan Proponents believe that the projections are reasonable and attainable, some or all of the estimates will vary, and variations between the actual financial results and those that are projected may be material.

C. Distributions Under the Asbestos Trust Distribution Procedures

There can be no certainty as to the precise amounts that will be distributed by the Asbestos Trust in any particular time period or when Asbestos Claims will be paid by the Asbestos Trust. Payments that will be made on Asbestos Claims will be determined under the Asbestos Trust Distribution Procedures and will be based, on the one hand, upon estimates of the number, types, and amount of current Asbestos Claims and expected future Demands and, on the other hand, the value of the assets of the Asbestos Trust, the liquidity of the Asbestos Trust, the Asbestos Trust's expected future income and expenses, and other matters that are likely to affect the sufficiency of funds to pay all holders of Asbestos Claims.

The initial payment percentage will be determined after the Effective Date by comparing the assets of the Asbestos Trust against its projected liability for Channeled Asbestos Claims and Asbestos Trust Expenses. The Asbestos Trust's projected liability for Channeled Asbestos Claims is based upon a number of assumptions. Should any assumption from which the initial payment percentage was developed prove to be materially inaccurate based upon the Asbestos Trust's actual experience, the Asbestos Trust may have to adjust the payment percentage upwards or downwards from time to time, pursuant to the Asbestos Trust Documents, to reflect current estimates of the Asbestos Trust's assets and liabilities.

D. The Asbestos Permanent Channeling Injunction

The Asbestos Permanent Channeling Injunction, which, among other things, bars the assertion of any Channeled Asbestos Claims against Hopeman, Reorganized Hopeman, and the other Protected Parties (except as provided in the Plan), is the cornerstone of the Plan. In 1994, the United States Congress added subsection (g) to section 524 of the Bankruptcy Code in order to confirm the authority of the Bankruptcy Court, subject to the conditions specified therein, to issue injunctions, such as the Asbestos Permanent Channeling Injunction, with respect to present and future claims for asbestos-induced injury or wrongful death. Although the Plan, the Asbestos Trust Agreement, and the Asbestos Trust Distribution Procedures all have been drafted with the intention of complying with section 524(g) of the Bankruptcy Code, and although satisfaction of the conditions imposed by section 524(g) is a condition precedent to confirmation of the Plan, there is no guarantee that the validity and enforceability of the Asbestos Permanent Channeling Injunction to Channeling Injunction section 524(g) or the application of the Asbestos Permanent Channeling Injunction to Channeled Asbestos Claims will not be challenged, either before or after confirmation of the Plan. Although the Plan Proponents believe adequate bases exist for the courts to uphold section 524(g) and the Asbestos Permanent Channeling Injunction, there can be no assurance that, in the

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 118 of 129

future, courts might not invalidate all or a portion of section 524(g) or the Asbestos Permanent Channeling Injunction.

XI.

Tax Consequences of the Plan

HOLDERS OF CLAIMS AND EQUITY INTERESTS SHOULD SEEK ADVICE BASED UPON THEIR PARTICULAR CIRCUMSTANCES FROM AN INDEPENDENT TAX ADVISOR.

THE FOLLOWING DISCUSSION SUMMARIZES CERTAIN FEDERAL INCOME TAX CONSEQUENCES OF THE IMPLEMENTATION OF THE PLAN TO HOPEMAN, REORGANIZED HOPEMAN, HOLDERS OF ASBESTOS CLAIMS, AND THE ASBESTOS TRUST. THE FOLLOWING SUMMARY DOES NOT DISCUSS THE FEDERAL INCOME TAX CONSEQUENCES TO HOLDERS WHOSE CLAIMS ARE ENTITLED TO PAYMENT IN FULL IN CASH OR ARE OTHERWISE UNIMPAIRED UNDER THE PLAN OR TO HOLDERS OF INTERESTS OR INTERCOMPANY CLAIMS.

THE FOLLOWING SUMMARY IS BASED UPON THE INTERNAL REVENUE CODE OF 1986 (AS AMENDED, THE "INTERNAL REVENUE CODE"), TREASURY REGULATIONS PROMULGATED THEREUNDER, JUDICIAL DECISIONS, AND PUBLISHED ADMINISTRATIVE RULES AND PRONOUNCEMENTS OF THE IRS AS IN EFFECT ON THE DATE HEREOF. CHANGES IN SUCH RULES OR NEW INTERPRETATIONS THEREOF MAY HAVE RETROACTIVE EFFECT AND COULD SIGNIFICANTLY AFFECT THE FEDERAL INCOME TAX CONSEQUENCES DESCRIBED BELOW.

THE FEDERAL INCOME TAX CONSEQUENCES OF THE PLAN ARE COMPLEX AND ARE SUBJECT TO SIGNIFICANT UNCERTAINTIES. HOPEMAN DOES NOT CURRENTLY INTEND TO SEEK A RULING FROM THE IRS CONCERNING ANY OF THE TAX ASPECTS OF THE PLAN. IN ADDITION, THIS SUMMARY DOES NOT ADDRESS FOREIGN, STATE, OR LOCAL TAX CONSEQUENCES OF THE PLAN. NOR DOES IT PURPORT TO ADDRESS THE FEDERAL INCOME TAX CONSEQUENCES OF THE PLAN TO SPECIAL CLASSES OF TAXPAYERS (SUCH AS FOREIGN TAXPAYERS, BROKER-DEALERS, BANKS, MUTUAL FUNDS, INSURANCE COMPANIES, FINANCIAL INSTITUTIONS, THRIFTS, SMALL BUSINESS INVESTMENT COMPANIES, REGULATED INVESTMENT COMPANIES, REAL ESTATE INVESTMENT COMPANIES, TAX-EXEMPT ORGANIZATIONS, TRADERS IN SECURITIES THAT ELECT TO USE A MARK-TO-MARKET METHOD OF ACCOUNTING FOR THEIR SECURITY HOLDING AND PASS-THROUGH ENTITIES AND INVESTORS IN SUCH ENTITIES).

ACCORDINGLY, THE FOLLOWING SUMMARY OF CERTAIN FEDERAL INCOME TAX CONSEQUENCES IS FOR INFORMATIONAL PURPOSES ONLY AND IS NOT A SUBSTITUTE FOR CAREFUL TAX PLANNING AND ADVICE BASED UPON THE INDIVIDUAL CIRCUMSTANCES PERTAINING TO A HOLDER OF A CLAIM OR INTEREST. ALL HOLDERS OF CLAIMS AND INTERESTS ARE URGED TO CONSULT

THEIR OWN TAX ADVISORS FOR THE FEDERAL, STATE, LOCAL, AND OTHER TAX CONSEQUENCES APPLICABLE TO THEM UNDER THE PLAN.

A. Treatment of Transfer of Asbestos Trust Assets and Taxation of the Asbestos Trust

It is currently intended that the Asbestos Trust will constitute a "qualified settlement fund" within the meaning of section 468B of the Internal Revenue Code and the Treasury regulations promulgated thereunder. The applicable Treasury regulations provide that to be treated as a qualified settlement fund, a fund, account, or trust must be (i) established pursuant to an order of, or be approved by, a government authority, including a court, and must be subject to the continuing jurisdiction of that government authority; (ii) established to resolve or satisfy one or more contested or uncontested claims that have resulted or may result from an event or a related series of events that has occurred and that has given rise to at least one claim asserting, among other things, liability under the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, 42 U.S.C. § 9601 et seq., or liability arising out of a tort, breach of contract, or violation of law; and (iii) a trust under applicable state law or have its assets physically segregated from the other assets of the transferor and persons related to the transferor.

A transferor generally is entitled to a current federal income tax deduction for all transfers of cash and other property (other than its own notes) to a qualified settlement fund to the same extent the transferor would have been entitled to a deduction if such amounts had been paid directly to the holder of a claim that will be discharged upon the establishment of the qualified settlement fund. Assuming the Asbestos Trust is treated as a qualified settlement fund, and (to the extent permissible by law) Hopeman intends to claim a current federal income tax deduction for transfers of Cash to the Asbestos Trust to the same extent that Hopeman would have been entitled to a deduction if such amounts had been paid directly to the holder of an Asbestos Claim. In connection therewith, Hopeman generally will not be entitled to a deduction to the extent that it funds the Asbestos Trust with Cash attributable to amounts not included in its income.

Assuming the Asbestos Trust is treated as a qualified settlement fund, the Asbestos Trust will generally be subject to a separate entity level tax at the maximum rate applicable to trusts and estates, and, in determining the taxable income of the Asbestos Trust, (i) any amounts transferred to the Asbestos Trust to resolve or satisfy a liability for which the Asbestos Trust is established generally will be excluded from the Asbestos Trust's income, and (ii) administrative costs (including state and local taxes) incurred by the Asbestos Trust generally will be deductible.

Assuming the Asbestos Trust is treated as a qualified settlement fund, trade or business expenses generally will not be deductible for federal income tax purposes. In general, the adjusted tax basis of property received (or treated as received for federal income tax purposes) by a qualified settlement fund from a transferor pursuant to the Plan will be the fair market value of such property at the time of receipt.

B. Consequences to Holders of Channeled Asbestos Claims

Except as provided for in <u>Section 8.12</u>, <u>Section 8.13</u> and <u>Section 8.15</u> of the Plan, each Channeled Asbestos Claim will be resolved and, if eligible, satisfied in cash from the Asbestos Trust in accordance with the Asbestos Trust Distribution Procedures. The federal income tax

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 120 of 129

treatment of the receipt of payments from the Asbestos Trust by a holder of such an Asbestos Claim generally will depend upon the nature of the Asbestos Claim. Because the amounts received by a holder of an Asbestos Claim (other than an Asbestos Trust Expense) generally will be attributable to, and compensation for, such holder's personal physical injuries or sickness, within the meaning of section 104 of the Internal Revenue Code, any such amounts received by the holder generally should be nontaxable. However, to the extent payments from the Asbestos Trust to a holder of an Asbestos Claim are attributable to medical expense deductions allowed under section 213 of the Internal Revenue Code for a prior taxable year, such payments will be taxable as ordinary income to the recipient. To the extent that the payments from the Asbestos Trust to holders of Asbestos Claims constitute amounts received on account of claims other than personal physical injury or sickness, such payments generally will be includable in the gross income of such holders.

C. Information Reporting and Withholding

In connection with the Plan, Reorganized Hopeman and the Asbestos Trust will comply with all applicable withholding and reporting requirements imposed by federal, state, local, and foreign taxing authorities, and all Distributions under the Plan will be subject to those withholding and reporting requirements. Creditors may be required to provide certain tax information as a condition to receiving Distributions pursuant to the Plan. Notwithstanding any other provision of the Plan, each Person receiving a Distribution pursuant to the Plan will have sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding, and other tax obligations, on account of that Distribution.

The federal income tax consequences to any particular holder of a Claim also may be affected by matters not discussed in this Disclosure Statement. For example, if a holder of a Claim previously deducted a loss with respect to its Claim, the receipt of money under the Plan may require such holder to recognize income. Furthermore, certain holders of Claims, including foreign persons, life insurance companies, and tax-exempt organizations may be subject to special rules not addressed in this Disclosure Statement. There also may be state, local, or foreign tax considerations applicable to each holder.

XII.

Plan Voting Instructions, Procedures and Deadline

Along with this Disclosure Statement, Hopeman has included a copy of the Plan and an individual ballot for holders of General Unsecured Claims and either an individual ballot or a master ballot for holders of Asbestos Claims, all of whom are impaired and entitled to vote for the acceptance or rejection of the Plan. A Claim is impaired if the Plan proposes to modify, alter or exchange its legal, contractual or equitable rights.

A. Claim Holders Entitled to Vote

Administrative Expense Claims and Priority Tax Claims have not been classified for purposes of voting or receiving Distributions. Rather, all such Claims are treated separately as unclassified Claims, and the holders thereof are not entitled to vote on the Plan.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 121 of 129

Claims in Class 1 (Priority Claims) and Class 2 (Secured Claims) are Unimpaired. Holders of Claims in Class 1 (Priority Claims) and Class 2 (Secured Claims) are conclusively presumed to have accepted the Plan and, accordingly, are not entitled to vote to accept or reject the Plan.

Claims in Class 3 (General Unsecured Claims) and Class 4 (Channeled Asbestos Claims) are Impaired. Each holder of a General Unsecured Claim in Class 3 or a Channeled Asbestos Claim in Class 4 is entitled to vote to accept or reject the Plan ("<u>Voting Creditors</u>").

Claims in Class 6 (Equity Interests) also are Impaired. Each holder of an Interest in Class 6 (Equity Interests) will not receive or retain any property on account of their Equity Interests. As a result, such holders are conclusively presumed to have rejected the Plan and, accordingly, are not entitled to vote to accept or reject the Plan.

B. Voting Instructions and Voting Deadline

Pursuant to the Solicitation Procedures Order, all Voting Creditors will be sent a Ballot together with the solicitation version of this Disclosure Statement. Please read the Ballot carefully and follow the instructions contained therein. Please use only the Ballot that accompanies the Disclosure Statement to cast your vote.

Hopeman has engaged Verita as its voting agent (the "<u>Voting Agent</u>") to assist in the transmission of voting materials and in the tabulation of votes on the Plan. FOR YOUR VOTE TO BE COUNTED, YOUR VOTE MUST BE RECEIVED BY THE VOTING AGENT AT THE ADDRESS SET FORTH BELOW OR THROUGH THE E-BALLOT PORTAL ON OR BEFORE THE VOTING DEADLINE OF 4:00 P.M. (PREVAILING EASTERN TIME) ON ________ UNLESS EXTENDED BY HOPEMAN.

IF THE INSTRUCTIONS ON YOUR BALLOT REQUIRE YOU TO RETURN THE BALLOT TO YOUR ATTORNEYS, YOU MUST DELIVER YOUR BALLOT TO THEM IN SUFFICIENT TIME FOR THEM TO PROCESS IT AND RETURN IT TO THE BALLOTING AGENT BEFORE THE VOTING DEADLINE.

IF A BALLOT IS DAMAGED OR LOST, YOU MAY CONTACT THE VOTING AGENT AT THE NUMBER SET FORTH BELOW TO RECEIVE A REPLACEMENT BALLOT. ANY BALLOT THAT IS EXECUTED AND RETURNED BUT WHICH DOES NOT INDICATE AN ACCEPTANCE OR REJECTION OF THE PLAN WILL NOT BE COUNTED.

IF YOU HAVE ANY QUESTIONS CONCERNING VOTING PROCEDURES, YOU MAY CONTACT THE VOTING AGENT AT:

Hopeman Ballot Processing Center c/o Kurtzman Carson Consultants LLC dba Verita Global 222 N. Pacific Coast Highway, Suite 300 El Segundo, CA 90245 Telephone: (877) 709-4752 (U.S./Canada) or (424) 236-7232 (International) Submit an inquiry at https://www.veritaglobal.net/hopeman/inquiry Additional copies of this Disclosure Statement and the Plan are available upon request made to the Voting Agent, at the telephone number or e-mail address set forth immediately above.

C. Summary of Voting Procedures for Holders of Asbestos Claims

The Solicitation Order sets forth certain procedures governing the voting by holders of Asbestos Claims ("<u>Voting Procedures</u>"). Specifically, pursuant to the Voting Procedures, each Asbestos Claim is to be temporarily allowed for voting purposes only in the amount of \$1.00 per claimant.

Hopeman does not have individual addresses for the majority of the holders of Asbestos Claims, most of whom have asserted Claims against Hopeman through their law firms. To avoid unnecessary administrative burden, the Plan Proponents will send a master ballot in the form attached to this Disclosure Statement as **Exhibit E**, to all counsel of known holders of Asbestos Claims. The Plan Proponents will send an individual ballot (in the form attached to this Disclosure Statement as **Exhibit D**) to individual holders of Asbestos Claims only where the circumstances set forth in the Voting Procedures for such delivery are met.

XIII.

Confirmation of the Plan

A. Confirmation Hearing

Section 1128(a) of the Bankruptcy Code requires the Bankruptcy Court, after appropriate notice, to hold a hearing on confirmation of a plan (the "<u>Confirmation Hearing</u>"). The Solicitation Procedures Order scheduled the Confirmation Hearing for

B. General Requirements of Section 1129 of the Bankruptcy Code

At the Confirmation Hearing, the Bankruptcy Court will determine whether the following confirmation requirements specified in section 1129 of the Bankruptcy Code have been satisfied:

- 1. The Plan complies with the applicable provisions of the Bankruptcy Code.
- 2. Hopeman has complied with the applicable provisions of the Bankruptcy Code.
- 3. The Plan has been proposed in good faith and not by any means proscribed by law.
- 4. Any payment made or promised by Hopeman or by a person acquiring property under the Plan for services or for costs and expenses in, or in connection with, the Chapter 11 Case, or in connection with the Plan and incident to the Chapter 11 Case, has been or will be approved by the Bankruptcy Court.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 123 of 129

- 5. Hopeman has disclosed the identity and affiliations of any individual proposed to serve, after confirmation of the Plan, as a director or officer of Hopeman or a successor to Hopeman under the Plan. The appointment of (or continuance by) such individual to (or in) such position or office is consistent with the interests of creditors and equity security holders and with public policy. Hopeman has disclosed the identity of any insider that will be employed or retained by Hopeman, and the nature of any compensation for such insider.
- 6. With respect to each Class of Claims or Equity Interests, each holder of an impaired Claim or impaired Interest either has accepted the Plan or will receive or retain under the Plan, on account of such holder's Claim or Interest, property of a value, as of the Effective Date, that is not less than the amount such holder would receive or retain if Hopeman were liquidated on the Effective Date under chapter 7 of the Bankruptcy Code. *See* discussion of "Best Interests Test," below.
- 7. Each Class of Claims and Equity Interests has either accepted the Plan or is not Impaired under the Plan.
- 8. Except to the extent that a holder of a particular Claim has agreed to different treatment of such Claim, the Plan provides that Administrative Expense Claims and Priority Claims (other than Priority Tax Claims) will be paid in full on the Effective Date. Priority Tax Claims, if any, will be paid, over a period not exceeding five years after the date of the order for relief, in regular installments of Cash payments, equal to the Allowed amount of such Claims (as of the Effective Date) and in a manner no less favorable than the most favored non-priority unsecured Claim provided for by the Plan, and, with respect to a secured Claim that would otherwise meet the description of an unsecured Claim of a governmental unit undersection 507(a)(8) of the Bankruptcy Code, but for the secures status of that Claim, the holder of that Claim will receive on account of that Claim, Cash payments, in the same manner as described in this paragraph.
- 9. At least one Class of Impaired Claims has accepted the Plan, with the determination of such Class acceptance not including any acceptance of the Plan by any insider holding a Claim in such Class.

C. Best Interests Test

The Bankruptcy Code requires that each holder of an impaired Claim or Equity Interest either (i) accept the Plan; or (ii) receive or retain under the Plan property of a value, as of the Effective Date, that is not less than the value such holder would receive if Hopeman was liquidated under chapter 7 of the Bankruptcy Code.

The Liquidation Analysis is discussed above in <u>Section V.B</u>. of this Disclosure Statement and is attached hereto as <u>Exhibit B</u>. When the results of the Liquidation Analysis are compared to

the distributions expected under the Plan, as set forth in <u>Articles IV and V</u> of the Plan, it is clear that every creditor and interest holder will receive at least as much under the Plan as such creditor or interest holder would receive in a chapter 7 liquidation.

D. Feasibility

The Bankruptcy Code requires that a debtor demonstrate that confirmation of a plan is not likely to be followed by liquidation or the need for further financial reorganization of the debtor. In connection with the development of the Plan, and for purposes of determining whether the Plan satisfies that "feasibility" requirement of section 1129(a)(11) of the Bankruptcy Code, Hopeman's management has analyzed the ability of Reorganized Hopeman to meet its obligations under the Plan and to conduct its business after the Effective Date.

The pro forma balance sheets and projected financial performance for Reorganized Hopeman and the projected Asbestos Trust Contribution (collectively, the "<u>Reorganized Hopeman</u> <u>Projections</u>"), which are attached as <u>Exhibit C</u> to this Disclosure Statement, should be read in conjunction with <u>Article X.C</u> above. The Projections were not prepared with a view toward compliance with the guidelines established by the American Institute of Certified Public Accountants, the practices recognized to be in accordance with Generally Accepted Accounting Principles, or the rules and regulations of the Securities and Exchange Commission regarding projections. Furthermore, the Projections have not been audited by an independent accountant. The Plan Proponents do not intend, and disclaims any obligation, to furnish updated projections to holders of Claims prior to the Effective Date.

The Projections were prepared exclusively by Hopeman's management and financial advisors and are based upon, and assume the successful implementation of, the Plan. The Projections, while presented with numerical specificity, are necessarily based on a variety of estimates and assumptions that, though considered reasonable at the time made, may prove not to be accurate, and are inherently subject to significant business, economic, and competitive uncertainties and contingencies, many of which are beyond Hopeman's control. Hopeman cannot make any representations as to Reorganized Hopeman's ability to achieve the results set forth in the Projections. Some assumptions on which the Projections are based may not materialize, and events and circumstances occurring after the date on which the Projections were prepared may be different from those assumed or anticipated, and may materially and adversely impact Reorganized Hopeman's future financial performance. The Projections, therefore, cannot be relied upon as a guarantee or other assurance of Reorganized Hopeman's actual future financial performance or of a particular outcome or result.

Based upon the Projections, the Plan Proponents believe that Reorganized Hopeman will be able to make all payments required pursuant to the Plan and, therefore, that confirmation of the Plan is not likely to be followed by liquidation or the need for further reorganization.

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 125 of 129

E. Cramdown

Section 1129(b) of the Bankruptcy Code provides that if the applicable requirements of section 1129(a) of the Bankruptcy Code, except for section 1129(a)(8),⁹ are met with respect to a plan, then a court, on request of the plan proponent, "shall confirm the plan . . . if the plan does not discriminate unfairly, and is *fair and equitable*, with respect to each class of claims or interests that is impaired under, and has not accepted, the plan." 11 U.S.C. § 1129(b) (emphasis added). The "fair and equitable" test applies to classes of different priority and statures (*e.g.*, secured versus unsecured) and includes the general requirement that no class of claims receives more than 100% of the allowed amount of the claims in such class.

The holders of Equity Interests in Hopeman are not receiving a distribution under the Plan and thus are deemed to vote "no" to the Plan. Assuming the other confirmation requirements of section 1129 of the Bankruptcy Code are satisfied, the Plan Proponent may seek confirmation of the Plan pursuant to section 1129(b) of the Bankruptcy Code on the basis that the Plan is "fair and equitable" to holders of claims in Classes 3 and 5.

XIV.

Recommendation

The Plan Proponents believe that the Plan provides for the highest and most timely recovery available to holders of Claims in Class 3 (General Unsecured Claims) and Class 4 (Channeled Asbestos Claims). Therefore, the Plan Proponents recommend that all holders of General Unsecured Claims in Class 3 and Channeled Asbestos Claims in Class 4 vote to accept (vote in favor of) the Plan and urge each of them to evidence such acceptance by returning their respective ballots so that they are *actually received* by the Balloting Agent **on or before 5:00 p.m.** (Eastern Time) on _____, 2025.

Dated: April 29 May 20, 2025

HOPEMAN BROTHERS, INC.

By: ____

Christopher Lascell President

⁹ Section 1129(a)(8) of the Bankruptcy Code states "[w]ith respect to each class of claims or interests—

(B) such class is not impaired under the plan."

⁽A) such class has accepted the plan; or

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 126 of 129

EXHIBIT A

PLAN OF REORGANIZATION

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 127 of 129

EXHIBIT B

LIQUIDATION ANALYSIS

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 128 of 129

EXHIBIT C

REORGANIZED HOPEMAN PROJECTIONS

Case 24-32428-KLP Doc 768 Filed 05/21/25 Entered 05/21/25 09:51:46 Desc Main Document Page 129 of 129

Summary report: Litera Compare for Word 11.9.1.1 Document comparison done on 5/21/2025 9:13:48 AM			
		Style name: Firm Standard	
		Intelligent Table Comparison: Active	
Original DMS: iw://hunton.cloudimanage.com/DMS/351041352/2			
Modified DMS: iw://hunton.cloudimanage.com/DMS/351309556/1			
Changes:			
Add	33		
Delete	28		
Move From	0		
Move To	0		
Table Insert	0		
Table Delete	0		
Table moves to	0		
Table moves from	0		
Embedded Graphics (Visio, ChemDraw, Images etc.)	0		
Embedded Excel	0		
Format changes	0		
Total Changes:	61		